# POWER PURCHASE AGREEMENT

### FOR

# FIRM QUALIFIED CLEAN ENERGY

### FROM

# HYDROELECTRIC GENERATION

### BETWEEN

### MASSACHUSETTS ELECTRIC COMPANY AND

### NANTUCKET ELECTRIC COMPANY d/b/a NATIONAL GRID

AND

# H.Q. ENERGY SERVICES (U.S.) INC.

as of June 13, 2018

# TABLE OF CONTENTS

1.	DEFINI	TIONS	1
2. EFFECTIVE DATE; TERM			17
	2.1	Effective Date	17
	2.2	Term	17
3.	PROJEC	CT DEVELOPMENT AND OPERATION	17
	3.1	Commercial Operation Date.	17
	3.2	Delay Damages	
	3.3	Progress Reports; Site Access; Capacity Deficiency	19
	3.4	Conditions Precedent to the Commercial Operation Date	20
	3.5	Operation of the Hydro-Québec Power Resources	21
	3.6	Interconnection and Delivery Services	24
4.	DELIVE	ERY OF PRODUCTS	25
	4.1	Obligation to Sell and Purchase Products	25
	4.2	Scheduling and Delivery	25
	4.3	Failure of Seller to Deliver Products.	27
	4.4	Failure by Buyer to Accept Delivery of Products; Default under TSA	30
	4.5	Delivery Point	
	4.6	Metering	
	4.7	Environmental Attributes	31
5.	PRICE A	AND PAYMENTS FOR PRODUCTS	
	5.1	Price for Products	33
	5.2	Payment and Netting	
	5.3	Interest on Late Payment or Refund	34
	5.4	Taxes, Fees and Levies	35

# TABLE OF CONTENTS (CONT'D)

6.	SECUR	ITY FOR PERFORMANCE	5
	6.1	Grant of Security Interest	5
	6.2	Seller's Support	5
	6.3	Delivery of Credit Support	5
	6.4	Reduction and Substitution of Posted Collateral	7
	6.5	Administration of Posted Collateral	7
	6.6	Exercise of Rights Against Posted Collateral40	)
	6.7	Unsecured Credit Limit	l
	6.8	Return of Credit Support41	l
7.	SENTATIONS, WARRANTIES, COVENANTS AND WLEDGEMENTS42	2	
	7.1	Representations and Warranties of Buyer42	2
	7.2	Representations and Warranties of Seller43	3
	7.3	Continuing Nature of Representations and Warranties45	5
	7.4	Assignment of Transmission Rights45	5
	7.5	Forward Capacity Market Participation45	5
8.	REGUL	ATORY APPROVAL; TERMINATION OF TSA	5
	8.1	Receipt of Regulatory Approval46	5
	8.2	Transmission Approvals	5
	8.3	Transmission Service Agreement46	5
	8.4	DEP Advisory Ruling	7
9.	BREAC	HES; REMEDIES48	3
	9.1	Events of Default by Either Party48	3

# TABLE OF CONTENTS (CONT'D)

	9.2	Events of Default by Seller		
	9.3	Remedies		
10.	FORCE	MAJEURE		
	10.1	Force Majeure		
11.	DISPUT	TE RESOLUTION		
	11.1	Consultation		
	11.2	Disputes to be Resolved by FERC		
	11.3	Consent to Jurisdiction		
	11.4	Waiver of Jury Trial		
12.	CONFII	CONFIDENTIALITY		
	12.1	Nondisclosure		
	12.2	Public Statements		
13.	INDEM	NIFICATION60		
	13.1	Indemnification Obligations60		
	13.2	Failure to Defend60		
14.	ASSIGNMENT AND CHANGE OF CONTROL			
	14.1	Prohibition on Assignments		
	14.2	Permitted Assignment by Seller		
	14.3	Change in Control over Seller		
	14.4	Permitted Assignment by Buyer61		
	14.5	Prohibited Assignments		

# TABLE OF CONTENTS (CONT'D)

15.	TITLE; RISK OF LOSS		
16.	AUDIT		
	16.1	Audit	2
	16.2	Access to Financial Information	2
17.	NOTICE	ES62	2
18.	WAIVER AND MODIFICATION		1
19.	INTERF	RETATION	4
	19.1	Choice of Law	1
	19.2	Headings	1
	19.3	Forward Contract	1
	19.4	Standard of Review	1
	19.5	Change in ISO-NE Rules and Practices	5
	19.6	Dodd Frank Act Representations	5
	19.7	Change in Law or Buyer's Accounting Treatment, Subsequent Judicial or Regulatory Action	5
20.	ASSIGNMENT OF TRANSMISSION RIGHTS		7
21.	WAIVE	R OF IMMUNITIES68	3
22.	COUNTERPARTS; FACSIMILE SIGNATURES68		
23.	NO DUTY TO THIRD PARTIES		
24.	SEVERABILITY		
25.	INDEPENDENT CONTRACTOR		
26.	ENTIRE AGREEMENT		
<u>Exhib</u>	<u>its</u>		

# TABLE OF CONTENTS (CONT'D)

- Exhibit A Description of Hydro-Québec Power Resources
- Exhibit B Schedule of Guaranteed Qualified Clean Energy to be Delivered from the Hydro-Québec Power Resources
- Exhibit C Sample Market Exposure Calculation
- Exhibit D Qualified Clean Energy and Environmental Attribute Pricing
- Exhibit E Description of the New Transmission Facilities
- Exhibit F Form of Progress Report
- Exhibit G Form of Seller Guaranty
- Exhibit H Description of Methodology to Ensure Incremental Hydroelectric Generation

# POWER PURCHASE AGREEMENT

This **POWER PURCHASE AGREEMENT FOR FIRM QUALIFIED CLEAN ENERGY FROM HYDROELECTRIC GENERATION** (as amended from time to time in accordance with the terms hereof, this "<u>Agreement</u>") is entered into as of June 13, 2018 (the "<u>Effective</u> <u>Date</u>"), by and between MASSACHUSETTS ELECTRIC COMPANY and NANTUCKET ELECTRIC COMPANY d/b/a NATIONAL GRID, each a Massachusetts corporation (together, "<u>Buyer</u>"), and H.Q. ENERGY SERVICES (U.S.) INC., a Delaware corporation ("<u>Seller</u>"). Buyer and Seller are individually referred to herein as a "<u>Party</u>" and are collectively referred to herein as the "<u>Parties</u>."

**WHEREAS,** an Affiliate (as defined herein) of Seller owns and operates the Hydro-Québec Power Resources (as defined herein); and

**WHEREAS,** the output of the Hydro-Québec Power Resources, delivered through the New Transmission Facilities (as defined herein), shall constitute incremental hydroelectric generation during the Services Term (as defined herein); and

**WHEREAS,** U.S. Transmission Provider (as defined herein) and an Affiliate of Seller submitted a joint proposal in response to the Request for Proposals for Long-Term Energy Contracts for Clean Energy Projects dated March 31, 2017; and

WHEREAS, pursuant to Section 83D of the Green Communities Act as added by Chapter 188 of the Acts of 2016, An Act to Promote Energy Diversity ("<u>Section 83D</u>"), Buyer is authorized to enter into certain long-term contracts for the purchase of energy or energy and renewable energy certificates from clean energy generators meeting the requirements of Section 83D; and

**WHEREAS,** Buyer and Seller desire to enter into this Agreement whereby Buyer shall purchase and Seller shall sell Qualified Clean Energy and associated Environmental Attributes (each as defined herein) in accordance with the terms of this Agreement.

**NOW, THEREFORE,** in consideration of the premises and of the mutual agreements contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereby agree as follows:

## 1. **DEFINITIONS**

In addition to terms defined in the recitals hereto, the following terms shall have the meanings set forth below. Any capitalized terms used in this Agreement and not defined herein shall have the same meaning as ascribed to such terms under the ISO-NE Practices and ISO-NE Rules.

"<u>Actual Transmission Capacity</u>" shall mean, at the applicable date of determination under Section 3.3(c), the actual transmission transfer capability of the New Transmission Facilities, as built.

"<u>Adverse Determination</u>" shall have the meaning set forth in Section 19.7(b) hereof.

"<u>Advisory Ruling</u>" shall have the meaning set forth in Section 8.4(a) hereof.

"<u>Affiliate</u>" shall mean, with respect to any Person, any other Person that directly or indirectly through one or more intermediaries Controls, is Controlled by, or is under common Control with, such first Person; provided, however, that, with respect to Seller, a Person shall not be an "Affiliate" of Seller unless such Person is Hydro-Québec (including, for the avoidance of doubt, a division of Hydro-Québec) or Controlled by Hydro-Québec.

"<u>Agreement</u>" shall have the meaning set forth in the first paragraph of this Agreement.

"<u>APS</u>" shall mean the requirements established pursuant to Mass. Gen. Laws ch. 25A, Section 11F<sup>1</sup>/<sub>2</sub> and the regulations promulgated thereunder that require all retail electricity suppliers in Massachusetts to provide a minimum percentage of electricity from alternative energy portfolio standard generation units, and such successor laws and regulations as may be in effect from time to time.

"<u>Baseline Hydroelectric Generation Imports</u>" shall have the meaning set forth in Exhibit H.

"<u>Baseline Hydroelectric Generation Imports Shortfall</u>" shall have the meaning set forth in Exhibit H.

"<u>Baseline Hydroelectric Generation Imports Shortfall Damages</u>" shall have the meaning set forth in Exhibit H.

"<u>Business Day</u>" shall mean a day on which Federal Reserve member banks in New York, New York are open for business; and a Business Day shall start at 8:00 a.m. and end at 5:00 p.m. Eastern Prevailing Time.

"<u>Buyer</u>" shall have the meaning set forth in the first paragraph of this Agreement.

"<u>Buyer's Exposure</u>" shall mean any positive difference between (a) Market Exposure and (b) the sum of (i) any payment due from Buyer to Seller pursuant to Article 5 which has not yet been made and (ii) any Credit Support provided pursuant to Sections 6.2(b) and 6.2(c).

"<u>Buyer's Taxes</u>" shall have the meaning set forth in Section 5.4(a) hereof.

"<u>Capacity Deficiency</u>" shall mean, at the applicable date of determination under Section 3.3(c), the amount (expressed in MW), if any, by which the Actual Transmission Capacity is less than the proposed transmission transfer capability of the New Transmission Facilities as set forth in Exhibit E.

"<u>Cash</u>" shall mean U.S. dollars held by or on behalf of a Party as Posted Collateral hereunder.

"<u>Certificate</u>" shall mean an electronic certificate created pursuant to the GIS Operating Rules or any successor thereto to represent certain generation attributes of each MWh of Energy generated within the ISO-NE control area and the generation attributes of certain Energy imported into the ISO-NE control area. For the avoidance of doubt, Certificate(s) shall include any and all Environmental Attributes associated with the Qualified Clean Energy and Qualified Shortfall Energy from the Hydro-Québec Power Resources and shall represent title to and claim over all such Environmental Attributes.

"<u>CES</u>" shall mean the Clean Energy Standard requirements established pursuant to the regulations promulgated at 310 CMR 7.75 that require all retail electricity suppliers in Massachusetts to provide a minimum percentage of electricity from certain clean energy generating sources, and such successor laws and regulations as may be in effect from time to time.

"<u>CFTC rules</u>" shall have the meaning specified in Section 19.6 hereof.

"<u>Collateral Account</u>" shall have the meaning specified in Section 6.5(a)(iii)(B) hereof.

"<u>Collateral Interest Rate</u>" shall mean the rate published in *The Wall Street Journal* as the "Prime Rate" from time to time (or, if more than one such rate is published, the arithmetic mean of such rates), or, if such rate is no longer published, a successor rate agreed to by Buyer and Seller, in each case, determined as of the date the obligation to pay interest arises, but in no event more than the maximum rate permitted by applicable Law in transactions involving entities having the same characteristics as the Parties.

"<u>Collateral Requirement</u>" shall mean at any time the amount of Credit Support required under this Agreement at such time.

"<u>Commercial Operation Date</u>" shall mean the date on which the conditions set forth in Section 3.4(b) have been satisfied, as set out in a written notice from Seller to Buyer.

"<u>Commercially Reasonable Efforts</u>" or "<u>Commercially Reasonable Manner</u>" shall mean, with respect to any purchase or sale or other action required to be made, attempted, or taken by a Party under this Agreement, such efforts as a prudent business would undertake (x) for the protection of its own interests under the conditions affecting such purchase or sale or other action, including the amount of notice of the need to take such purchase or sale or other action, the duration and type of the purchase or sale or other action, the competitive environment in which such purchase or sale or other action occurs, the risk and cost to the Party required to take such purchase or sale or other action, the obligations of the Party under this Agreement to the other Party, and (y) to mitigate any disproportionate impact of such purchase or sale or other action on the other Party.

"<u>Contract Maximum Amount</u>" shall mean the highest MWh per hour value of Guaranteed Qualified Clean Energy.

"<u>Contract Year</u>" shall mean the twelve (12) consecutive calendar months starting on the first day of the calendar month following the Commercial Operation Date and each

subsequent twelve (12) consecutive calendar month period; provided that the first Contract Year shall include the days in the prior month in which the Commercial Operation Date occurred.

"<u>Control</u>" shall mean the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, by contract or otherwise.

"Cover Damages" shall mean, with respect to any Uncured Delivery Shortfall, an amount equal to the positive net amount, if any, of (a) the sum, without duplication, of (i) the Replacement Price applicable to that Uncured Delivery Shortfall multiplied by the amount of the Uncured Delivery Shortfall (in MWh) plus (ii) any additional transmission costs and charges reasonably incurred by Buyer to transmit Replacement Energy with respect to that Uncured Delivery Shortfall to the Delivery Point, but solely to the extent that such costs and charges exceed the amount that Buyer incurred under the TSA with respect to such Uncured Delivery Shortfall, plus (iii) any other costs reasonably incurred by Buyer to purchase Replacement Energy and/or Replacement Environmental Attributes with respect to that Uncured Delivery Shortfall, plus (iv) any applicable penalties and other costs assessed by ISO-NE or any other Person against Buyer as a result of that Uncured Delivery Shortfall, plus (v) any other costs and losses reasonably incurred by Buyer as a result of that Uncured Delivery Shortfall, minus (b) the applicable Price that would have been paid pursuant to Section 5.1 hereof for the quantity of Products not Delivered as a result of that Uncured Delivery Shortfall multiplied by the amount of the Uncured Delivery Shortfall (in MWh), plus (c) any costs or charges incurred by Buyer under the TSA associated with the aggregate quantity of Energy (expressed in MWh) not Delivered as a result of that Uncured Delivery Shortfall; provided, however, that, for the avoidance of doubt, if and when the TSA is terminated, then, with respect to any Uncured Delivery Shortfalls that occur after such termination, the Cover Damages shall exclude any costs or charges in respect of the TSA. Buyer shall provide a statement for the applicable period explaining in reasonable detail the calculation of any Cover Damages.

"<u>Credit Rating</u>" shall mean the rating then assigned to Seller Guarantor's unsecured, senior long-term debt obligations (not supported by third party credit enhancements) or if Seller Guarantor does not have a rating for its senior unsecured long-term debt then one rating notch below the rating then assigned to Seller Guarantor or such third party as an issuer and/or corporate credit rating by S&P, Moody's, or Fitch. In the event of an inconsistency in ratings (a "split rating"), the lowest of the Credit Ratings shall control.

"Credit Support" shall have the meaning specified in Section 6.2(e) hereof.

"<u>Credit Support Delivery Amount</u>" shall have the meaning specified in Section 6.3 hereof.

"<u>Credit Support Return Amount</u>" shall have the meaning specified in Section 6.4 hereof.

"<u>Curable Delivery Shortfall</u>" shall have the meaning specified in Section 4.3(c) hereof.

"<u>Custodian</u>" shall have the meaning specified in Section 6.5(a)(i) hereof.

"<u>Default</u>" shall mean any event or condition which, with the giving of notice or passage of time or both, could become an Event of Default.

"<u>Defaulted Delivery Shortfall</u>" shall have the meaning specified in Section 9.2(f) hereof.

"<u>Defaulting Party</u>" shall mean the Party with respect to which a Default or Event of Default has occurred.

"Delay Damages" shall mean the damages assessed pursuant to Section 3.2(a) hereof.

"<u>Deliver</u>" or "<u>Delivery</u>" shall mean with respect to (a) Qualified Clean Energy, to physically supply Qualified Clean Energy to the Delivery Point and transfer such Energy into Buyer's ISO-NE account settled at the Delivery Point in accordance with the terms of this Agreement and the ISO-NE Rules, (b) Qualified Shortfall Energy, to physically supply Qualified Shortfall Energy to a delivery point designated by Seller within the New England Control Area and transfer such Energy into Buyer's ISO-NE account settled at the Delivery Point or, as reasonably agreed from time to time by Buyer and Seller, the NEMASSBOST Hub in accordance with the terms of this Agreement and the ISO-NE Rules, to supply Environmental Attributes associated with such Qualified Clean Energy or Qualified Shortfall Energy, as applicable, in accordance with Section 4.7.

"<u>Delivery Point</u>" shall mean the southern terminus of the U.S. Transmission Line at Larrabee Road substation in Lewiston, Maine, as illustrated in Attachment A to the TSA.

"<u>Delivery Schedule</u>" shall mean Seller's obligation to Deliver and Buyer's rights to receive the MWhs of Qualified Clean Energy and associated Environmental Attributes during the Services Term as provided in Exhibit B.

"<u>Delivery Shortfall</u>" shall have the meaning set forth in Section 4.3(a) hereof.

"<u>Delivery Shortfall LMP</u>" shall have the meaning set forth in Section 4.3(c)(viii) hereof.

"<u>Dispute</u>" shall have the meaning set forth in Section 11.1(a) hereof.

"Disputing Party" shall have the meaning set forth in Section 6.6(a) hereof.

"<u>Downgrade Event</u>" shall mean an event where Seller Guarantor's Credit Rating falls below an Investment Grade Rating, or Seller Guarantor ceases to have a Credit Rating.

"<u>Eastern Prevailing Time</u>" shall mean either Eastern Standard Time or Eastern Daylight Savings Time, as in effect from time to time.

"Effective Date" shall have the meaning set forth in the first paragraph hereof.

"Energy" shall mean electric "energy," as such term is defined in the ISO-NE Tariff.

"Environmental Attributes" shall mean any and all generation attributes under the CES, the GWSA, and under any and all other international, federal, regional, state or other law, rule, regulation, bylaw, treaty or other intergovernmental compact, decision, administrative decision, program (including any voluntary compliance or membership program), competitive market or business method (including all credits, certificates, benefits, and emission measurements, reductions, offsets and allowances related thereto) that are attributable, now or in the future, to the favorable generation or environmental attributes of the Hydro-Québec Power Resources, the Qualified Clean Energy or the Qualified Shortfall Energy produced by the Hydro-Québec Power Resources including, but not limited to: (a) any such credits, certificates, benefits, offsets and allowances computed on the basis of the Hydro-Québec Power Resources' generation using renewable technology or displacement of fossil-fuel derived or other conventional energy generation; (b) any Certificates issued pursuant to the GIS in connection with Energy generated by the Hydro-Québec Power Resources; and (c) any voluntary emission reduction credits obtained or obtainable by Seller in connection with the Qualified Clean Energy or the Qualified Shortfall Energy, as the case may be; provided, however, that Environmental Attributes shall not include: (i) any state or federal production tax credits; (ii) any state or federal investment tax credits or other tax credits associated with the construction or ownership of the Hydro-Québec Power Resources; or (iii) any state, federal or private grants, financing, guarantees or other credit support relating to the construction or ownership, operation or maintenance of the Hydro-Québec Power Resources or the output thereof.

"<u>Event of Default</u>" shall have the meaning set forth in Section 9.1 hereof and shall include the events and conditions described in Section 9.1 and Section 9.2 hereof.

"<u>Excused Outage</u>" shall mean an Excused Outage (as defined in the TSA) other than any such Excused Outage due to outages or reductions in the availability of the Québec Line.

"<u>Federal Funds Rate</u>" shall mean the annualized interest rate for the applicable month as set forth in the statistical release designated as H.15, or any successor publication, published by the Board of Governors of the Federal Reserve System.

"<u>FERC</u>" shall mean the United States Federal Energy Regulatory Commission, and shall include its successors.

"FERC Jurisdictional Issue" shall have the meaning set forth in Section 11.2(a) hereof.

"Fitch" shall mean Fitch Investor's Service, Inc., or its successor.

"<u>Five-Year Average Baseline Hydroelectric Generation Imports Shortfall</u>" shall have the meaning set forth in Exhibit H.

"<u>Fixed Amount</u>" shall mean the applicable dollar amounts set forth in Section 6.7 under the column heading "Fixed Amount."

"<u>Fixed Credit Support</u>" shall have the meaning set forth in Section 6.2(b) hereof.

"Floor Price" shall have the meaning set forth in Exhibit H hereof.

"<u>Force Majeure</u>" shall have the meaning set forth in Section 10.1(a) hereof.

"<u>Generation Unit</u>" shall mean a facility that converts a fuel or an energy resource into electrical energy.

"<u>GIS</u>" shall mean the NEPOOL Generation Information System or any successor thereto, which includes a generation information database and certificate system, operated by NEPOOL, its designee or successor entity, that accounts for generation attributes of electricity generated or consumed within the New England Control Area.

"<u>GIS Operating Rules</u>" are the NEPOOL Generation Information System Operating Rules effective as of the Effective Date, as amended, superseded, or restated from time to time.

"<u>Good Utility Practice</u>" shall mean compliance with all applicable laws, codes, rules and regulations, all ISO-NE Rules and ISO-NE Practices, and any practices, methods and acts engaged in or approved by a significant portion of the electric industry in New England or the Province of Québec during the relevant time period, or any of the practices, methods and acts which, in the exercise of reasonable judgment in light of the facts known at the time the decision is made, could have been expected to accomplish the desired result consistent with good business practices, reliability, safety and expedition. Good Utility Practice is not intended to be limited to the optimum practice, method or act to the exclusion of all others, but rather is intended to include acceptable practices, methods and acts generally accepted in the electric industry in New England or the Province of Québec.

"<u>Governmental Entity</u>" shall mean any federal, state or local governmental agency, authority, department, instrumentality or regulatory body, and any court or tribunal, with jurisdiction over Seller, Buyer or the Hydro-Québec Power Resources.

"<u>Guaranteed Delivery Term Start Date</u>" shall mean December 13, 2022, as may be extended pursuant to Sections 3.1(c) through 3.1(f).

"<u>Guaranteed Qualified Clean Energy</u>" shall mean Seller's firm obligations to provide Qualified Clean Energy and associated Environmental Attributes pursuant to this Agreement, as provided in the Delivery Schedule included in Exhibit B.

"<u>GWSA</u>" shall mean the Massachusetts Global Warming Solutions Act (Mass. Gen. Laws ch. 298), and such successor laws and regulations as may be in effect from time to time.

"<u>Hydro-Québec</u>" shall mean Hydro-Québec, a body politic and corporate, duly incorporated and regulated by the Hydro-Québec Act (R.S.Q., Chapter H-5).

"Hydro-Québec Power Resources" shall mean, collectively, those existing hydroelectric generating stations, located in the Province of Québec and owned and operated as a system by Hydro-Québec or its subsidiaries from time to time, that produce electric energy, which consists predominantly of low-carbon and renewable hydro-electric energy during the Services Term, which are further described in Exhibit A.

"<u>ICE</u>" shall have the meaning set forth in the definition of Off-Peak Forward Price.

"<u>Immunities Act</u>" means the United States Foreign Sovereign Immunities Act of 1976, 28 U.S.C. § 1602 et seq.

"<u>Interconnecting Utility</u>" shall mean, as the context requires, TransÉnergie or the utility providing interconnection service for the Hydro-Québec Power Resources to the transmission system of that utility.

"Interconnection Agreements" shall mean, collectively, (a) the agreement(s) required to be executed among U.S. Transmission Provider, TransÉnergie and ISO-NE to interconnect the U.S. Transmission Line with the Québec Line at the U.S./Canada border and (b) the agreement(s) required to be executed among U.S. Transmission Provider and ISO-NE that sets forth such parties' respective rights and obligations following the interconnection at the Delivery Point of the U.S. Transmission Line with certain transmission facilities operated by ISO-NE.

"<u>Interest Amount</u>" shall mean with respect to a Party and an Interest Period, the sum of the daily interest amounts for all days in such Interest Period; each daily interest amount to be determined by such Party as follows: (a) the amount of Cash held by such Party on that day (but excluding any interest previously earned on such Cash); multiplied by (b) the Collateral Interest Rate for that day; divided by (c) 360.

"<u>Interest Period</u>" shall mean the period from (and including) the last Business Day on which an Interest Amount was Transferred by Buyer (or if no Interest Amount has yet been Transferred by Buyer, the Business Day on which Cash was Transferred to Seller) to (but excluding) the Business Day on which the current Interest Amount is to be Transferred.

"<u>ISO</u>" or "<u>ISO-NE</u>" shall mean ISO New England Inc., the independent system operator established in accordance with the RTO arrangements for New England, or its successor.

"<u>ISO-NE Practices</u>" shall mean the ISO-NE practices and procedures for delivery and transmission of energy in effect from time to time and shall include, without limitation, applicable requirements of the NEPOOL Agreement, and any applicable successor practices and procedures.

"<u>ISO-NE Rules</u>" shall mean all rules and procedures adopted by NEPOOL, ISO-NE, or the RTO, and governing wholesale power markets and transmission in New England, as such rules may be amended from time to time, including but not limited to, the ISO-NE Tariff, the ISO-NE Operating Procedures (as defined in the ISO-NE Tariff), the ISO-NE Planning Procedures (as defined in the ISO-NE Tariff), the Transmission Operating Agreement (as defined in the ISO-NE Tariff), the ISO-NE Participants Agreement, the manuals, procedures and business process documents published by ISO-NE via its web site and/or by its e-mail distribution to appropriate NEPOOL participants and/or NEPOOL committees or as otherwise made available publicly to the Parties, as amended, superseded or restated from time to time.

"<u>ISO-NE Tariff</u>" shall mean ISO-NE's Transmission, Markets and Services Tariff, as amended, superseded or restated from time to time.

"Late Payment Rate" shall have the meaning set forth in Section 5.3 hereof.

"Law" shall mean all federal, state and local statutes, regulations, rules, orders, executive orders, decrees, policies, judicial decisions and notifications, including without limitation international treaty obligations and also including without limitation Section 83D, the regulations promulgated under Section 83D, the Regulatory Approval and any other orders of the MDPU with respect to this Agreement.

"Letter of Credit" shall mean an irrevocable, non-transferable standby letter of credit issued by a Qualified Institution utilizing a form acceptable to the Party in whose favor such letter of credit is issued. All costs relating to any Letter of Credit shall be for the account of the Party providing that Letter of Credit.

"Letter of Credit Default" shall mean with respect to an outstanding Letter of Credit, the occurrence of any of the following events (a) the issuer of such Letter of Credit shall fail to be a Qualified Institution; (b) the issuer of the Letter of Credit if such failure shall be continuing after the lapse of any applicable grace period; (c) the issuer of the Letter of Credit shall disaffirm, disclaim, repudiate or reject, in whole or in part, or challenge the validity of, such Letter of Credit Support hereunder and that Letter of Credit is issued is required to provide Credit Support hereunder and that Party has not Transferred replacement Credit Support meeting the requirements of this Agreement; provided, however, that no Letter of Credit Default shall occur in any event with respect to a Letter of Credit after the time such Letter of Credit is required to be cancelled or returned in accordance with the terms of this Agreement.

"<u>Limited Suspension Period</u>" shall have the meaning set forth in 4.3(d) hereto.

"<u>Market Exposure</u>" shall mean the summation of the amounts calculated for each full calendar month remaining in the Services Term, up to a total of sixty (60) months, of (a) the product of (i) the On-Peak Forward Price minus the On-Peak Energy Price multiplied by (ii) the On-Peak quantity of Guaranteed Qualified Clean Energy, plus (b) the product of (i) the Off-Peak Forward Price minus the Off-Peak Energy Price multiplied by (ii) the Off-Peak quantity of Guaranteed Qualified Clean Energy. The format for calculating the Market Exposure is provided in Exhibit C.

"<u>MDPU</u>" shall mean the Massachusetts Department of Public Utilities and its successors.

"<u>Metered Output</u>" shall mean the Energy, expressed in MWh, generated by each of the Hydro-Québec Power Resources in each hour during the Services Term.

"<u>Meters</u>" shall have the meaning set forth in Section 4.6(a) hereof.

"<u>Minimum Required Baseline Hydroelectric Generation Imports</u>" shall have the meaning set forth in Exhibit H.

"Moody's" shall mean Moody's Investors Service, Inc., and any successor thereto.

"<u>MW</u>" shall mean a megawatt alternating current (AC).

"<u>MWh</u>" shall mean a megawatt-hour (one MWh shall equal 1,000 kWh).

"NEPOOL" shall mean the New England Power Pool and any successor organization.

"<u>NERC</u>" shall mean the North American Electric Reliability Corporation and shall include any successor thereto.

"<u>Network Upgrades</u>" shall mean those upgrades to the Pool Transmission Facilities and the Transmission Provider's transmission and distribution systems, as determined and identified by ISO-NE and contemplated by Section 7.5.

"New England Control Area" shall have the meaning as set forth in the ISO-NE Tariff.

"<u>New Transmission Facilities</u>" shall mean the Québec Line and the U.S. Transmission Line, which transmission facilities will be used by Seller for delivery of the Qualified Clean Energy under this Agreement.

"<u>Non-Defaulting Party</u>" shall mean the Party with respect to which a Default or Event of Default has not occurred.

"<u>Non-TSA Delivery Shortfall</u>" shall have the meaning set forth in Section 4.3(c)(iii) hereof.

"<u>Non-TSA Shortfall Default Percentage</u>" shall mean the percentage equal to one hundred percent (100%) minus the TSA Shortfall Default Percentage.

"<u>NYMEX</u>" shall have the meaning set forth in the definition of Off-Peak Forward Price.

"<u>Obligations</u>" shall have the meaning specified in Section 6.1 hereof.

"<u>Off-Peak</u>" shall mean all hours not defined as On-Peak.

"Off-Peak Energy Price" shall mean the Off-Peak Contract Price set forth in Exhibit C.

"<u>Off-Peak Forward Price</u>" shall mean the arithmetic average of the future Off-Peak market prices, which shall be the mid-point between the bid and ask prices at the ISO-NE Internal Hub as quoted by the Intercontinental Exchange ("<u>ICE</u>"), ICAP Energy, or the

New York Mercantile Exchange ("<u>NYMEX</u>"), at Buyer's discretion. If quoted market prices are available from ICE, ICAP Energy, or NYMEX, as applicable, for only part of the Services Term, Buyer will extrapolate the market prices for the period not available based on other available information in a Commercially Reasonable Manner. If quoted market prices from the selected source are not available for any part of the applicable Services Term, the Parties shall agree to alternative sources for market prices for such period.

"<u>On-Peak</u>" shall mean all hours included in the period beginning with the hours ending at 0800 and ending at the hour ending 2300 on all weekdays, excluding NERC holidays.

"On-Peak Energy Price" shall mean the On-Peak Contract Price set forth in Exhibit C.

"<u>On-Peak Forward Price</u>" shall mean the arithmetic average of the future On-Peak market prices, which shall be the mid-point between the bid and ask prices at the ISO-NE Internal Hub as quoted by ICE, ICAP Energy, or NYMEX, at Buyer's discretion. If quoted market prices are available from ICE, ICAP Energy, or NYMEX, as applicable, for only part of the Services Term, Buyer will extrapolate the market prices for the period not available based on other available information in a Commercially Reasonable Manner. If quoted market prices from the selected source are not available for any part of the applicable Services Term, the Parties shall agree to alternative sources for market prices for such period.

"<u>Party</u>" and "<u>Parties</u>" shall have the meaning set forth in the first paragraph of this Agreement.

"<u>Permits</u>" shall mean any permit, authorization, license, order, consent, waiver, exception, exemption, variance or other approval by or from, and any filing, report, certification, declaration, notice or submission to or with, any Governmental Entity required to authorize action, including any of the foregoing relating to the ownership, siting, construction, operation, use or maintenance of the Hydro-Québec Power Resources under any applicable Law and required for the Delivery of the Products in accordance with this Agreement.

"<u>Person</u>" shall mean an individual, partnership, corporation, limited liability company, limited liability partnership, limited partnership, association, trust, unincorporated organization, or a government authority or agency or political subdivision thereof.

"<u>Posted Collateral</u>" shall mean all Credit Support and all proceeds thereof that have been Transferred to or received by a Party under this Agreement and not Transferred to the Party providing the Credit Support or released by the Party holding the Credit Support. Any Interest Amount or portion thereof not Transferred will constitute Posted Collateral in the form of Cash.

"<u>Power Cost Reconciliation Tariff</u>" shall mean a fully reconciling cost recovery tariff mechanism that authorizes the establishment of a distribution charge that fully recovers Buyer's net costs under this Agreement (including annual remuneration of up to two and three-quarters percent (2.75%)). The rate reconciliation shall be designed in such a way

as to limit the build-up of any under or over-recoveries over the course of the year. A reconciliation shall occur at least annually, but may also be reconciled quarterly or monthly, to the extent necessary to eliminate regulatory lag for the recovery of costs or crediting of over-recoveries to customers.

"<u>Price</u>" shall mean the purchase price(s) for Products referenced in Section 5.1 hereof and set forth on Exhibit D.

"<u>Products</u>" shall mean Qualified Clean Energy, Qualified Shortfall Energy and associated Environmental Attributes.

"<u>Purchased Power Accounting Authorization</u>" shall mean authorization for Buyer, at Buyer's sole discretion, to take appropriate steps to assure avoidance of a material, negative balance sheet impact on Buyer or Buyer's direct or indirect parent company, upon appropriate filing with and approval by the MDPU.

"<u>Qualified Clean Energy</u>" shall mean energy produced by a hydroelectric generating resource. For the avoidance of doubt, Qualified Clean Energy as used in this Agreement refers only to that Qualified Clean Energy from the Hydro-Québec Power Resources and delivered over the New Transmission Facilities during the Services Term. This Energy must be tracked in the GIS to ensure a unit-specific accounting of the Delivery of Qualified Clean Energy to enable the Massachusetts Department of Environmental Protection to accurately account for such Qualified Clean Energy in the state greenhouse gas emissions inventory, created under chapter 298 of the Acts of 2008.

"<u>Qualified Clean Energy Generation Units</u>" shall mean Generation Units capable of producing Qualified Clean Energy, or Qualified Shortfall Energy, as the case may be.

"<u>Qualified Institution</u>" shall mean a major U.S. commercial bank or trust company, the U.S. branch office of a foreign bank, or another financial institution, in any case, organized under the laws of the United States or a political subdivision thereof having assets of at least \$10 billion and a credit rating of at least (a) "A3" from Moody's or "A-" from S&P, if such entity is rated by both S&P and Moody's or (b) "A-" by S&P or "A3" by Moody's, if such entity is rated by either S&P or Moody's but not both.

"<u>Qualified Shortfall Energy</u>" shall mean energy produced by a hydroelectric generating resource and Delivered over any transmission line by Seller to Buyer into the New England Control Area during the Services Term. For the avoidance of doubt, Qualified Shortfall Energy as used in this Agreement refers only to that Qualified Shortfall Energy from the Hydro-Québec Power Resources. This Energy must be tracked in the GIS to ensure a unit-specific accounting of the delivery of Qualified Shortfall Energy to enable the Massachusetts Department of Environmental Protection to accurately account for such Qualified Shortfall Energy in the state greenhouse gas emissions inventory, created under chapter 298 of the Acts of 2008.

"<u>Qualified Shortfall Energy LMP</u>" shall have the meaning set forth in Section 4.3(c)(viii) hereof.

"<u>Québec Line</u>" shall have the same meaning as set forth in the TSA.

"<u>Québec Line Approvals</u>" shall mean those agreements, tariffs and approvals associated with service on the Québec Line.

"<u>Reference Market-Maker</u>" shall mean a leading dealer in the relevant market that is selected in a Commercially Reasonable Manner and is not an Affiliate of either Party.

"Regulatory Approval" shall mean the MDPU approval, of this entire Agreement, which approval shall include without limitation: (a) confirmation that this Agreement has been approved under Section 83D and the regulations promulgated thereunder and that all of the terms of such Section 83D and such regulations apply to this Agreement; (b) definitive regulatory authorization for Buyer to recover all of its costs incurred under and in connection with this Agreement for the entire term of this Agreement through the implementation of a Power Cost Reconciliation Tariff and/or other cost recovery or reconciliation mechanisms; (c) definitive regulatory authorization for Buyer to recover remuneration of up to two and three-quarters percent (2.75%) of Buyer's annual payments under this Agreement for the term of this Agreement through the Power Cost Reconciliation Tariff; and (d) approval of any Purchased Power Accounting Authorization requested by Buyer in connection with the Regulatory Approval. Such approvals shall be acceptable in form and substance to Buyer in its sole discretion, shall not include any conditions or modifications that Buyer deems, in its sole discretion, to be unacceptable and shall be final and not subject to appeal or rehearing.

"<u>Regulatory Approval Delay</u>" shall have the same meaning as set forth in the TSA.

"<u>Rejected Purchase</u>" shall have the meaning set forth in Section 4.4(a) hereof.

"<u>Reliability Curtailment</u>" shall mean any curtailment of Delivery of Qualified Clean Energy resulting from (a) an emergency condition as defined in the Interconnection Agreements or the ISO-NE Tariff or tariff of another Independent System Operator, if applicable, or (b) any other order or directive of the Interconnecting Utility or the Transmission Provider pursuant to an Interconnection Agreement or tariff.

"<u>**Replacement Energy**</u>" shall mean energy purchased by Buyer as replacement for any Uncured Delivery Shortfall relating to the Qualified Clean Energy to be provided hereunder.

"<u>Replacement Environmental Attributes</u>" shall mean any generation or environmental attributes, including any Certificates or other certificates or credits related thereto reflecting generation by Qualified Clean Energy Generation Units that are purchased by Buyer as replacement for any Environmental Attributes not Delivered as required hereunder.

"<u>Replacement Price</u>" shall mean (a) the price at which Buyer, acting in a Commercially Reasonable Manner, purchases Replacement Energy and Replacement Environmental Attributes; provided, however, that in no event shall Buyer be required to utilize or change its utilization of its owned or controlled assets, contracts or market positions to minimize Seller's liability, or (b) if Buyer elects in its sole discretion not to purchase Replacement Energy and Replacement Environmental Attributes, the market value as determined by Buyer, acting in a Commercially Reasonable Manner, of (i) energy; provided that such price shall not exceed the LMP at the ISO-NE Internal Hub and (ii) Environmental Attributes; provided that such price shall not exceed the compliance payment required under the CES for the applicable year, in each case, as of the date and the time of the Uncured Delivery Shortfall.

"<u>Reporting Party</u>" shall have the meaning set forth in Section 19.6 hereof.

"<u>Request Date</u>" shall have the meaning set forth in Section 6.6(a) hereof.

"<u>Requesting Party</u>" shall have the meaning set forth in Section 6.6(a) hereof.

"<u>Resale Damages</u>" shall mean, with respect to any Rejected Purchase, an amount equal, without duplication, to (a) the positive net amount, if any, by which the applicable Price that would have been paid pursuant to Section 5.1 hereof for such Rejected Purchase, had it been accepted, exceeds the Resale Price, multiplied by the quantity of that Rejected Purchase (in MWh and/or Environmental Attributes, as applicable), plus (b) any applicable penalties and other costs assessed by ISO-NE or any other Person against Seller as a result of Buyer's failure to accept such Products in accordance with the terms of this Agreement, plus (c) transaction and other administrative costs reasonably incurred by Seller in re-selling such Rejected Purchase that Seller would not have incurred but for the Rejected Purchase. Seller shall provide a written statement for the applicable period explaining in reasonable detail the calculation of any Resale Damages.

"<u>Resale Price</u>" shall mean the price at which Seller, acting in a Commercially Reasonable Manner, sells or is paid for a Rejected Purchase; provided, however, that in no event shall Seller or its Affiliates be required to utilize or change its utilization of the Hydro-Québec Power Resources or its other assets, contracts or market positions in order to minimize Buyer's liability for such Rejected Purchase.

"Rounding Amount" shall have the meaning specified in Section 6.2(e) hereof.

"<u>**RPS**</u>" shall mean the requirements established pursuant to Mass. Gen. Laws ch. 25A, Section 11F and the regulations promulgated thereunder that require all retail electricity suppliers in Massachusetts to provide a minimum percentage of electricity from renewable generation units, and such successor laws and regulations as may be in effect from time to time.

"<u>**RTO**</u>" shall mean ISO-NE and any successor organization or entity to ISO-NE, as authorized by FERC to exercise the functions pursuant to FERC's Order No. 2000 and FERC's corresponding regulations, or any successor organization, or any other entity authorized to exercise comparable functions in subsequent orders or regulations of FERC.

"<u>S&P</u>" shall mean Standard & Poor's Financial Services LLC, and any successor thereto.

"<u>Schedule</u>" shall mean the actions of Seller and/or its designated representatives pursuant to Section 4.2, of notifying, requesting and confirming to ISO-NE and other Independent System Operator, as applicable, the quantity of Qualified Clean Energy or Qualified Shortfall Energy, as applicable, to be delivered on any given day or days (or in any given hour or hours) during the Services Term at the Delivery Point.

"<u>Section 83D</u>" shall have the meaning set forth in the recitals hereof.

"<u>Seller</u>" shall have the meaning set forth in the first paragraph of this Agreement.

"<u>Seller Guarantor</u>" shall mean Hydro-Québec.

"<u>Seller Guaranty</u>" shall have the meaning set forth in Section 6.2(a) hereof.

"Seller TSA" shall have the same meaning as the HQUS TSA (as defined in the TSA).

"<u>Seller's Taxes</u>" shall have the meaning set forth in Section 5.4(a) hereof.

"<u>Services Term</u>" shall have the meaning set forth in Section 2.2(b) hereof.

"Shortfall Cure Amount" shall have the meaning set forth in Section 4.3(c) hereof.

"<u>Shortfall Cure Period</u>" shall have the meaning set forth in Section 4.3(c)(i) hereof.

"Substitute Credit Support" shall have the meaning set forth in Section 6.5(e) hereof.

"<u>Summer Period</u>" shall mean the months of June through September.

"<u>Term</u>" shall have the meaning set forth in Section 2.2(a) hereof.

"<u>Termination Payment</u>" shall have the meaning set forth in Section 9.3(b) hereof.

"<u>TransÉnergie</u>" shall mean Hydro-Québec TransÉnergie, the division of Hydro-Québec that operates Hydro-Québec's transmission system.

"<u>**Transfer</u>**" shall mean, with respect to any Posted Collateral or Interest Amount, and in accordance with the instructions of the Party entitled thereto:</u>

(a) in the case of Cash, payment or transfer by wire transfer into one or more bank accounts specified by the Party to whom such Cash is being delivered; and

(b) in the case of Letters of Credit, delivery of the Letter of Credit or an amendment thereto to the Party to whom such Letter of Credit is being delivered.

"<u>Transmission Delay</u>" shall mean, without duplication of the extensions provided in Sections 3.1(c) through 3.1(e), any delay in the satisfaction of Section 3.1(a) or 3.4(b)(ii) (except to the extent such condition has been waived by the Parties), but only if such delay is not attributable to any HQUS Delay (as defined in the TSA).

"<u>Transmission Provider</u>" shall mean (a) ISO-NE or other RTO, as applicable, its respective successor or Affiliates; and/or (b) such other third parties from whom transmission services are necessary for Seller to fulfill its performance obligations to Buyer hereunder, as the context requires, including, for the avoidance of doubt U.S. Transmission Provider and TransÉnergie.

"<u>Transmission System</u>" shall mean the transmission facilities operated by a Transmission Provider, now or hereafter in existence, which provide energy transmission service for the Qualified Clean Energy to or from the Delivery Point.

"<u>TSA</u>" shall mean that certain Transmission Service Agreement, by and between Buyer and U.S. Transmission Provider, dated as of June 13, 2018.

"<u>TSA Commercial Operation Date</u>" shall mean the "Commercial Operation Date" as defined in the TSA.

"<u>TSA Delivery Shortfall</u>" shall have the meaning set forth in Section 4.3(c) hereof.

"<u>TSA Shortfall Default Percentage</u>" shall mean, with respect to a Defaulted Delivery Shortfall, the percentage of the Defaulted Delivery Shortfall that was caused by or arose from TSA Delivery Shortfalls, as determined by Buyer in a Commercially Reasonable Manner.

"<u>TSA Transmission Rights</u>" shall have the meaning set forth in Section 20(a) hereof.

"<u>Uncured Delivery Shortfall</u>" shall have the meaning set forth in Section 4.3(a) hereof.

<u>"U.S. Transmission Approvals</u>" shall mean those MDPU and FERC filings, agreements, tariffs and approvals associated with service on the U.S. Transmission Line, as listed in the TSA.

"<u>U.S. Transmission Line</u>" shall have the same meaning as the NECEC Transmission Line (as defined in the TSA).

"<u>U.S. Transmission Provider</u>" shall mean Central Maine Power Company, and its successors and assignees.

"<u>Unsecured Credit Limit</u>" shall mean Seller's Fixed Amount, as adjusted by Seller Guarantor's Credit Rating pursuant to Section 6.7.

"<u>Valuation Agent</u>" shall mean the Requesting Party; provided, however, that in all cases, if an Event of Default has occurred and is continuing with respect to the Party designated as the Valuation Agent, then in such case, and for so long as the Event of Default continues, the other Party shall be the Valuation Agent.

"<u>Valuation Date</u>" shall mean each Business Day.

"<u>Valuation Percentage</u>" shall have the meaning specified in Section 6.2(e) hereof.

"<u>Valuation Time</u>" shall mean the close of business on the Business Day before the Valuation Date or date of calculation, as applicable.

"<u>Value</u>" shall mean, with respect to Posted Collateral or Credit Support, the Valuation Percentage multiplied by the amount then available under the Letter of Credit to be unconditionally drawn by Buyer.

"<u>Winter Period</u>" shall mean the months of December, January and February.

# 2. EFFECTIVE DATE; TERM

2.1 <u>Effective Date</u>. Subject in all respects to Article 8, this Agreement is effective as of the Effective Date.

# 2.2 <u>Term</u>.

(a) The "<u>Term</u>" of this Agreement is the period beginning on the Effective Date and ending upon the final settlement of all obligations hereunder after the expiration of the Services Term or the earlier termination of this Agreement in accordance with its terms.

(b) The "<u>Services Term</u>" is the period during which Buyer is obligated to purchase Products Delivered to Buyer by Seller commencing on the Commercial Operation Date and continuing for a period of twenty (20) years from the Commercial Operation Date, unless this Agreement is earlier terminated in accordance with the provisions hereof. Seller shall provide at least twenty (20) days' prior written notice to Buyer of the date on which it believes that the Commercial Operation Date will occur.

(c) At the expiration of the Term or earlier termination of this Agreement pursuant to the terms hereof, the Parties shall no longer be bound by the terms and provisions hereof, except (i) to the extent necessary to make payments of any amounts owed to the other Party arising prior to or resulting from termination of, or on account of a breach of, this Agreement, (ii) to the extent necessary to enforce the rights and the obligations of the Parties arising under this Agreement before such expiration or termination, and (iii) the obligations of the Parties hereunder with respect to confidentiality and indemnification shall survive the expiration or termination of this Agreement.

# 3. PROJECT DEVELOPMENT AND OPERATION

# 3.1 <u>Commercial Operation Date</u>.

(a) Subject to the provisions of Sections 3.1(c) through 3.1(f), Seller shall achieve the Commercial Operation Date by the Guaranteed Delivery Term Start Date.

(b) Seller shall provide Buyer with written notice of the achievement of the Commercial Operation Date within seven (7) days after that achievement, which notice shall include information demonstrating with reasonable specificity that such achievement has occurred. Seller acknowledges that Buyer will receive such notice solely for information purposes, and Buyer shall have no responsibility or liability for the development, construction,

operation or maintenance of the Hydro-Québec Power Resources or any Transmission System or New Transmission Facilities.

(c) To the extent U.S. Transmission Provider elects to extend the deadline for the TSA Commercial Operation Date in accordance with the TSA, the Guaranteed Delivery Term Start Date hereunder shall be extended by such period(s), without additional Credit Support required by Seller under this Agreement.

(d) To the extent a Force Majeure event pursuant to Section 10.1 has occurred that prevents Seller from achieving the Commercial Operation Date by the Guaranteed Delivery Term Start Date, the Guaranteed Delivery Term Start Date hereunder shall be extended for the duration of the Force Majeure event, as contemplated by Section 10.1, without additional Credit Support required by Seller under this Agreement or any liability to Seller under this Section 3.1(d), but under no circumstances shall such extension exceed twelve (12) months. Without duplication, to the extent a Force Majeure event pursuant to Section 15.1 of the TSA not attributable to any action or omission of Seller or any of its Affiliates has occurred that prevents U.S. Transmission Provider from achieving the TSA Commercial Operation Date by the applicable Critical Milestone (as defined in the TSA) date, the Guaranteed Delivery Term Start Date shall be extended by such period(s) provided under the TSA, as contemplated by Section 10.1, without additional Credit Support required by, or any liability to, Seller under this Agreement, but under no circumstances shall such extension exceed twelve (12) months.

(e) In the event that the TSA Commercial Operation Date is extended due to a Regulatory Approval Delay in accordance with Section 4.1(e) of the TSA, the Guaranteed Delivery Term Start Date hereunder shall be extended by such period(s) provided under the TSA, subject to reduction of the extension, as provided in Section 4.1(e) of the TSA, without additional Credit Support required by Seller under this Agreement.

(f) To the extent Seller is prevented from achieving the Commercial Operation Date by the Guaranteed Delivery Term Start Date due to a Transmission Delay, the Guaranteed Delivery Term Start Date hereunder shall be extended for the duration of the Transmission Delay, without additional Credit Support required by Seller under this Agreement. Buyer shall be entitled to the remedies provided under the TSA with regard to any such Transmission Delay, and Seller shall have no liability to Buyer for Delay Damages solely as a result of any Owner Delay or Concurrent Delay (each as defined in the TSA).

(g) The Parties agree that time is of the essence with respect to the achievement of the Commercial Operation Date and is part of the consideration to Buyer in entering into this Agreement.

(h) Nothing in this Section 3.1 shall be construed to limit or otherwise modify any of the obligations of U.S. Transmission Provider under the TSA or the remedies to which Buyer may otherwise be entitled under the TSA or the express terms of this Agreement.

# 3.2 <u>Delay Damages</u>.

(a) If the Commercial Operation Date is not achieved by the Guaranteed Delivery Term Start Date (as extended pursuant to Sections 3.1(c) through 3.1(f)), Seller shall

pay to Buyer damages for each day from and after such date in an amount equal to \$100 per MWh per hour of Contract Maximum Amount (e.g., if the Contract Maximum Amount were 364 MWh per hour Seller would pay Buyer damages in the amount of \$36,400 per day), commencing on the Guaranteed Delivery Term Start Date (as extended pursuant to Sections 3.1(c) through 3.1(f)) and ending on the earlier of (i) the Commercial Operation Date, (ii) the date that Buyer exercises its right to terminate this Agreement under Section 9.3, and (iii) the date that is twelve (12) months after the Guaranteed Delivery Term Start Date ("Delay Damages"). Delay Damages shall be due under this Section 3.2(a) without regard to whether Buyer exercises its right to terminate this Agreement under Section 9.3, Delay Damages shall be due and owing only to the extent that such Delay Damages were due and owing at the date of such termination.

(b) Each Party agrees and acknowledges that (i) the damages that Buyer would incur due to Seller's delay in achieving the Commercial Operation Date by the Guaranteed Delivery Term Start Date would be difficult or impossible to predict with certainty, and (ii) it is impractical and difficult to assess actual damages in the circumstances stated, and therefore the Delay Damages as agreed to by the Parties and set forth herein are a fair and reasonable calculation of such damages. Notwithstanding the foregoing, this Article shall not limit the amount of damages payable to Buyer hereunder if this Agreement is terminated as a result of Seller's failure to achieve the Commercial Operation Date. Any such termination damages shall be determined in accordance with Article 9.

(c) By the fifteenth (15th) day following the end of the calendar month in which Delay Damages first become due and continuing by the fifteenth (15th) day of each calendar month during the period in which Delay Damages accrue (and the following month if applicable), Buyer shall deliver to Seller an invoice showing Buyer's computation of such damages and any amount due Buyer in respect thereof for the preceding calendar month. On or before the later of (x) fifteen (15) days from receipt of the applicable invoice or (y) the last day of the calendar month in which the applicable invoice was received (or in either event the next Business Day if such day is not a Business Day), Seller shall pay to Buyer, by wire transfer of immediately available funds to an account specified in writing by Buyer or by any other means agreed to by the Parties in writing from time to time, the amount set forth as due in such invoice. If Seller fails to pay such amounts when due, Buyer may draw upon the Credit Support or the Seller Guaranty for payment of such Delay Damages, and Buyer may exercise any other remedies available for Seller's default hereunder.

### 3.3 <u>Progress Reports; Site Access; Capacity Deficiency</u>.

(a) <u>Progress Reports</u>. Within ten (10) days after the end of each calendar quarter after the Effective Date and until the Commercial Operation Date, Seller shall provide Buyer with a report regarding progress towards the Commercial Operation Date, including the status of permitting and acquisition of real property rights for the Québec Line, in accordance with the form attached hereto as Exhibit F, and shall provide supporting documents and detail regarding the same upon Buyer's request. Seller shall permit Buyer and its advisors and consultants to review and discuss with Seller and its advisors and consultants such progress reports during business hours and upon reasonable notice to Seller.

(b) <u>Site Access</u>. Buyer and its representatives shall have the right but not the obligation, during business hours, at Buyer's sole cost and expense and upon reasonable notice to Seller, to inspect the Hydro-Québec Power Resources or the Québec Line, subject to Seller's reasonable site access rules.

(c) <u>Capacity Deficiency</u>. To the extent that a Capacity Deficiency exists on the Commercial Operation Date as permitted by Section 4.4.1(b)(i) of the TSA, then on the Commercial Operation Date, the Contract Maximum Amount shall be automatically reduced commensurate with the Capacity Deficiency, which reduced Contract Maximum Amount shall be stated in a notice from Seller to Buyer, which notice shall be binding absent manifest error; provided that, pursuant to Section 4.4.1(b) of the TSA, in the event that, during the twenty-four (24) months following the TSA Commercial Operation Date, such Capacity Deficiency is reduced or eliminated, then the Contract Maximum Amount shall be automatically increased commensurate with such reduction or elimination, which increased Contract Maximum Amount (i) shall not exceed the MWh/hour set forth in the definition of "Contract Maximum Amount" herein and (ii) shall be stated in a notice from Seller to Buyer, which notice shall be binding absent manifest error.

# 3.4 <u>Conditions Precedent to the Commercial Operation Date</u>.

(a) Seller's obligation to Deliver the Products and Buyer's obligation to pay Seller for such Products commences on the Commercial Operation Date.

(b) The Commercial Operation Date shall occur on a date on which the following conditions precedent have been satisfied:

- (i) Buyer has received Regulatory Approval;
- U.S. Transmission Provider has satisfied the conditions to "Commercial Operation," as set forth in Section 4.3 of the TSA, including the completion of the AC Upgrades (as defined in the TSA) and the CCIS Capacity Upgrades (as defined in the TSA) as contemplated by Section 4.3(c) of the TSA;
- (iii) the Québec Line has been constructed in accordance with Attachment A of the TSA, and is capable of operating at the Design Capability (as defined in the TSA), except as otherwise permitted pursuant to Section 4.4.1(b) of the TSA;
- (iv) Seller has satisfied all requirements in order to provide for unitspecific accounting of Environmental Attributes, enabling the Massachusetts Department of Environmental Protection to accurately account for such Qualified Clean Energy in the state greenhouse gas emissions inventory, created under chapter 298 of the Acts of 2008, and satisfying the requirements of this Agreement;

- (v) Seller (or the party with whom Seller contracts pursuant to Section 3.5(e)) has (x) established all ISO-NE-related accounts and entered into all ISO-NE-related agreements required for the performance of Seller's obligations in connection with the Hydro-Québec Power Resources and this Agreement, which agreements shall be in full force and effect, including the registration of the Hydro-Québec Power Resources in the GIS and (y) satisfied all other requirements of the ISO-NE Rules and the ISO-NE Practices for the Delivery of Products to Buyer at the Delivery Point;
- (vi) no default or Event of Default by Seller shall have occurred and remain uncured;
- (vii) the Hydro-Québec Power Resources and the Québec Line are owned or leased by, and under the care, custody and control of, Seller or its Affiliate;
- (viii) the Hydro-Québec Power Resources are in compliance with all applicable Laws and all Permits for the Delivery of the Products in accordance with this Agreement; and
- (ix) Seller confirms receipt by Hydro-Québec and its Affiliates of all Permits (including all Québec Line Approvals) required for the Delivery of the Products in accordance with this Agreement including an authorization from the National Energy Board of Canada to export electricity to Seller during the Services Term in a quantity at least equal to the proposed transmission transfer capability of the Québec Line, which authorization does not include any conditions or modifications that Hydro-Québec deems to be unacceptable.

### 3.5 <u>Operation of the Hydro-Québec Power Resources</u>.

Compliance With Utility Requirements. To the extent necessary for it to (a) perform its obligations under this Agreement (including without limitation under Sections 3.1, 4.5 and 4.7), Seller shall comply with, and shall cause the Hydro-Québec Power Resources and the Québec Line to comply with: (i) Good Utility Practice; and (ii) all applicable rules, procedures, operating policies, criteria, guidelines and requirements imposed by ISO-NE, any Transmission Provider, any Interconnecting Utility, Régie de l'Énergie and/or any other regional reliability entity with jurisdiction over the Hydro-Québec Power Resources or the Québec Line, including, in each case, all practices, requirements, rules, procedures and standards related to Seller's or its Affiliate's ownership, operation and maintenance of the Hydro-Québec Power Resources and the Québec Line and Seller's performance of its obligations under this Agreement (including, without limiting Seller's obligations in Sections 3.1, 4.5 and 4.7, obligations related to the generation of the Hydro-Québec Power Resources and operation of the Québec Line and the Scheduling, interconnection of the Québec Line with the U.S. Transmission Line at the U.S./Canada border, and transmission of Qualified Clean Energy to the U.S./Canada border, and the transfer of associated Environmental Attributes), whether such requirements were imposed

prior to or after the Effective Date. Seller or its Affiliates shall be solely responsible for registering as the "Generator Owner and Generator Operator" of the Hydro-Québec Power Resources with Régie de l'Énergie and any applicable regional reliability entities, as applicable.

(b) <u>Permits</u>. Seller shall maintain or cause to be maintained in full force and effect all Permits (including without limitation all Québec Line Approvals) necessary for it to perform its obligations under this Agreement, including all Permits necessary to operate and maintain the Hydro-Québec Power Resources and the Québec Line, but excluding any U.S. Transmission Approvals (which, for the avoidance of doubt, shall be the responsibility of U.S. Transmission Provider).

(c) <u>Maintenance and Operation of Hydro-Québec Power Resources</u>. Seller shall, at all times during the Term and as necessary for it to perform its obligations under this Agreement, cause the Hydro-Québec Power Resources and the Québec Line to be maintained and operated in accordance with Good Utility Practice. As between Buyer and Seller, Seller shall bear all costs related thereto. Seller or its Affiliates may contract with other Persons to provide operation and maintenance functions, so long as Seller or its Affiliates maintain overall control over the operation and maintenance of the Hydro-Québec Power Resources throughout the Term.

(d) <u>Interconnection Agreement</u>. Seller shall cause TransÉnergie to comply with the terms and conditions of the Interconnection Agreement to which it is a party and shall be responsible for obtaining the Hydro-Québec Power Resources' interconnection to TransÉnergie's transmission and distribution systems at a level that is capable of satisfying the Capacity Capability Interconnection Standard under the ISO-NE Rules, as required by ISO-NE and contemplated by Section 7.5.

(e) <u>ISO-NE Status</u>. Seller shall, at all times during the Services Term, either: (i) be an ISO-NE "Market Participant" pursuant to the ISO-NE Rules; or (ii) have entered into an agreement with a Market Participant that shall perform all of Seller's ISO-NE-related obligations in connection with the Hydro-Québec Power Resources and this Agreement.

(f) <u>Qualified Clean Energy Generation Units</u>. Seller shall be solely responsible at Seller's cost for demonstrating that the Hydro-Québec Power Resources from which the Products are Delivered are Qualified Clean Energy Generation Units, and for Delivering Baseline Hydroelectric Generation Imports in accordance with Section 4.3(b) and Exhibit H, throughout the Services Term. Seller shall take all actions necessary to register for and maintain participation in the GIS to register, monitor, track, and transfer Environmental Attributes. Seller shall provide such additional information as Buyer may request relating to such qualifications and participation, and the registration, monitoring, tracking and transfer of Environmental Attributes.

(g) <u>Compliance Reporting</u>. Seller shall provide such information as required of the supplier under the GWSA, the CES and the regulations, guidelines and policies issued thereunder. In addition, within thirty (30) days following the end of each calendar quarter, Seller shall provide Buyer information pertaining to emissions, fuel types, and any other information to the extent reasonably requested by Buyer to comply with the CES, the disclosure requirements contained under applicable law and any other such disclosure regulations which may be imposed

upon Buyer during the Term, which information requirements will be provided to Seller by Buyer at least fifteen (15) days before the beginning of the calendar quarter for which the information is required. Seller shall provide Buyer with such additional information to the extent reasonably requested by Buyer to comply with the CES, RPS, APS or any other applicable program under Massachusetts law requiring electric generation with certain Environmental Attributes.

(h) <u>Insurance</u>. Seller shall cause Hydro-Québec to maintain self-insurance with respect to the types of coverage and with such terms and conditions as are reasonable for a power system and transmission line, as applicable, of the type and size of the Hydro-Québec Power Resources and the Québec Line, and as otherwise legally required for the Hydro-Québec Power Resources and the Québec Line and consistent with Good Utility Practice, for the duration of the Term.

(i) <u>Contacts</u>. Each Party shall identify a principal contact or contacts, which contact(s) shall have adequate authority and expertise to make day-to-day decisions with respect to the administration of this Agreement.

(j) <u>Compliance with Law</u>. Without limiting the generality of any other provision of this Agreement, Seller shall be responsible for complying with all applicable requirements of Law, including all applicable rules, procedures, operating policies, criteria, guidelines and requirements imposed by FERC and any other Governmental Entity, whether imposed pursuant to existing Law or procedures or pursuant to changes enacted or implemented during the Term, including all risks of operational and environmental matters relating to the Hydro-Québec Power Resources. Seller shall indemnify Buyer against any and all claims arising out of or related to such environmental matters and against any costs imposed on Buyer as a result of Seller's violation of any applicable Law, or ISO-NE or NERC requirements. For the avoidance of doubt, Seller shall be responsible for procuring, at its expense, all Permits required for the operation of the Hydro-Québec Power Resources and all Québec Line Approvals in compliance with applicable requirements of Law.

(k) <u>FERC Status</u>. Seller shall be responsible for ensuring that it is in compliance with all FERC directives and requirements necessary for Seller to fulfill its obligations under this Agreement. As part of Seller's satisfaction of this responsibility, Seller shall obtain and maintain FERC authorization to sell Products at market-based rates at all times on and after the Commercial Operation Date.

(1) <u>Maintenance</u>. No later than (i) the Commercial Operation Date and (ii) two months prior to the end of each calendar year thereafter during the Term, Seller shall submit to Buyer a schedule of planned maintenance for the following calendar year for the Québec Line. The Parties acknowledge that, throughout the Term, U.S. Transmission Provider and TransÉnergie shall coordinate all planned maintenance for the New Transmission Facilities with ISO-NE, consistent with ISO-NE Rules, and shall promptly provide applicable information concerning scheduled outages, as determined by ISO-NE, to Buyer. To maximize the value of the Products, to the extent possible and consistent with ISO-NE Rules, Seller shall cause maintenance of the Québec Line not to be scheduled during the Winter Period or Summer Period, and shall cause the Québec Line to be operated so as to maximize the Delivery of

Qualified Clean Energy during the hours of anticipated peak load and Energy prices in New England; provided, however, that planned maintenance may be scheduled during such period to the extent the failure to perform such planned maintenance is contrary to operation of the Québec Line in accordance with Good Utility Practice.

(m) <u>Agreement between Seller and U.S. Transmission Provider; Agreement between Buyer and U.S. Transmission Provider</u>.

- Unless Seller obtains the prior written consent of Buyer (such consent not to be unreasonably withheld, conditioned or delayed), Seller shall not: (A) agree to any amendment to Sections 3.3.5, 14.3, 14.6, 14.7, 14.8, and 14.10 of the Seller TSA or (B) agree to an amendment and restatement, supplement, or other modification or amendment of the Seller TSA that adversely and materially affects Buyer's rights under this Agreement. Seller shall provide Buyer a copy of any proposed amendment to the Seller TSA no fewer than ten (10) Business Days prior to the execution thereof.
- (ii) Unless Buyer obtains the prior written consent of Seller (such consent not to be unreasonably withheld, conditioned or delayed), Buyer shall not: (A) agree to any amendment to the sections of the TSA for which Seller is a third-party beneficiary as contemplated by Section 23.8 of the TSA, (B) agree to any amendment and restatement, supplement, or other modification or amendment of the TSA that adversely and materially affects Seller's rights under this Agreement or (C) agree to terminate the TSA by mutual agreement with U.S. Transmission Provider.
- (iii) In the event that Buyer grants a written waiver of any term or condition of the TSA, and such waiver has a material adverse effect on the rights or obligations of Seller under this Agreement, the Parties shall negotiate in good faith to amend or clarify this Agreement or to waive the applicable provision of this Agreement to mitigate the material adverse effect on Seller resulting from such waiver and to restore the Parties to their respective positions had such waiver of the TSA term or condition not been granted.

3.6 <u>Interconnection and Delivery Services</u>. Seller and Buyer acknowledge and agree that U.S. Transmission Provider shall be responsible for all costs associated with any additions, upgrades, reinforcements or other modifications to the New England Transmission System required for the interconnection of the U.S. Transmission Line at the Delivery Point at a level that is capable of satisfying the Capacity Capability Interconnection Standard under the ISO-NE Rules, in each case, as required by ISO-NE and contemplated by Section 7.5, including the costs to complete the Network Upgrades to the extent required under Section 7.5, consistent with all standards and requirements set forth by the FERC, ISO-NE, any other applicable Governmental Entity and the Interconnecting Utility.

# 4. **DELIVERY OF PRODUCTS**

### 4.1 Obligation to Sell and Purchase Products.

Beginning on the Commercial Operation Date and subject to Section (a) 4.1(b), Seller shall sell and Deliver, and Buyer shall purchase and receive, all right, title and interest in and to the Products in accordance with the terms and conditions of this Agreement and as set forth in the Delivery Schedule; provided, however, that the Products Delivered in any hour (or shorter period to the extent that ISO-NE schedules Energy deliveries over a shorter period) shall not exceed the lesser of (x) the total Metered Output generated by the Hydro-Québec Power Resources in such period and (y) the amount of Qualified Clean Energy Delivered to the Delivery Point in such period, in accordance with the terms and conditions of this Agreement. For the avoidance of doubt, (i) if the aggregate amount of Metered Output generated by the Hydro-Québec Power Resources during any hour (or shorter period to the extent that ISO-NE schedules Energy deliveries over a shorter period) is in excess of the amount of Qualified Clean Energy Delivered to the Delivery Point for that period, the Products associated with such excess shall not be considered to be Products for such period for purposes of this Agreement, and (ii) if the amount of Qualified Clean Energy Delivered to the Delivery Point for any hour (or shorter period to the extent that ISO-NE schedules Energy deliveries over a shorter period) exceeds the aggregate amount of Metered Output generated by the Hydro-Québec Power Resources in that period, the Products associated with such excess shall not be considered to be Products for such period for purposes of this Agreement. The aforementioned obligations for Seller to sell and Deliver the Products and for Buyer to purchase and receive the same are firm and not subject to interruption except to the extent caused by Force Majeure, excused under Section 4.2(a) or cured in accordance with Section 4.3(c). The Delivery Schedule is set forth in Exhibit B. All Deliveries of Energy and associated Environmental Attributes must be produced by the Hydro-Ouébec Power Resources that are specified in Exhibit A and Delivered in accordance with this Agreement.

(b) Buyer shall not be obligated to accept or pay for any Environmental Attribute or comparable certificate, credit, attribute or other similar product produced by or associated with the Hydro-Québec Power Resources which does not constitute an Environmental Attribute associated with the specified MWh of generation from Qualified Clean Energy Generation Units.

(c) Except in the case of any default by Buyer, Seller shall not sell, divert, grant, transfer or assign the Products or any right, claim, certificate or other attribute associated with such Products to any Person other than Buyer during the Term. Seller shall not claim or enter into any agreement or arrangement under which such Products can be claimed by any Person other than Buyer. Buyer shall have the exclusive right to resell or convey such Products in its sole discretion.

4.2 <u>Scheduling and Delivery</u>.

(a) During the Services Term and in accordance with Section 4.1, Seller shall Schedule and Deliver Guaranteed Qualified Clean Energy in accordance with this Agreement, all ISO-NE Practices and ISO-NE Rules and other Independent System Operator's rules and practices, as applicable. Seller shall transfer the Qualified Clean Energy or Qualified Shortfall

Energy, as applicable, to Buyer through Internal Bilateral transactions executed through ISO-NE and settled at the Delivery Point or, in the case of Qualified Shortfall Energy and as reasonably agreed from time to time by Buyer and Seller, the NEMASSBOST Hub, in each case, in accordance with all ISO-NE Practices and ISO-NE Rules. Any such Internal Bilateral transactions will specify the hourly delivery of Scheduled Qualified Clean Energy or Qualified Shortfall Energy and will be entered into daily, with any necessary adjustments being made pursuant to ISO-NE settlement protocols, and Seller will not receive any payment associated with a Marginal Loss Revenue Fund allocation in connection with any such Internal Bilateral transactions. Any such Internal Bilateral transactions will be entered in the Day-Ahead Energy Market or, as reasonably agreed from time to time by Buyer, in the Real-Time Energy Market, and consistent with ISO-NE Rules and ISO-NE Practices at the time, and, unless due to the failure of Buyer to confirm any Internal Bilateral transaction submitted by Seller by the applicable scheduling deadline, Buyer shall have no obligation to pay for any Qualified Clean Energy not transferred to Buyer in the Day-Ahead Energy Market or Real-Time Energy Market or for which Buyer is not credited in the ISO-NE Settlement Market System (including, without limitation, as a result of an outage on any electric transmission system). Notwithstanding any other provision of this Agreement, if during the Services Term, the LMP at the Delivery Point is negative, or, in the reasonable opinion of Seller, is likely to become negative, then Seller may deliver to Buyer a written notice stating that such condition has occurred or is likely to occur and the period during which such condition has occurred or is likely to occur. Buyer and Seller hereby agree that in such event Seller shall be under no obligation to schedule or transfer Deliveries of Qualified Clean Energy to the Delivery Point during such period. Seller shall provide Buyer with detailed information, including but not limited to, the start and stop times of such periods of curtailment under this Section 4.2(a) as well as the quantity curtailed, for all such periods of non-delivery during the preceding calendar month, which information shall be provided prior to Seller's delivery of the invoice to Buyer.

(b) The Parties agree to use Commercially Reasonable Efforts to comply with all applicable ISO-NE Rules and ISO-NE Practices in connection with the Scheduling and Delivery of Energy hereunder. Without limiting the generality of the foregoing, Seller shall submit an Internal Bilateral transaction for the Scheduled Qualified Clean Energy, or Qualified Shortfall Energy, as the case may be, by the applicable scheduling deadline and Buyer shall confirm the Internal Bilateral transaction submitted by Seller by the applicable scheduling deadline. Penalties or similar charges assessed by a Transmission Provider, ISO-NE or other Independent System Operators, as applicable, and caused by Seller's noncompliance with the Scheduling obligations set forth in this Section 4.2 shall be the responsibility of Seller.

(c) Without limiting the generality of this Section 4.2, Seller or the party with whom Seller contracts pursuant to Section 3.5(e) shall at all times during the Services Term be designated as the "Lead Market Participant" (or any successor designation) for the Hydro-Québec Power Resources and shall be solely responsible for any obligations and liabilities imposed by ISO-NE, other Independent System Operators, as applicable, or under the ISO-NE Rules and ISO-NE Practices with respect to the Delivery of Qualified Clean Energy up to the Delivery Point, or Qualified Shortfall Energy up to the designated delivery point within the New England Control Area, as applicable, including all charges, penalties, financial assurance obligations, losses, ancillary service charges, and other ISO-NE, other Independent System Operator, as applicable, or applicable system costs, charges or revenues, in each case, associated

with the Delivery of Qualified Clean Energy up to the Delivery Point, or Qualified Shortfall Energy up to the designated delivery point within the New England Control Area, as applicable. To the extent Buyer incurs such costs, charges, penalties or losses which are the responsibility of Seller (including amounts not credited to Buyer as described in Section 4.2(a)), Seller shall reimburse Buyer for the same.

### 4.3 <u>Failure of Seller to Deliver Products</u>.

(a) In the event that Seller fails to satisfy any of its obligations to Deliver any of the Products or any portion of the Products hereunder in accordance with Section 4.1, Section 4.2 and Section 4.7, and to the extent such failure is not caused by Force Majeure or otherwise excused under the express terms of this Agreement (a "Delivery Shortfall") and Seller has not cured such failure as provided, and to the extent permitted, in Section 4.3(c) (an "Uncured Delivery Shortfall"), Seller shall pay Buyer an amount for such Uncured Delivery Shortfall equal to the Cover Damages. Such payment shall be due no later than the date for Buyer's payment for the next applicable month as set forth in Section 5.2 hereof. Each Party agrees and acknowledges that (i) the damages that Buyer would incur due to an Uncured Delivery Shortfall would be difficult or impossible to predict with certainty, and (ii) it is impractical and difficult to assess actual damages in the circumstances stated, and therefore the Cover Damages.

(b) The Parties' rights and obligations in respect of Baseline Hydroelectric Generation Imports are as set forth in Exhibit H.

(c) Notwithstanding anything to the contrary in Section 4.1(a) or this Section 4.3, in the event of a Delivery Shortfall caused solely by or arising solely from (x) a Non-Excused Outage (as defined in the TSA) (a "<u>TSA Delivery Shortfall</u>") and/or (y) an outage or reduction in the availability of the Québec Line as a result of a physical condition that affects the transfer capability of the Québec Line (each a "<u>Curable Delivery Shortfall</u>"), Seller shall have the right to Deliver Qualified Shortfall Energy and associated Environmental Attributes, in addition to the Guaranteed Qualified Clean Energy, in order to cure all or any portion of the amount of such Curable Delivery Shortfall (such amount, as expressed in MWh, the "<u>Shortfall</u>"), as follows:

- (i) Seller may Deliver the Shortfall Cure Amount in the same Contract Year in which the Curable Delivery Shortfall occurred or in the immediately succeeding Contract Year (the "<u>Shortfall Cure</u> <u>Period</u>").
- (ii) Unless and until the TSA is terminated, if a Curable Delivery Shortfall is caused by or arises from a TSA Delivery Shortfall and Seller cures such TSA Delivery Shortfall, in whole or in part, through the Delivery of Qualified Shortfall Energy during the Shortfall Cure Period, Buyer shall increase the Price to be paid for such Qualified Shortfall Energy by the amount that would have been due from Buyer to the U.S. Transmission Provider under the TSA with respect to such TSA Delivery Shortfall (such TSA amounts to be converted to a \$/MWh basis and limited to the cured

TSA Delivery Shortfall) had the Non-Excused Outage (as defined in the TSA) giving rise to such TSA Delivery Shortfall not occurred. If the TSA is terminated, the Price for Qualified Shortfall Energy Delivered after such TSA termination shall not include any amount that would have been due from Buyer to the U.S. Transmission Provider under the TSA.

- (iii) The total amount of Curable Delivery Shortfalls caused by or arising from any event or condition other than a Non-Excused Outage (as defined in the TSA) (each a "<u>Non-TSA Delivery</u> <u>Shortfall</u>") in any Contract Year that may be cured through Deliveries of Qualified Shortfall Energy shall not exceed the amount, expressed in MWh, equal to (A) the Contract Maximum Amount, multiplied by (B) 720. No such limitation shall apply to Deliveries of Qualified Shortfall Energy to cure TSA Delivery Shortfalls.
- (iv) Any Curable Delivery Shortfall occurring during the final Contract Year may only be cured through Deliveries of Qualified Shortfall Energy during such final Contract Year or through the payment of Cover Damages.
- (v) As set forth in Exhibit H, any Qualified Shortfall Energy Delivered under this Agreement shall not be taken into account in determining whether Seller has satisfied its obligation to deliver Baseline Hydroelectric Generation Imports for the applicable period set forth in Exhibit H.
- (vi) In the event that a Curable Delivery Shortfall occurs during the Winter Period or Summer Period, the Shortfall Cure Amount of Qualified Shortfall Energy with respect to such Curable Delivery Shortfall may only be delivered during the corresponding period of the Shortfall Cure Period. In the event that a Curable Delivery Shortfall occurs during On-Peak hours, the Shortfall Cure Amount of Qualified Shortfall Energy with respect to such Curable Delivery Shortfall may only be delivered during On-Peak hours during the Shortfall Cure Period. For example, if a Curable Delivery Shortfall occurs during On-Peak hours during the Winter Period, the Shortfall Cure Amount of Qualified Shortfall Energy with respect to that Curable Delivery Shortfall must be delivered during On-Peak hours in the Winter Period during the Shortfall Cure Period.
- (vii) If, at the end of any Contract Year, (x) the amount of Non-TSA Delivery Shortfalls in such Contract Year exceeds the amount, expressed in MWh, equal to the Contract Maximum Amount, multiplied by 720, or (y) there are any other Uncured Delivery

Shortfalls for the Shortfall Cure Period ending at the end of such Contract Year for which Cover Damages have not previously been paid, then Seller shall pay Buyer an amount for such Uncured Delivery Shortfalls equal to the Cover Damages.

- (viii) At the end of each Contract Year, Buyer shall calculate (A) the weighted average LMP at the Delivery Point for the time period in which each Curable Delivery Shortfall occurred during such Contract Year (the "Delivery Shortfall LMP") and (B) the weighted average LMP at the Delivery Point for the time period in which each Qualified Shortfall Energy Delivered occurred during such Contract Year (the "Qualified Shortfall Energy LMP"). If the Delivery Shortfall LMP is greater than the Qualified Shortfall Energy LMP for a Contract Year, then Seller shall pay to Buyer an amount equal to the product of such difference multiplied by the Oualified Shortfall Energy Delivered during such Contract Year. If the Qualified Shortfall Energy LMP is greater than the Delivery Shortfall LMP for a Contract Year, then Buyer shall pay to Seller an amount equal to the product of such difference multiplied by the Qualified Shortfall Energy Delivered during such Contract Year. If the Delivery Shortfall LMP is equal to the Qualified Shortfall Energy LMP, then no payment shall be required.
- (ix) No Qualified Shortfall Energy shall be delivered over the U.S. Transmission Line to the Delivery Point in any hour in which Seller is excused from performing its obligations under this Agreement due to a Force Majeure unless Seller is also Delivering the entire amount of the Guaranteed Qualified Clean Energy in that hour.

(d) Notwithstanding any other provision of this Agreement, Seller shall have the right, but not the obligation, to Deliver the Shortfall Cure Amount pursuant to Section 4.3(a) of this Agreement but shall have no obligation to pay Cover Damages with respect to the portion of any TSA Delivery Shortfall that occurs during the period (the "Limited Suspension Period") when (i) Buyer has a right to terminate the TSA due to an event of default under Section 14.2(e) of the TSA and (ii) Buyer has not exercised the right to terminate the TSA as a result of that event of default. In the event that either (x) the event of default under Section 14.2(e) of the TSA is cured pursuant to the terms of the TSA or (y) Buyer terminates the TSA as a result of such event of default, Seller's right to Deliver the Shortfall Cure Amount pursuant to Section 4.3(a) of this Agreement shall continue as contemplated by Section 4.3(c), and Seller's obligation to pay Cover Damages with respect to any TSA Delivery Shortfall that occurs after such Limited Suspension Period and that Seller has not cured as provided, and to the extent permitted, in Section 4.3(c) shall resume immediately upon the curing of such event of default under the TSA or immediately upon termination of the TSA. Nothing set forth in Section 4.3(c) shall affect the rights or obligations of the Parties with respect to an Event of Default under Section 9.2(f).

# 4.4 Failure by Buyer to Accept Delivery of Products; Default under TSA.

(a) If Buyer fails to accept all or part of any of the Products to be purchased by Buyer hereunder, and such failure to accept (i) is not the result of Reliability Curtailment or (ii) is not otherwise excused under the terms of this Agreement (a "<u>Rejected Purchase</u>"), then Buyer shall pay Seller, on the date payment would otherwise be due in respect of the month in which the failure occurred, an amount for such Rejected Purchase equal to the Resale Damages. Each Party agrees and acknowledges that (A) the damages that Seller would incur due to a Rejected Purchase would be difficult or impossible to predict with certainty, and (B) it is impractical and difficult to assess actual damages in the circumstances stated, and therefore the Resale Damages as agreed to by the Parties and set forth herein is a fair and reasonable calculation of such damages.

(b) In the event that, due to a Buyer breach under the TSA, Seller is unable to Deliver any of the Products in accordance with the obligations under this Agreement, including due to a loss of applicable transmission rights, such failure shall not be deemed a Delivery Shortfall and Buyer shall pay Seller, on the date payment would otherwise be due in respect of each month in which the failure has occurred, an amount equal to the Resale Damages. Each Party agrees and acknowledges that (i) the damages that Seller would incur due to a Buyer default under the TSA would be difficult or impossible to predict with certainty, and (ii) it is impractical and difficult to assess actual damages in the circumstances stated, and therefore the Resale Damages as agreed to by the Parties and set forth herein is a fair and reasonable calculation of such damages.

# 4.5 <u>Delivery Point</u>.

(a) All Qualified Clean Energy shall be Delivered hereunder by Seller to Buyer at the Delivery Point. Except as set forth in Article 20, Seller shall be responsible for delivering its Qualified Clean Energy to the Delivery Point, and all Deliveries of Qualified Shortfall Energy as provided herein, in each case, consistent with all standards and requirements set forth by the FERC, ISO-NE, other Independent System Operators, as applicable, and any other applicable Governmental Entity or applicable tariff.

(b) Except as set forth in Article 20, Seller shall be responsible for all applicable charges associated with transmission over the Québec Line, interconnection of the Hydro-Québec Power Resources to TransÉnergie's transmission and distribution systems, service and delivery charges in respect of the Delivery of Qualified Clean Energy and any Qualified Shortfall Energy up to the Delivery Point, including all related ISO-NE or other Independent System Operator administrative fees, uplift, and socialized charges, and all other charges or revenues up to the Delivery Point in connection with Seller's obligations hereunder, including without limitation the Delivery of Qualified Clean Energy to and at the Delivery Point. Seller shall indemnify and hold harmless Buyer for any such charges or fees imposed upon Buyer by operation of ISO-NE Rules or otherwise in connection with Seller's performance of its obligations hereunder.

# <u>4.6</u> <u>Metering</u>.

(a) <u>Metering</u>. All electric metering associated with the Hydro-Québec Power Resources, including each Hydro-Québec Power Resource's meter and any other real-time meters, billing meters and back-up meters (collectively, the "<u>Meters</u>"), shall be installed, operated, maintained and tested at Seller's expense in accordance with Good Utility Practice and any applicable requirements and standards issued by NERC, the Interconnecting Utility and the GIS Operating Rules. All metering associated with the Hydro-Québec Power Resources shall be sufficient to permit the Metered Output to be tracked in the GIS to ensure unit-specific accounting of the Delivery of Qualified Clean Energy or Qualified Shortfall Energy to enable the Massachusetts Department of Environmental Protection to accurately account for such Qualified Clean Energy or Qualified Shortfall Energy in the state greenhouse gas emissions inventory, created under chapter 298 of the Acts of 2008.

(b) <u>Measurements</u>. Readings of the Meters at each Hydro-Québec Power Resource by the Interconnecting Utility in whose territory that Hydro-Québec Power Resource is located (or an independent Person mutually acceptable to the Parties) shall be conclusive as to the amount of Metered Output.

(c) <u>Inspection, Testing and Calibration</u>. Buyer shall have the right to inspect and test (at its own expense) any of the Meters at the Hydro-Québec Power Resources at reasonable times and upon reasonable notice from Buyer to Seller. Buyer and Seller shall negotiate in good faith and in a Commercially Reasonable Manner to agree upon procedures pursuant to which Seller shall test the Meters from time to time. Buyer shall have the right to have a representative present during any testing or calibration of the Meters at the Hydro-Québec Power Resources by Seller or its Affiliate. Seller shall provide Buyer with timely notice of any such testing or calibration.

(d) <u>Audit of Meters</u>. Buyer shall have access to the Meters and the right to audit all information and test data related to such Meters.

(e) <u>Notice of Malfunction</u>. Seller shall provide Buyer with prompt notice of any malfunction or other failure of the Meters or other telemetry equipment necessary to accurately report the quantity of Metered Output. If any Meter is found to be inaccurate by more than the level of inaccuracy that is allowed under the GIS Operating Rules at the time with respect to the Hydro-Québec Power Resources, the meter readings shall be adjusted as far back as can reasonably be ascertained, but in no event shall such period exceed six (6) months from the date that such inaccuracy was discovered, and any adjustment shall be reflected in the next invoice provided by Seller to Buyer hereunder.

# 4.7 <u>Environmental Attributes</u>.

(a) Seller shall transfer to Buyer all of the right, title and interest in and to the Environmental Attributes, including any and all Certificates, associated with Qualified Clean Energy or any Qualified Shortfall Energy, as applicable, from the Hydro-Québec Power Resources Delivered to Buyer under this Agreement and paid for by Buyer during the Services Term.

(b) All Energy and associated Environmental Attributes Delivered to Buyer under this Agreement shall meet the requirements set forth in the definitions for Qualified Clean Energy or Qualified Shortfall Energy, as applicable, and the requirements of the CES as in effect at the time of Delivery (subject to the ensuing sentence in this Section 4.7(b)), and Seller's failure to satisfy such requirements shall constitute an Event of Default pursuant to Section 9.1(c). In the event that the CES is repealed, replaced, amended or otherwise modified during the Term, Seller shall be required to maintain unit-specific transfer, tracking and reporting of Environmental Attributes but shall otherwise only be required to use Commercially Reasonable Efforts to cause the Energy and associated Environmental Attributes Delivered to Buyer under this Agreement to comply with any additional obligations resulting from the CES as amended or otherwise modified during the Term, notwithstanding anything to the contrary in this Agreement. The amount of Certificates transferred from Seller to Buyer under this Agreement for any hour (or shorter period to the extent that ISO-NE schedules Energy deliveries over a shorter period) will be the equivalent of the lesser of the aggregate amount of Metered Output or the Qualified Clean Energy or Qualified Shortfall Energy, as applicable, Delivered during that hour (or such shorter period).

Seller shall comply with all GIS Operating Rules including, without (c) limitation, such rules relating to the creation, tracking, recording and transfer of all Environmental Attributes associated with Qualified Clean Energy or Qualified Shortfall Energy, as applicable, to be purchased by Buyer under this Agreement, which compliance shall be at Seller's sole cost. In addition, at Buyer's request, Seller shall register with and comply with the rules and requirements of another system that tracks, monetizes or otherwise creates or enhances value for Environmental Attributes, which compliance shall be at Seller's sole cost; provided that Seller shall neither be required to incur any costs associated with any registration and compliance with the rules and requirements of another system other than administrative (including, without limitation, legal and accounting), filing and reporting costs associated with any such registration and compliance or costs associated with any such registration and compliance that would have otherwise been incurred by Seller irrespective of such registration and compliance, nor shall Seller in any event be required to incur any costs associated with any registration and compliance with the rules and requirements of another system that would not have otherwise been incurred by Seller irrespective of such registration and compliance if such costs would reasonably be expected to exceed \$5,000,000 in the aggregate during the Term.

(d) Seller shall transfer to Buyer the Environmental Attributes associated with the Qualified Clean Energy or Qualified Shortfall Energy, as applicable, Delivered hereunder by means of an irrevocable Forward Certificate Transfer (as defined in the GIS Operating Rules) within fifteen (15) days after the end of the calendar month in which such Qualified Clean Energy or Qualified Shortfall Energy, as applicable, was generated, which Forward Certificate Transfer shall be denoted in the GIS as not being capable of rescission by Seller, and Buyer shall accept such Forward Certificate Transfer as provided in the GIS Operating Rules. No payment shall be due to Seller for any Environmental Attributes until either (x) the Certificates are actually deposited in Buyer's GIS account or a GIS account designated by Buyer to Seller in writing, or (y) (i) Buyer and Seller have entered into the irrevocable Forward Certificate Transfer (as defined in the GIS Operating Rules) of the Environmental Attributes and any Certificates to be Delivered to Buyer in the GIS, and (ii) the Energy with which such Certificates are associated has been Delivered to Buyer. (e) To the extent that the Environmental Attributes are tracked in any system other than the GIS, Seller shall cause all credits, certificates, benefits, reductions, offsets and allowances created or accounted for in such other system to be (x) retired or to otherwise cease to be available for use in such other system, or (y) transferred to Buyer's account in such other tracking system so long as such a transfer would not affect the treatment of such Environmental Attributes in the GIS or under applicable Law.

(f) The Parties intend for the transactions entered into hereunder to be physically settled, meaning that such Environmental Attributes are to be Delivered in the GIS account of Buyer or its designee as set forth in this Section 4.7.

# 5. PRICE AND PAYMENTS FOR PRODUCTS

5.1 <u>Price for Products</u>. The price for the Delivered Products shall be as specified in Exhibit D.

# 5.2 <u>Payment and Netting</u>.

(a) <u>Billing Period</u>. The calendar month shall be the standard period for all charges and payments under this Agreement. On or before the fifteenth (15th) day following the end of each month, Seller shall render to Buyer an invoice for the payment obligations incurred hereunder during the preceding month, based on the Qualified Clean Energy and Qualified Shortfall Energy, as applicable, Delivered in the preceding month, and associated Environmental Attributes deposited in Buyer's GIS account or a GIS account designated by Buyer to Seller in writing in the preceding month. Such invoice shall contain supporting detail for all charges reflected on the invoice, and Seller shall provide Buyer with additional supporting documentation and information as Buyer may reasonably request.

(b) <u>Timeliness of Payment</u>. All undisputed charges under this Agreement shall be due and payable in accordance with each Party's invoice instructions on or before the later of (x) fifteen (15) days from receipt of the applicable invoice or (y) the last day of the calendar month in which the applicable invoice was received (or in either event the next Business Day if such day is not a Business Day). Each Party shall make payments by electronic funds transfer, or by other mutually agreeable method(s), to the account designated by the other Party. Any undisputed amounts not paid by the due date shall be deemed delinquent and shall accrue interest at the Late Payment Rate, such interest to be calculated from and including the due date to but excluding the date the delinquent amount is paid in full.

# (c) <u>Disputes and Adjustments of Invoices</u>.

(i) All invoices rendered under this Agreement shall be subject to adjustment after the end of each month in order to true-up charges based on changes resulting from ISO-NE billing statements or revisions, if any, to previous ISO-NE billing statements. If ISO-NE resettles any invoice which relates to the Products sold under this Agreement and (A) any charges thereunder are the responsibility of the other Party under this Agreement or (B) any credits issued thereunder would be due to the other Party under this Agreement, then the Party receiving the invoice from ISO-NE shall, in the case of (A) above, invoice the other Party or, in the case of (B) above, pay the amount due to the other Party. Any invoices issued or amounts due pursuant to this Section shall be invoiced or paid, as applicable, as provided in this Section 5.2.

Within twelve (12) months of the issuance of an invoice, Seller (ii) shall adjust any invoice for any arithmetic or computational error, and shall provide documentation and information supporting such adjustment to Buyer. Within twelve (12) months of the receipt of an invoice (or an adjusted invoice), Buyer may dispute any charges on that invoice. In the event of such a dispute, Buyer shall give notice to Seller and shall state the basis for the dispute. Payment of the disputed amount shall not be required until the dispute is resolved. Upon resolution of the dispute, any required payment or refund shall be made within ten (10) days after such resolution along with interest accrued at the Late Payment Rate from and including the due date (or, in the case of a refund, the payment date) but excluding the date paid. Any claim for additional payment is waived unless Seller issues an adjusted invoice within twelve (12) months of issuance of the original invoice. Anv dispute of charges is waived unless Buyer provides notice of the dispute to Seller within twelve (12) months of receipt of the invoice (or adjusted invoice) including such charges.

(d) <u>Netting of Payments</u>. The Parties hereby agree that they may discharge mutual debts and payment obligations due and owing to each other under this Agreement (including, without limitation, any amount due to Buyer under Exhibit H) on the same date through netting of such monetary obligations, in which case all amounts owed by each Party to the other Party for the purchase and sale of Products during the monthly billing period under this Agreement, including any related damages calculated pursuant to this Agreement, interest, and payments or credits, may be netted so that only the excess amount remaining due shall be paid by the Party who owes it. If no mutual debts or payment obligations exist and only one Party owes a debt or obligation to the other during the monthly billing period, such Party shall pay such sum in full when due. The Parties agree to provide each other with reasonable detail of such net payment or net payment request.

5.3 <u>Interest on Late Payment or Refund</u>. A late payment charge shall accrue on any late payment or refund as specified above at the lesser of (a) the prime rate specified in the "Money & Investing" section of *The Wall Street Journal* (or, if such rate is not published therein, in a successor index mutually selected by the Parties) plus one percent (1%), and (b) the maximum rate permitted by applicable Law in transactions involving entities having the same characteristics as the Parties (the "Late Payment Rate").

# 5.4 <u>Taxes, Fees and Levies</u>.

Seller or its Affiliates shall be obligated to pay all present and future taxes, (a) governmental tariffs, import fees and taxes, and any other fees and levies, imposed on or associated with the Hydro-Québec Power Resources or delivery or sale of the Products on or before the Delivery Point ("Seller's Taxes"). Buyer shall be obligated to pay all present and future taxes, fees and levies, imposed on or associated with such Products after the Delivery Point or imposed on or associated with the purchase of such Products by Buyer (other than ad valorem, franchise or income taxes which are related to the sale of the Products and are, therefore, the responsibility of Seller) ("Buyer's Taxes"). In the event that Seller shall be required by law or regulation to remit or pay any Buyer's Taxes, at Seller's election, Buyer shall reimburse Seller for such payment or Seller may deduct the amount of any such Buyer's Taxes from any amounts due to Buyer hereunder. In the event that Buyer shall be required by law or regulation to remit or pay any Seller's Taxes, at Buyer's election, Seller shall reimburse Buyer for such payment or Buyer may deduct the amount of any such Seller's Taxes from the amount due to Seller under Section 5.2. Buyer shall have the right to all credits, deductions and other benefits associated with taxes paid by Buyer or reimbursed to Seller by Buyer as described herein. Nothing shall obligate or cause a Party to pay or be liable to pay any taxes, fees and levies for which it is exempt under law.

(b) Seller shall bear all risks, financial and otherwise, throughout the Term, associated with the eligibility to receive any federal or state tax credits, to qualify for accelerated depreciation for Seller's accounting, reporting or tax purposes, or to receive any other favorable tax or accounting right or benefit, or any grant or subsidy from a Governmental Entity or other Person. Seller's obligations under this Agreement shall be effective regardless of whether the Hydro-Québec Power Resources are eligible for or receives, or the transactions contemplated under this Agreement are eligible for or receive, any federal or state tax credits, grants or other subsidies or any particular accounting, reporting or tax treatment.

# 6. SECURITY FOR PERFORMANCE

6.1 <u>Grant of Security Interest</u>. Subject to the terms and conditions of this Agreement, Seller hereby pledges to Buyer as security for all outstanding obligations under this Agreement (other than indemnification obligations surviving the expiration of the Term) and any other documents, instruments or agreements executed in connection therewith (collectively, the "<u>Obligations</u>"), and grants to Buyer a first priority continuing security interest, lien on, and right of set-off against all Posted Collateral delivered to or received by Buyer hereunder. Upon the return by Buyer to Seller of any Posted Collateral, the security interest and lien granted hereunder on that Posted Collateral will be released immediately and, to the extent possible, without further action by either Party.

# 6.2 <u>Seller's Support</u>.

(a) Within five (5) Business Days following the receipt of the Regulatory Approval, Seller shall cause Seller Guarantor to deliver to Buyer a guaranty by Seller Guarantor of Seller's payment obligations under this Agreement substantially in the form of Exhibit G (the "<u>Seller Guaranty</u>"). Seller shall be required to cause Seller Guarantor to maintain the Seller Guaranty thereafter continuing through and including the date that all of Seller's obligations

under this Agreement have been satisfied. Such Seller Guaranty shall be capped at the amount of the Unsecured Credit Limit under Section 6.7.

(b) Seller shall be required to post Credit Support in the amount of \$20,000.00 per MWh/hour of the Contract Maximum Amount (e.g., \$21,800,000 in the event that the Contract Maximum Amount is 1,090 MWh/hour) to secure Seller's Obligations ("<u>Fixed Credit</u> <u>Support</u>"). Fifty percent (50%) of the Fixed Credit Support shall be provided to Buyer on the Effective Date, and the remaining fifty percent (50%) of the Fixed Credit Support shall be provided to Buyer within fifteen (15) days after the receipt of the Regulatory Approval.

(c) At any time following the receipt of the Regulatory Approval, if Buyer's Exposure exceeds Seller Guarantor's Unsecured Credit Limit pursuant to Section 6.7, Buyer may demand that Seller provide additional Credit Support in an amount equal to the amount by which the Unsecured Credit Limit is exceeded, rounded up to the nearest \$250,000.

(d) If at any time following the receipt of the Regulatory Approval there shall occur a Downgrade Event, Seller Guarantor's Unsecured Credit Limit shall automatically be zero. Seller shall then provide Credit Support in an amount equal to Buyer's Exposure.

(e) The Credit Support Delivery Amount, as defined below, will be rounded up, and the Credit Support Return Amount, as defined below, will be rounded down, in each case, to the nearest integral multiple of \$10,000 ("<u>Rounding Amount</u>"), except as otherwise provided herein.

The following items will qualify as "<u>Credit Support</u>" hereunder in the amount noted under "Valuation Percentage": "<u>Valuation Percentage</u>"

100%

(B) Letters of Credit	100% unless either (i) a Letter of
	Credit Default shall have occurred
	and be continuing with respect to
	such Letter of Credit, or (ii) twenty
	(20) or fewer Business Days remain
	prior to the expiration of such Letter
	of Credit, in which cases the
	Valuation Percentage shall be 0%.

(f) All calculations with respect to Credit Support shall be made by the Valuation Agent as of the Valuation Time on the Valuation Date.

6.3 <u>Delivery of Credit Support</u>.

On any Business Day during the Services Term on which (a) no Event of Default has occurred and is continuing with respect to Buyer, and (b) no termination date has occurred or has been designated as a result of an Event of Default with respect to Buyer for which there exist any unsatisfied payment obligations with respect to Buyer, then Buyer may request, by written notice, that Seller Transfer to Buyer, or cause to be Transferred to Buyer, Credit Support for the benefit of Buyer, having a Value of at least the Collateral Requirement ("<u>Credit</u> <u>Support Delivery Amount</u>"). Such Credit Support shall be delivered to Buyer on the next Business Day if the request is received no later than 1:00 p.m. Eastern Prevailing Time on the Business Day that the request for Credit Support was made; otherwise Credit Support is due by the close of business on the second Business Day after the request is received.

### 6.4 <u>Reduction and Substitution of Posted Collateral</u>.

On any Business Day during the Services Term on which (a) no Event of Default has occurred and is continuing with respect to Seller, (b) no termination date has occurred or has been designated as a result of an Event of Default with respect to Seller for which there exist any unsatisfied payment Obligations, and (c) the Posted Collateral posted by Seller exceeds the required Credit Support (rounding downwards for any fractional amount to the next interval of the Rounding Amount), then Seller may, at its sole cost, request that Buyer return Credit Support in the amount of such difference ("Credit Support Return Amount") and Buyer shall be obligated to do so. Such Posted Collateral shall be returned to Seller by the close of business on the second Business Day after Buyer's receipt of such request. The Parties agree that if Seller has posted more than one type of Credit Support to Buyer, Seller can, in its sole discretion, select the type of Credit Support for Buyer to return; provided, however, that Buyer shall not be required to post additional Credit Support if, immediately after such return, Seller would be required to post additional Credit Support pursuant to the calculation of Credit Support.

# 6.5 <u>Administration of Posted Collateral</u>.

(a) <u>Cash</u>. Posted Collateral provided in the form of Cash to Buyer hereunder shall be subject to the following provisions:

(i) So long as no Event of Default has occurred and is continuing with respect to Buyer, Buyer will be entitled to either hold Cash or to appoint an agent that is a Qualified Institution (a "Custodian") to hold Cash for Buyer. In the event that an Event of Default has occurred and is continuing with respect to Buyer, then the provisions of Section 6.5(a)(ii) shall not apply with respect to Buyer and Cash shall be held in a Qualified Institution in accordance with the provisions of Section 6.5(a)(iii)(B). Upon notice by Buyer to Seller of the appointment of a Custodian, Seller's Obligations to make any Transfer will be discharged by making the Transfer to that Custodian. The holding of Cash by a Custodian will be deemed to be the holding of Cash by Buyer for which the Custodian is acting. If Buyer or its Custodian fails to satisfy any conditions for holding Cash as set forth above, or if Buyer is not entitled to hold Cash at any time, then Buyer will Transfer, or cause its Custodian to Transfer, the Cash to a Oualified Institution and the Cash shall be maintained in

accordance with Section 6.5(a)(iii)(B). Except as set forth in Section 6.5(c), Buyer will be liable for the acts or omissions of the Custodian to the same extent that Buyer would be held liable for its own acts or omissions.

- (ii) Notwithstanding the provisions of applicable Law, if no Event of Default has occurred and is continuing with respect to Buyer and no termination date has occurred or been designated as a result of an Event of Default with respect to Buyer for which there exists any unsatisfied payment obligations with respect to Buyer, then Buyer shall have the right to sell, pledge, rehypothecate, assign, invest, use, comingle or otherwise use in its business any Cash that it holds as Posted Collateral hereunder, free from any claim or right of any nature whatsoever of Seller, including any equity or right of redemption by Seller.
- (iii) Notwithstanding Section 6.5(a)(ii), if neither Buyer nor the Custodian is eligible to hold Cash pursuant to Section 6.5(a)(i) then:

(A) the provisions of Section 6.5(a)(ii) will not apply with respect to Buyer; and

Buyer shall be required to Transfer (or cause to be **(B)** Transferred) not later than the close of business within five (5) Business Days following the beginning of such ineligibility all Cash in its possession or held on its behalf to a Qualified Institution to be held in a segregated, safekeeping or custody account (the "Collateral Account") within such Qualified Institution with the title of the account indicating that the property contained therein is being held as Cash for Buyer. The Qualified Institution shall serve as Custodian with respect to the Cash in the Collateral Account, and shall hold such Cash in accordance with the terms of this Article 6 and for the security interest of Buyer and execute such account control agreements as are necessary or applicable to perfect the security interest of Seller therein pursuant to Section 9-314 of the Uniform Commercial Code or otherwise, and subject to such security interest, for the ownership and benefit of Seller. The Qualified Institution holding the Cash will invest and reinvest or procure the investment and reinvestment of the Cash in accordance with the written instructions of Buyer, subject to the approval of such instructions by Seller (which approval shall not be unreasonably withheld). Buyer shall have no responsibility for any losses resulting from any investment or reinvestment effected in accordance with Seller's approval.

(iv) So long as no Event of Default with respect to Seller has occurred and is continuing, and no termination date has occurred or been designated for which any unsatisfied payment obligations of Seller exist as the result of an Event of Default with respect to Seller, in the event that Buyer or its Custodian is holding Cash, Buyer will Transfer (or cause to be Transferred) to Seller, in lieu of any interest or other amounts paid or deemed to have been paid with respect to such Cash (all of which shall be retained by Buyer), the Interest Amount. Interest on Cash shall accrue at the Collateral Interest Rate. Interest accrued during the previous month shall be paid by Buyer to Seller on the 3rd Business Day of each calendar month and on any Business Day that posted Credit Support in the form of Cash is returned to Seller, but solely to the extent that, after making such payment, the amount of the Posted Collateral will be at least equal to the required Credit Support. On or after the occurrence of an Event of Default with respect to Seller or a termination date as a result of an Event of Default with respect to Seller, Buyer or its Custodian shall retain any such Interest Amount as additional Posted Collateral hereunder until the Obligations of Seller under the Agreement have been satisfied in the case of a termination date or for so long as such Event of Default is continuing in the case of an Event of Default.

(b) Buyer's Rights and Remedies. If at any time an Event of Default with respect to Seller has occurred and is continuing, then, unless Seller has paid in full all of its Obligations that are then due, including those under Section 9.3(b), Buyer may exercise one or more of the following rights and remedies: (i) all rights and remedies available to a secured party under applicable Law with respect to Posted Collateral held by Buyer and the Seller Guaranty, (ii) the right to set-off any amounts payable by Seller with respect to any Obligations against any Posted Collateral or the cash equivalent of any Posted Collateral held by Buyer, or (iii) the right to liquidate any Posted Collateral held by Buyer or draw on the Seller Guaranty up to the aggregate amount of the Obligations that are due at the time of such liquidation or drawing and to apply the proceeds of such liquidation of the Posted Collateral and such draw on the Seller Guaranty to any such amounts payable to Buyer in such order as Buyer may elect. For purposes of this Section 6.5, Buyer may draw on the Posted Collateral or the Seller Guaranty from time to time up to the aggregate amount of the Obligations that are due at the time of such liquidation or drawing. Cash proceeds that are not applied to the Obligations shall be maintained in accordance with the terms of this Article 6. Seller shall remain liable for amounts due and owing to Buyer that remain unpaid after the application of Posted Collateral and draws on the Seller Guaranty, pursuant to this Section 6.5.

(c) <u>Letters of Credit</u>. Credit Support provided in the form of a Letter of Credit shall be subject to the following provisions.

(i) As one method of providing increased Credit Support, Seller may increase the amount of an outstanding Letter of Credit or establish one or more additional Letters of Credit.

- (ii) Upon the occurrence of a Letter of Credit Default, Seller agrees to Transfer to Buyer either a substitute Letter of Credit or Cash, in each case, on or before the first (1<sup>st</sup>) Business Day after the occurrence thereof (or the third (3<sup>rd</sup>) Business Day after the occurrence thereof if only clause (a) under the definition of Letter of Credit Default applies).
- (iii) Notwithstanding Sections 6.3 and 6.4, (1) Buyer need not return a Letter of Credit unless the entire principal amount is required to be returned, (2) Buyer shall consent to a reduction of the principal amount of a Letter of Credit to the extent that a Credit Support Delivery Amount would not be created thereby (as of the time of the request or as of the last time the Credit Support Delivery Amount was determined), and (3) if there is more than one form of Posted Collateral when a Credit Support Return Amount is to be Transferred, Buyer may elect which to Transfer.

(d) <u>Care of Posted Collateral</u>. Buyer shall exercise reasonable care to assure the safe custody of all Posted Collateral to the extent required by applicable Law, and in any event Buyer will be deemed to have exercised reasonable care if it exercises at least the same degree of care as it would exercise with respect to its own property. Except as specified in the preceding sentence, Buyer will have no duty with respect to the Posted Collateral, including without limitation, any duty to enforce or preserve any rights thereto.

(e) <u>Substitutions</u>. Unless otherwise prohibited herein, upon notice to Buyer specifying the items of Posted Collateral to be exchanged, Seller may, on any Business Day, deliver to Buyer other Credit Support ("<u>Substitute Credit Support</u>"). On the Business Day following the day on which the Substitute Credit Support is delivered to Buyer, Buyer shall return to Seller the items of Credit Support specified in Seller's notice; provided, however, that Buyer shall not be required to return the specified Posted Collateral if, immediately after such return, Seller would be required to post additional Credit Support pursuant to Sections 6.2(a) and 6.2(b).

6.6 Exercise of Rights Against Posted Collateral.

(a) <u>Disputes Regarding Amount of Credit Support</u>. If either Party disputes the amount of Credit Support to be provided or returned (such Party the "<u>Disputing Party</u>"), then the Disputing Party shall (i) deliver the undisputed amount of Credit Support to the other Party (such Party, the "<u>Requesting Party</u>") and (ii) notify the Requesting Party of the existence and nature of the dispute no later than 5:00 p.m. Eastern Prevailing Time on the Business Day that the request for Credit Support was made (the "<u>Request Date</u>"). On the Business Day following the Request Date, the Parties shall consult with each other in order to reconcile the two conflicting amounts. If the Parties are not able to resolve their dispute, the Credit Support shall be recalculated, on the Business Day following the Request Date, by each Party requesting quotations from two (2) Reference Market-Makers for a total of four (4) quotations. The highest and lowest of the four (4) quotations shall be discarded and the arithmetic average shall be taken of the remaining two (2), which shall be used in order to determine the amount of Credit Support

required. On the same day the Credit Support amount is recalculated, the Disputing Party shall deliver any additional Credit Support required pursuant to the recalculation or the Requesting Party shall return any excess Credit Support that is no longer required pursuant to the recalculation.

(b) <u>Further Assurances</u>. Promptly following a request by a Party, the other Party shall use Commercially Reasonable Efforts to execute, deliver, file, and/or record any financing statement, specific assignment, or other document and take any other action that may be necessary or desirable to create, perfect, or validate any Posted Collateral or other security interest or lien, to enable the requesting party to exercise or enforce its rights or remedies under this Agreement, or with respect to Posted Collateral, or accrued interest.

(c) <u>Further Protection</u>. Seller will promptly give notice to Buyer of, and defend against, any suit, action, proceeding, or lien (other than a banker's lien in favor of the Custodian appointed by Buyer so long as no amount owing from Seller to such Custodian is overdue) that involves the Posted Collateral delivered to Buyer by Seller or that could adversely affect any security interest or lien granted pursuant to this Agreement.

### 6.7 <u>Unsecured Credit Limit</u>.

Credit Ratings (Seller Guarantor)		Unsecured Credit Limits (\$ in Millions)
S&P and/or Fitch Rating	Moody's Rating	Fixed Amount
AA- or higher	Aa3 or higher	\$15.0
A+, A	A1, A2	\$12.5
A-	A3	\$10.0
BBB+	Baa1	\$7.5
BBB	Baa2	\$5.0
BBB-	Baa3	\$2.5
Below BBB- / Unrated	Below Baa3 / Unrated	\$0

The Unsecured Credit Limit is provided in the table below:

6.8 <u>Return of Credit Support</u>. Any unused Credit Support provided under this Agreement shall be returned to Seller only after any such Credit Support has been used to satisfy any outstanding obligations of Seller in existence at the time of the expiration or termination of this Agreement. Provided such obligations have been satisfied, such Credit Support shall be returned to Seller within thirty (30) days after the earlier of (a) the expiration of the Term or (b) termination of this Agreement under Article 8, Section 9.3(b) or Section 10.1(c).

# 7. REPRESENTATIONS, WARRANTIES, COVENANTS AND ACKNOWLEDGEMENTS

7.1 <u>Representations and Warranties of Buyer</u>. Buyer hereby represents and warrants to Seller as follows:

(a) <u>Organization and Good Standing</u>; <u>Power and Authority</u>. Buyer is a corporation duly incorporated, validly existing and in good standing under the laws of the Commonwealth of Massachusetts. Subject to the receipt of the Regulatory Approval, Buyer has all requisite power and authority to execute, deliver, and perform its obligations under this Agreement.

(b) <u>Due Authorization; No Conflicts</u>. The execution and delivery by Buyer of this Agreement, and the performance by Buyer of its obligations hereunder, have been duly authorized by all necessary actions on the part of Buyer and do not and, under existing facts and Law, shall not: (i) contravene its certificate of incorporation or any other governing documents; (ii) conflict with, result in a breach of, or constitute a default under any note, bond, mortgage, indenture, deed of trust, license, contract or other agreement to which it is a party or by which any of its properties may be bound or affected; (iii) subject to receipt of the Regulatory Approval, violate any order, writ, injunction, decree, judgment, award, statute, law, rule, regulation or ordinance of any Governmental Entity or agency applicable to it or any of its properties; or (iv) result in the creation of any lien, charge or encumbrance upon any of its properties pursuant to any of the foregoing.

(c) <u>Binding Agreement</u>. This Agreement has been duly executed and delivered on behalf of Buyer and, assuming the due execution hereof and performance hereunder by Seller and receipt of the Regulatory Approval, constitutes a legal, valid and binding obligation of Buyer, enforceable against it in accordance with its terms, except as such enforceability may be limited by law or principles of equity.

(d) <u>No Proceedings</u>. As of the Effective Date, except to the extent relating to the Regulatory Approval, there are no actions, suits or other proceedings, at law or in equity, by or before any Governmental Entity or agency or any other body pending or, to the best of its knowledge, threatened against or affecting Buyer or any of its properties (including, without limitation, this Agreement) which relate in any manner to this Agreement or any transaction contemplated hereby, or which Buyer reasonably expects to lead to a material adverse effect on (i) the validity or enforceability of this Agreement or (ii) Buyer's ability to perform its obligations under this Agreement.

(e) <u>Consents and Approvals</u>. Except to the extent associated with the Regulatory Approval, the execution, delivery and performance by Buyer of its obligations under this Agreement do not and, under existing facts and Law, shall not, require any Permit or any other action by, any Person which has not been duly obtained, made or taken or that shall be duly obtained, made or taken on or prior to the date required, and all such approvals, consents, permits, licenses, authorizations, filings, registrations and actions are in full force and effect, final and non-appealable as required under applicable Law.

(f) <u>Negotiations</u>. The terms and provisions of this Agreement are the result of arm's length and good faith negotiations on the part of Buyer and equal bargaining power of the Parties. No principle of law or equity regarding construing ambiguities in this Agreement against the drafting Party shall apply.

(g) <u>Bankruptcy</u>. There are no bankruptcy, insolvency, reorganization, receivership or other such proceedings pending against or being contemplated by Buyer, or, to Buyer's knowledge, threatened against it.

(h) <u>No Default</u>. No Default or Event of Default has occurred and is continuing and no Default or Event of Default shall occur as a result of the performance by Buyer of its obligations under this Agreement.

7.2 <u>Representations and Warranties of Seller</u>. Seller hereby represents and warrants to Buyer as of the Effective Date as follows:

(a) <u>Organization and Good Standing; Power and Authority</u>. Seller is a corporation, duly formed, validly existing and in good standing under the laws of Delaware. Subject to the receipt of the Permits and any Québec Line Approvals, Seller has all requisite power and authority to execute, deliver, and perform its obligations under this Agreement.

(b) <u>Authority</u>. Seller (i) has the power and authority to own and operate its businesses and properties, to own or lease the property it occupies and to conduct the business in which it is currently engaged; (ii) is duly qualified and in good standing under the laws of each jurisdiction where its ownership, lease or operation of property or the conduct of its business requires such qualification; and (iii) holds all rights and entitlements necessary to purchase the Products from its Affiliate that owns and operates the Hydro-Québec Power Resources and, subject to the receipt of the Permits and any Québec Line Approvals, to deliver the Products to Buyer in accordance with this Agreement.

Due Authorization; No Conflicts. The execution and delivery by Seller of (c) this Agreement, and the performance by Seller of its obligations hereunder, have been duly authorized by all necessary actions on the part of Seller and do not and, under existing facts and Law, shall not: (i) contravene any of its governing documents; (ii) conflict with, result in a breach of, or constitute a default under any note, bond, mortgage, indenture, deed of trust, license, contract or other agreement to which it is a party or by which any of its properties may be bound or affected; (iii) assuming receipt of the Permits and any Québec Line Approvals, violate any order, writ, injunction, decree, judgment, award, statute, law, rule, regulation or ordinance of any Governmental Entity or agency applicable to it or any of its properties; or (iv) result in the creation of any lien, charge or encumbrance upon any of its properties pursuant to any of the foregoing. Seller is qualified to perform as a Market Participant under the ISO-NE Tariff, or is gualified to transact through another Market Participant under the ISO-NE Tariff. Seller will not be disqualified from or be materially adversely affected in the performance of any of its obligations under this Agreement by reason of market power or affiliate transaction issues under federal or state regulatory requirements.

(d) <u>Binding Agreement</u>. This Agreement has been duly executed and delivered on behalf of Seller and, assuming the due execution hereof and performance hereunder

by Buyer and receipt of the Permits and any Québec Line Approvals, constitutes a legal, valid and binding obligation of Seller, enforceable against it in accordance with its terms, except as such enforceability may be limited by law or principles of equity.

(e) <u>No Proceedings</u>. Except to the extent associated with the Permits and any U.S. Transmission Approvals, there are no actions, suits or other proceedings, at law or in equity, by or before any Governmental Entity or agency or any other body pending or, to the best of its knowledge, threatened against or affecting Seller or any of its properties (including, without limitation, this Agreement) which relate in any manner to this Agreement or any transaction contemplated hereby, or which Seller reasonably expects to lead to a material adverse effect on (i) the validity or enforceability of this Agreement or (ii) Seller's ability to perform its obligations under this Agreement.

(f) <u>Consents and Approvals</u>. Subject to the receipt of the Permits on or prior to the date such Permits are required under applicable Law and subject to the receipt of any Québec Line Approvals on or prior to the date such Québec Line Approvals are required under applicable Law, the execution, delivery and performance by Seller of its obligations under this Agreement do not and, under existing facts and Law, shall not, require any Permit or any other action by, any Person which has not been duly obtained, made or taken, and all such approvals, consents, permits, licenses, authorizations, filings, registrations and actions are in full force and effect, final and non-appealable. To Seller's knowledge, Seller shall be able to receive the Permits (including the Québec Line Approvals) in due course and as required under applicable Law to the extent that those Permits have not previously been received.

(g) <u>Project Operational</u>. The Hydro-Québec Power Resources are operational and are operated and maintained in accordance with Good Utility Practice and Seller meets all requirements of the ISO-NE Rules and ISO-NE Practices to the extent required for the Delivery of the Products to Buyer.

(h) <u>Self-Insurance</u>. Hydro-Québec maintains self-insurance with respect to the types of coverage and with such terms and conditions as are reasonable for a power system and transmission line, as applicable, of the type and size of the Hydro-Québec Power Resources and the Québec Line, consistent with Good Utility Practice and as otherwise legally required for the Hydro-Québec Power Resources and the Québec Line.

(i) <u>Qualified Clean Energy</u>. The Hydro-Québec Power Resources shall be, as of and after the Effective Date, Qualified Clean Energy Generation Units, and the Energy sold under this Agreement is Qualified Clean Energy or Qualified Shortfall Energy, as applicable.

(j) <u>Title to Products</u>. Seller shall purchase all of the Products sold hereunder directly from its Affiliate that owns and operates the Hydro-Québec Power Resources and has and shall have good and marketable title to all Products sold and Delivered to Buyer under this Agreement, free and clear of all liens, charges and encumbrances, at the time of Delivery hereunder. Seller has not sold and shall not sell any such Products to any other Person, and no Person other than Seller can claim an interest in any Products to be sold to Buyer under this Agreement.

(k) <u>Negotiations</u>. The terms and provisions of this Agreement are the result of arm's length and good faith negotiations on the part of Seller and equal bargaining power of the Parties. No principle of law or equity regarding construing ambiguities in this Agreement against the drafting Party shall apply.

(1) <u>Bankruptcy</u>. There are no bankruptcy, insolvency, reorganization, receivership or other such proceedings pending against or being contemplated by Seller or its Affiliate that owns the Hydro-Québec Power Resources, or, to Seller's knowledge, threatened against either of them.

(m) <u>No Misrepresentations</u>. The reports and other submittals by Seller to Buyer under this Agreement are not false or misleading in any material respect.

(n) <u>No Default</u>. No Default or Event of Default has occurred and is continuing and no Default or Event of Default shall occur as a result of the performance by Seller of its obligations under this Agreement.

(o) <u>Site Control</u>. As of the Effective Date, Seller or an Affiliate of Seller has all real property rights to operate the Hydro-Québec Power Resources, to interconnect the Hydro-Québec Power Resources to the Interconnecting Utility and to perform Seller's obligations under this Agreement.

7.3 <u>Continuing Nature of Representations and Warranties</u>. The representations and warranties set forth in this Article 7 are made as of the Effective Date and deemed made continually throughout the Term, subject to the removal of the references to the Regulatory Approval, Permits and Québec Line Approvals, as and when the Regulatory Approval, Permits and Québec Line Approvals are obtained. If at any time during the Term, a Party has knowledge of any event or information which causes any of the representations and warranties in this Article 7 to be untrue or misleading, such Party shall provide the other Party with prompt written notice of the event or information, the representations and warranties affected, and the corrective action such Party shall take. The notice required pursuant to this Section 7.3 shall be given as soon as practicable after the occurrence of each such event.

7.4 <u>Assignment of Transmission Rights</u>. Buyer and Seller shall use Commercially Reasonable Efforts to give effect to the assignments contemplated by Article 20, as required from time to time.

7.5 <u>Forward Capacity Market Participation</u>. Seller shall participate in such Forward Capacity Auction qualification process as required to allow Seller to qualify a Seasonal Claimed Capability of not less than 1,090 MW over the U.S. Transmission Line no later than the Guaranteed Delivery Term Start Date, as it may be extended pursuant to Sections 3.1(c) through 3.1(f). The Parties acknowledge and agree that (a) such participation by Seller is only intended to allow ISO-NE to determine which Network Upgrades would be required to (i) deliver such Seasonal Claimed Capability and (ii) satisfy the Capacity Capability Interconnection Standard under the ISO-NE Rules and (b) as contemplated by the TSA, such Network Upgrades, if any, shall be at U.S. Transmission Provider's sole expense. For the avoidance of doubt, but without limiting the condition set forth in Section 3.4(b)(ii), Seller shall have no obligation during the

Services Term to pay for such Network Upgrades or to complete the Forward Capacity Auction qualification process.

# 8. **REGULATORY APPROVAL; TERMINATION OF TSA**

8.1 <u>Receipt of Regulatory Approval</u>. Except to the extent waived by Buyer, the obligations of the Parties to perform this Agreement, other than the Parties' obligations under Sections 6.2(b) and 6.8 and Articles 8, 11 and 12, are conditioned upon and shall not become effective or binding until the receipt of the Regulatory Approval. Buyer shall file for Regulatory Approval as soon as practicable following the execution of this Agreement, and in any event shall use Commercially Reasonable Efforts to file within sixty (60) days after the Effective Date. Buyer shall notify Seller within five (5) Business Days after receipt of the Regulatory Approval or receipt of an order of the MDPU regarding this Agreement. This Agreement may be terminated by either Buyer or Seller, without liability under this Agreement as a result of such termination, subject to the return of Credit Support as provided in Section 6.8, in the event that the Regulatory Approval is not received by

8.2 <u>Transmission Approvals</u>. The obligation of the Parties to perform this Agreement is further conditioned upon the receipt of U.S. Transmission Approvals and Québec Line Approvals, in each case, on or prior to the date such U.S. Transmission Approvals are required to be obtained in accordance with the TSA or Québec Line Approvals are required to be obtained in accordance with this Agreement.

# 8.3 <u>Transmission Service Agreement</u>.

(a) Seller may terminate this Agreement in the event that the TSA is terminated for any reason other than a default by Buyer under the TSA, which is also an Event of Default by Buyer under Section 9.1(f)(ii) or (iii) of this Agreement. Unless the TSA is terminated (x) prior to the Commercial Operation Date pursuant to Section 3.3.3(a) or 3.3.5 of the TSA or (y) due to a Non-Excused Outage under the TSA, subject to the return of Credit Support as provided in Section 6.8, such terminated prior to the Commercial Operation Date pursuant by Seller shall be without liability under this Agreement. If the TSA is terminated prior to the Commercial Operation Date pursuant to Section 3.3.3(a) or 3.3.5 of the TSA, then upon Seller's termination of this Agreement under this Section 8.3(a), Seller shall pay Buyer the Termination Payment calculated in accordance with Section 9.3(b)(ii)(B). If the termination of the TSA, then upon Seller's termination of this Agreement under this Section 9.3(b)(iv)(B). Nothing herein shall pay Buyer the Termination Payment in accordance with Section 9.3(b)(iv)(B). Nothing herein shall be construed as a waiver or modification of any rights or remedies of Buyer under the TSA.

(b) Buyer may terminate this Agreement in the event that the TSA is terminated for any reason other than a default by Buyer under the TSA which is also an Event of Default by Buyer under Section 9.1(f)(ii) or (iii) of this Agreement. In the event that Buyer terminates this Agreement under this Section 8.3(b), subject to the return of Credit Support as provided in Section 6.8, termination pursuant to this Section 8.3(b) shall be without liability under this Agreement. For the avoidance of doubt, Buyer's right to terminate this Agreement under this Section 8.3(b) shall be without limitation of any right that Buyer may have to

terminate this Agreement due to an Event of Default by Seller under Section 9.2 of this Agreement.

(c) If either Buyer or Seller elects to terminate this Agreement as a result of a termination of the TSA as provided in this Section 8.3, such Party will give the other Party at least one hundred eighty (180) days advance written notice of such termination; provided, however, that during any such notice period this Agreement shall remain in full force and effect.

### 8.4 <u>DEP Advisory Ruling</u>.

(a) Except to the extent mutually waived by both Parties, the obligations of the Parties to perform this Agreement, other than the Parties' obligations under Sections 6.2(b) and 6.8 and Articles 8, 11 and 12, are conditioned upon and shall not become effective or binding until the receipt of an advisory ruling from the Massachusetts Department of Environmental Protection, pursuant to 310 CMR 2.09, (i) confirming that the Products Delivered under this Agreement, including, without limitation, any Qualified Shortfall Energy and associated Environmental Attributes Delivered pursuant to Section 4.3(c), meet the requirements of the CES as in effect on the date such advisory ruling is issued and (ii) describing the reporting and other filing and administrative requirements in order for such Products Delivered under this Agreement to satisfy the requirements of the CES (the "<u>Advisory Ruling</u>"). Buyer and Seller shall jointly file a request for the Advisory Ruling as soon as practicable following the execution of this Agreement, and in any event shall use Commercially Reasonable Efforts to file such request within sixty (60) days after the Effective Date.

This Agreement may be terminated by either Buyer or Seller, without (b) liability under this Agreement as a result of such termination, subject to the return of Credit Support as provided in Section 6.8, in the event that (i) one Party notifies the other Party within five (5) Business Days after receipt of the Advisory Ruling that the Advisory Ruling is not acceptable to such Party, acting reasonably; provided that the reason the Party determines the Advisory Ruling is not acceptable shall be that such ruling would have a material adverse effect on (1) the ability of such Party to perform its obligations under this Agreement or (2) the benefits such Party expects to receive from the Products being Delivered under this Agreement meeting the requirements of the CES, (ii) the Massachusetts Department of Environmental Protection affirmatively denies the request for an Advisory Ruling, or (iii) the Advisory Ruling is not received within ninety (90) days after the Effective Date; provided that termination under clause (iii) shall occur prior to receipt of the Advisory Ruling and within one hundred fifty (150) days The Parties may agree to jointly request a correction by the after the Effective Date. Massachusetts Department of Environmental Protection to any provision of the Advisory Ruling; provided that no such correction may be the basis for a Party's determination that the Advisory Ruling is not acceptable under clause (i) of this Section 8.4(b), unless such correction (A) has a material adverse effect on such Party's ability to perform its purchase and sale obligations under this Agreement, (B) materially alters the Price, (C) materially increases the cost to such Party of performance under this Agreement, or (D) would result in adverse balance sheet or creditworthiness impacts of such Party.

# 9. BREACHES; REMEDIES

9.1 <u>Events of Default by Either Party</u>. It shall constitute an event of default ("<u>Event</u> <u>of Default</u>") by either Party hereunder if:

(a) <u>Representation or Warranty</u>. Any breach of any representation or warranty of such Party set forth herein, or in filings or reports made pursuant to this Agreement occurs where such breach is not fully cured and corrected within thirty (30) days after the Non-Defaulting Party has provided written notice to the Defaulting Party; or

(b) <u>Payment Obligations</u>. Any undisputed payment due and payable hereunder is not made on the date due, and such failure continues for more than ten (10) Business Days after notice thereof is given by the Non-Defaulting Party to the Defaulting Party; or

- (c) <u>Other Covenants</u>. Other than:
  - (i) any of Seller's obligations to Deliver any of the Products or any portion of the Products or to deliver any of the Baseline Hydroelectric Generation Imports, or
  - (ii) a Rejected Purchase, or
  - (iii) an Event of Default described in Section 9.1(a), 9.1(b), 9.1(e) or 9.2,

such Party fails to perform, observe or otherwise to comply with any obligation hereunder, and such failure continues for more than thirty (30) days after notice thereof is given by the Non-Defaulting Party to the Defaulting Party; provided, however, that such period shall be extended for an additional period of up to thirty (30) days, if, despite using Commercially Reasonable Efforts, the Defaulting Party is unable to cure within the initial thirty (30) day period so long as such cure is diligently pursued by the Defaulting Party until such Default had been corrected, but in any event shall be cured within sixty (60) days after the notice from the Non-Defaulting Party; or

(d) <u>Bankruptcy</u>. Such Party (i) is adjudged bankrupt or files a petition in voluntary bankruptcy under any provision of any bankruptcy law or consents to the filing of any bankruptcy or reorganization petition against such Party under any such law, or (without limiting the generality of the foregoing) files a petition to reorganize pursuant to 11 U.S.C. § 101 or any similar statute applicable to such Party, as now or hereinafter in effect, (ii) makes an assignment for the benefit of creditors, or admits in writing an inability to pay its debts generally as they become due, or consents to the appointment of a receiver or liquidator or trustee or assignee in bankruptcy or insolvency of such Party, or (iii) is subject to an order of a court of competent jurisdiction appointing a receiver or liquidator or trustee of such Party or of a major part of such Party's property, which is not dismissed within sixty (60) days; or

(e) <u>Permit Compliance</u>. Such Party fails to obtain and maintain or cause to be obtained and maintained in full force and effect any Permit (other than the Regulatory Approval

or U.S. Transmission Approvals) necessary for such Party to perform its obligations under this Agreement.

(f) <u>Suspension or Termination of Transmission Rights</u>. With respect to Buyer only, (i) Buyer has failed to timely assign to Seller the right to use during the Services Term all the transmission rights held by Buyer pursuant to the TSA as contemplated in Article 20; (ii) Buyer's rights under the TSA are suspended due to a default by Buyer under the TSA or the TSA is terminated due to a default by Buyer under the TSA; or (iii) Buyer has disaffirmed, disclaimed or repudiated the TSA.

9.2 <u>Events of Default by Seller</u>. In addition to the Events of Default described in Section 9.1, each of the following shall constitute an Event of Default by Seller hereunder:

(a) <u>Failure to Maintain Credit Support or Seller Guaranty</u>. The failure of Seller to provide, maintain and/or replenish the Credit Support or the Seller Guaranty as required pursuant to Article 6, and such failure continues for more than five (5) Business Days after Buyer has provided written notice thereof to Seller. For the avoidance of doubt, it shall be deemed an Event of Default if Seller provides Credit Support in the form of a Letter of Credit and, with respect to an outstanding Letter of Credit, one of the following events occurs with respect to the issuer of such Letter of Credit: (i) such issuer shall fail to be a Qualified Institution and Seller fails to replace such Credit Support within five (5) Business Days after such failure; (ii) such issuer shall fail to comply with or perform its obligations under such Letter of Credit and Seller fails to replace such Credit Support within five (5) Business Days after such failure; or (iii) such issuer shall disaffirm, disclaim, repudiate or reject, in whole or in part, or challenge the validity of, such Letter of Credit, and such failure, disaffirmation, disclamation, repudiation or rejection continues for more than five (5) Business Days after Buyer has provided written notice thereof to Seller; or

(b) <u>Termination of TSA for Failure to Achieve Critical Milestone</u>. The termination of the TSA by Buyer pursuant to Section 14.2(b) of the TSA; or

(c) <u>Failure to Achieve the Commercial Operation Date</u>. The failure of Seller to meet the Commercial Operation Date by the Guaranteed Delivery Term Start Date, as the same may be extended in accordance with Sections 3.1(c) through 3.1(f); or

(d) <u>Failure to Satisfy ISO-NE Obligations</u>. The failure of Seller to satisfy, or cause to be satisfied (other than by Buyer), any material obligation under the ISO-NE Rules or ISO-NE Practices or any other obligation of Seller with respect to ISO-NE or other Independent System Operators, as applicable, and such failure has an adverse effect on Seller's ability to perform its obligations under this Agreement or on Buyer or Buyer's rights or ability to receive the benefits under this Agreement; provided, however, that, if Seller's failure to satisfy any obligation of Seller under the ISO-NE Rules or ISO-NE Practices does not have an adverse effect on Buyer or Buyer's ability to receive the benefits under this Agreement, Seller shall have the opportunity to cure such failure within thirty (30) days after its occurrence; or

(e) <u>Assignment</u>. The assignment of this Agreement by Seller, or the sale or transfer of Hydro-Québec or its subsidiaries' interests (or any part thereof) in the Hydro-Québec Power Resources, except as permitted in accordance with Article 14; or

(f) Recurring Delivery Reduction. The aggregate Uncured Delivery Shortfalls in any Shortfall Cure Period are more than twenty percent (20%) of the Guaranteed Qualified Clean Energy for such Shortfall Cure Period (a "Defaulted Delivery Shortfall"); provided, however, that, if (x) any such Uncured Delivery Shortfall is caused by or arises from Non-TSA Delivery Shortfalls, in whole or in part, (y) Seller presents a request to delay termination of this Agreement pursuant to this Section 9.2(f) for a period not to exceed twelve (12) months, together with a detailed plan (including reasonably satisfactory evidence of TransÉnergie's financial and technical capability to timely effectuate such plan) to restore the availability of the Québec Line such that Seller will be capable of fulfilling its minimum Delivery obligations hereunder (such that no Event of Default continues), before the end of such period and without duplication of any similar request to delay termination that has been or will be made under the TSA with respect to the same event of condition, and (z) Seller pays the Cover Damages required under Section 4.3, then Buyer shall forbear terminating this Agreement under this Section 9.2(f) for the duration of such period; provided, further, however, that, in the event that Buyer, acting in a Commercially Reasonable Manner, determines that Seller is not fulfilling its minimum Delivery obligations hereunder at the end of such period, or if Seller fails to exercise diligent, Commercially Reasonable Efforts to timely effectuate such plan, Buyer may terminate this Agreement under this Section 9.2(f).

9.3 <u>Remedies</u>.

(a) <u>Suspension of Performance and Remedies at Law</u>. Upon the occurrence and during the continuance of an Event of Default, the Non-Defaulting Party shall have the right, but not the obligation, to (i) withhold any payments due the Defaulting Party under this Agreement, (ii) suspend its performance hereunder, (iii) in the case of a default of Seller's payment obligations under this Agreement, exercise and enforce any and all of its rights and remedies under the Seller Guaranty, and (iv) exercise such other remedies as provided for in this Agreement including, without limitation, the termination right set forth in Section 9.3(b), and, to the extent not inconsistent with the terms of this Agreement, such remedies available at law and in equity. In addition to the foregoing, except for breaches for which an express remedy or measure of damages is provided herein, the Non-Defaulting Party shall retain its right of specific performance to enforce the Defaulting Party's obligations under this Agreement.

(b) <u>Termination and Termination Payment</u>. Upon the occurrence of an Event of Default, a Non-Defaulting Party may terminate this Agreement at its sole discretion by providing written notice of such termination to the Defaulting Party. If the Non-Defaulting Party terminates this Agreement, it shall be entitled to calculate and receive as its sole remedy for such Event of Default a "<u>Termination Payment</u>" as follows:

 (i) <u>Termination by Buyer Prior to Commercial Operation Date</u>. If Buyer terminates this Agreement because of an Event of Default by Seller occurring prior to the Commercial Operation Date, the Termination Payment due to Buyer shall be equal to the sum of (x) all Delay Damages due and owing by Seller through the date of such termination <u>plus</u> (y) the full amount of the Fixed Credit Support required to be provided to Buyer by Seller in accordance with Section 6.2(b).

### (ii) <u>Termination by Seller Prior to Commercial Operation Date</u>.

(A) If Seller terminates this Agreement because of an Event of Default by Buyer prior to the Commercial Operation Date, Seller shall only receive a Termination Payment if the Commercial Operation Date either occurs on or before the Guaranteed Delivery Term Start Date or would have occurred by such date but for the Event of Default by Buyer giving rise to the termination of this Agreement. In such case, the Termination Payment due to Seller shall be calculated according to the methodology in Section 9.3(b)(iv)(A), as if the Commercial Operation Date had occurred prior to the date of the termination by Seller.

(B) If Seller terminates this Agreement prior to the Commercial Operation Date pursuant to Section 8.3(a) and the TSA is terminated pursuant to Section 3.3.3(a) or 3.3.5 of the TSA, the Termination Payment due to Buyer shall be equal to the sum of (x) all Delay Damages due and owing by Seller through the date of such termination <u>plus</u> (y) the full amount of the Fixed Credit Support required to be provided to Buyer by Seller in accordance with Section 6.2(b).

### (iii) <u>Termination by Buyer on or after Commercial Operation Date</u>.

If Buyer terminates this Agreement because of an Event of (A) Default by Seller occurring on or after the Commercial Operation Date (other than an Event of Default pursuant to Section 9.2(f)), then the Termination Payment due to Buyer shall be equal to the amount, if positive, without duplication, calculated according to the following formula: (w) the present value, discounted at a rate equal to the prime rate specified in the "Money & Investing" section of The Wall Street Journal determined as of the date of the notice of default, plus 300 basis points, for each month remaining in the Services Term, of (I) the amount, if any, by which the forward market price of Energy and Environmental Attributes, as determined by the average of the quotes of at least two nationally recognized energy consulting firms or brokers chosen by Buyer, for Replacement Energy and Replacement Environmental Attributes, exceeds the applicable Price that would have been paid pursuant to Exhibit D, multiplied by (II) the amount of Guaranteed Qualified Clean Energy; plus (x) any costs reasonably incurred or to be reasonably incurred by Buyer for the Qualified Clean Energy not Delivered as a result of such termination in excess of such costs for Replacement Energy, plus (y) any costs and losses reasonably incurred by Buyer as a result of the Event of Default and termination of the Agreement, minus (z) any Owner Termination Payment (as defined in the TSA) received by Buyer pursuant to Section 3.3.5(b) of the TSA.

(B) If Buyer terminates this Agreement because of an Event of Default pursuant to Section 9.2(f), Buyer shall determine the TSA Shortfall Default Percentage and the Non-TSA Shortfall Default Percentage, and the Termination Payment due to Buyer with respect to such Event of Default shall be equal to the sum of X + Y, where:

X = the TSA Shortfall Default Percentage multiplied by the lesser of:

(I) the Termination Payment calculated under Section 9.3(b)(iii)(A) (as if Section 9.3(b)(iii)(A) had applied to such Event of Default); and

(II) the amount, if positive, without duplication, calculated according to the following formula: (x) the present value, discounted at a rate equal to the prime rate specified in the "Money & Investing" section of The Wall Street Journal determined as of the date of the notice of default, plus 300 basis points, for the lesser of each month remaining in the Services Term or the first sixty (60) months following such termination, of (a) the amount, if any, by which the forward market price of Energy and Environmental Attributes, as determined by the average of the quotes of at least two nationally recognized energy consulting firms or brokers chosen by Buyer, for Replacement Energy and Replacement Environmental Attributes, exceeds the applicable Price that would have been paid pursuant to Exhibit D, multiplied by (b) the amount of Guaranteed Qualified Clean Energy; plus (y) any costs reasonably incurred or to be reasonably incurred by Buyer for the Qualified Clean Energy not Delivered as a result of such termination in excess of such costs for Replacement Energy, plus (z) any costs and losses reasonably incurred by Buyer as a result of the Event of Default and termination of the Agreement.

### and

Y = the Non-TSA Shortfall Default Percentage multiplied by the Termination Payment calculated under Section 9.3(b)(iii)(A) (as if Section 9.3(b)(iii)(A) had applied to such Event of Default).

All such amounts shall be determined by Buyer in good faith and in a Commercially Reasonable Manner, and Buyer shall provide Seller with a reasonably detailed calculation of the Termination Payment due under this Section 9.3(b)(iii).

### (iv) <u>Termination by Seller On or After Commercial Operation Date</u>.

(A) If Seller terminates this Agreement because of an Event of Default by Buyer occurring on or after the Commercial Operation Date, the Termination Payment due to Seller shall be equal to the amount, if positive, calculated according to the following formula:

(x) the present value, discounted at a rate equal to the prime rate specified in the "Money & Investing" section of *The Wall Street Journal* determined as of the date of the notice of default, plus 300 basis points, for each month remaining in the Services Term, of (I) the amount, if, any, by which the applicable Price that would have been paid pursuant to Exhibit D, exceeds the forward market price of Energy and Environmental Attributes as determined by the average of the quotes of at least two nationally recognized energy consulting firms or brokers chosen by Seller, for Replacement Energy and Replacement Environmental Attributes, multiplied by (II) the amount of Guaranteed Qualified Clean Energy; plus,

(y) any costs and losses (including any transmission payments) reasonably incurred by Seller as a result of the Event of Default and termination of the Agreement.

(B) If Seller terminates this Agreement on or after the Commercial Operation Date pursuant to Section 8.3(a) due to a termination of the TSA resulting from a Non-Excused Outage under the TSA, the Termination Payment due to Buyer from Seller shall be equal to the lesser of:

(I) the Termination Payment calculated under Section 9.3(b)(iii)(A) (as if Section 9.3(b)(iii)(A) had applied to such termination event); and

(II) the amount, if positive, without duplication, calculated according to the following formula: (x) the present value, discounted at a rate equal to the prime rate specified in the "Money & Investing" section of *The Wall Street Journal* determined as of the date of the notice of default, plus 300 basis points, for the lesser of each month remaining in the Services Term or the first sixty (60) months following such termination, of (a) the amount, if

any, by which the forward market price of Energy and Environmental Attributes, as determined by the average of the quotes of at least two nationally recognized energy consulting firms or brokers chosen by Buyer, for Replacement Energy and Replacement Environmental Attributes, exceeds the applicable Price that would have been paid pursuant to Exhibit D, multiplied by (b) the amount of Guaranteed Qualified Clean Energy; plus (y) any costs reasonably incurred or to be reasonably incurred by Buyer for the Qualified Clean Energy not Delivered as a result of such termination in excess of such costs for Replacement Energy, plus (z) any costs and losses reasonably incurred by Buyer as a result of the termination of the Agreement.

All such amounts shall be determined by Seller in good faith and in a Commercially Reasonable Manner, and Seller shall provide Buyer with a reasonably detailed calculation of the Termination Payment due under this Section 9.3(b)(iv).

- (v) <u>Acceptability of Liquidated Damages</u>. Each Party agrees and acknowledges that (A) the damages and losses (including without limitation the loss of environmental, reliability and economic benefits contemplated under this Agreement) that the Parties would incur due to an Event of Default would be difficult or impossible to predict with certainty, and (B) it is impractical and difficult to assess actual damages in the circumstances stated, and therefore the Termination Payment as agreed to by the Parties and set forth herein is a fair and reasonable calculation of such damages.
- (vi) <u>Payment of Termination Payment</u>. The Defaulting Party shall make the Termination Payment within ten (10) Business Days after such notice is effective, regardless of whether the Termination Payment calculation is disputed. If the Defaulting Party disputes the Non-Defaulting Party's calculation of the Termination Payment, in whole or in part, the Defaulting Party shall within ten (10) Business Days after receipt of the calculation of the Termination Payment, provide to the Non-Defaulting Party a detailed written explanation of the basis for such dispute. If the Parties are unable to resolve the dispute within thirty (30) days, Article 11 shall apply.

(c) <u>Set-off</u>. The Non-Defaulting Party shall be entitled, at its option and in its discretion, to withhold and set off any amounts owed by the Non-Defaulting Party to the Defaulting Party against any payments and any other amounts owed by the Defaulting Party to the Non-Defaulting Party, including any Termination Payment payable as a result of any early termination of this Agreement.

(d) Limitation of Remedies, Liability and Damages. EXCEPT AS EXPRESSLY HEREIN, THERE SET FORTH IS NO WARRANTY OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE, AND ANY AND ALL IMPLIED WARRANTIES ARE DISCLAIMED. THE PARTIES CONFIRM THAT THE EXPRESS REMEDIES AND MEASURES OF DAMAGES PROVIDED IN THIS AGREEMENT SATISFY THE ESSENTIAL PURPOSES HEREOF. FOR BREACH OF ANY PROVISION FOR WHICH AN EXPRESS REMEDY OR MEASURE OF DAMAGES IS PROVIDED, SUCH EXPRESS REMEDY OR MEASURE OF DAMAGES SHALL BE THE SOLE AND EXCLUSIVE REMEDY, THE OBLIGOR'S LIABILITY SHALL BE LIMITED AS SET FORTH IN SUCH PROVISION AND ALL OTHER REMEDIES OR DAMAGES AT LAW OR IN EQUITY ARE WAIVED. IF NO REMEDY OR MEASURE OF DAMAGES IS EXPRESSLY PROVIDED HEREIN, THE OBLIGOR'S LIABILITY SHALL BE LIMITED TO DIRECT ACTUAL DAMAGES ONLY, SUCH DIRECT ACTUAL DAMAGES SHALL BE THE SOLE AND EXCLUSIVE REMEDY, AND ALL OTHER REMEDIES OR DAMAGES AT LAW OR IN EQUITY ARE WAIVED. UNLESS EXPRESSLY HEREIN PROVIDED, NEITHER PARTY SHALL BE LIABLE FOR CONSEQUENTIAL, INCIDENTAL, PUNITIVE, EXEMPLARY OR INDIRECT DAMAGES, LOST PROFITS OR OTHER BUSINESS INTERRUPTION DAMAGES, BY STATUTE, IN TORT OR CONTRACT, UNDER ANY INDEMNITY PROVISION OR OTHERWISE. TO THE EXTENT ANY DAMAGES REQUIRED TO BE PAID HEREUNDER ARE LIQUIDATED, THE PARTIES ACKNOWLEDGE THAT THE DAMAGES ARE DIFFICULT OR IMPOSSIBLE TO DETERMINE. **OTHERWISE OBTAINING ADEOUATE** OR AN REMEDY IS INCONVENIENT AND THE DAMAGES CALCULATED HEREUNDER CONSTITUTE A REASONABLE APPROXIMATION OF THE HARM OR LOSS.

### **10. FORCE MAJEURE**

### 10.1 Force Majeure.

The term "Force Majeure" shall mean (i) an event: (A) that was not (a) within the control of the Party claiming its occurrence; (B) that could not have been prevented or avoided by such Party through the exercise of reasonable diligence; and (C) that directly prohibits or prevents such Party from performing its obligations under this Agreement, (ii) an Excused Outage, (iii) any failure by Seller to satisfy any of its obligations to Deliver any of the Products or any portion of the Products that is caused by or arises from a Force Majeure under the TSA that is not caused, in whole or in part, by any act or omission by Seller or any of its Affiliates and that directly prohibits or prevents Seller from performing its obligations under this Agreement, or (iv) a Reliability Curtailment that is not caused, in whole or in part, by any act or omission by Seller or any of its Affiliates and that directly prohibits or prevents Seller from performing its obligations under this Agreement. Under no circumstances shall Force Majeure include (t) a change in Law or accounting standards or rules or a change in the interpretation or applicability thereof under Section 19.7(a), (u) an Adverse Determination, (v) any full or partial curtailment in the electric output of the Hydro-Québec Power Resources or in the operation of the Québec Line, in each case, that is caused by or arises from a mechanical or equipment breakdown or other mishap or events or conditions attributable to normal wear and tear or flaws at the Hydro-Québec Power Resources or the Québec Line, unless such curtailment or mishap is caused by one of the following: acts of God such as floods, hurricanes or tornados, or other

significantly unusual and abnormal weather conditions such as severe blizzards or severe ice storms; sabotage; terrorism; or war, (w) any occurrence or event that merely increases the costs or causes an economic hardship to a Party, including due to changes in market conditions, (x) any occurrence or event that was caused by or contributed to by the Party claiming the Force Majeure, (y) Seller's ability to sell the Qualified Clean Energy or Qualified Shortfall Energy or associated Environmental Attributes at a price greater than that set out in this Agreement, or (z) Buyer's ability to procure the Qualified Clean Energy at a price lower than that set out in this Agreement. In addition, a delay or inability to perform attributable to a Party's lack of preparation, a Party's failure to timely obtain and maintain, or cause to be obtained and maintained, all necessary Permits (excepting the Regulatory Approval) or qualifications, a failure to satisfy contractual conditions or commitments, or lack of or deficiency in funding or other resources shall each not constitute a Force Majeure or be the basis for a claim of Force Majeure. Seller may not raise a claim of Force Majeure based, in whole or in part, on curtailment by a Transmission Provider unless (I) either Party has contracted for firm transmission with a Transmission Provider for the Products to be delivered to or received at the Delivery Point and (II) such curtailment is due to "force majeure" or "uncontrollable force" or a similar term as defined under the Transmission Provider's transmission service agreement or tariff.

Subject to Section 3.1(d), if either Party is unable, wholly or in part, by (b) Force Majeure to perform obligations under this Agreement, such performance shall be excused and suspended so long as the circumstances that give rise to such inability exist or would exist if the Party claiming the Force Majeure used Commercially Reasonable Efforts to cure such circumstances, but for no longer period. The Party whose performance is affected shall give prompt notice thereof; such notice may be given orally or in writing but, if given orally, it shall be promptly confirmed in writing, providing details regarding the nature, extent and expected duration of the Force Majeure, its anticipated effect on the ability of such Party to perform obligations under this Agreement, and the estimated duration of any interruption in service or other adverse effects resulting from such Force Majeure, and shall be updated or supplemented to keep the other Party advised of the effect and remedial measures being undertaken to overcome the Force Majeure. Such inability to perform shall be promptly corrected to the extent it may be corrected through the exercise of due diligence. Neither Party shall be liable for any losses or damages arising out of a suspension of performance that occurs because of Force Majeure.

(c) Notwithstanding the foregoing, if the Force Majeure prevents (x) full performance or (y) Delivery of the Guaranteed Qualified Clean Energy in any material respect under this Agreement for a period of twelve (12) consecutive months or more:

except in the case provided in clause (ii) below, the Party whose performance is not prevented by Force Majeure shall have the right, upon written notice to the other Party, to terminate this Agreement; provided, however, that, in the case of any failure by Seller to satisfy any of its obligations to Deliver any of the Products or any portion of the Products that is caused by or arises from a Force Majeure affecting the Hydro-Québec Power Resources or the Québec Line, if (A) Seller presents a request to delay termination of this Agreement pursuant to this Section

10.1(c) for a period not to exceed twelve (12) months, together with a detailed plan (including reasonably satisfactory evidence of Hydro-Québec's or TransÉnergie's financial and technical capability to timely effectuate such plan) to restore the electric output of the Hydro-Québec Power Resources and the availability of the Québec Line such that Seller will be capable of fulfilling its minimum Delivery obligations hereunder, to Buyer before the end of the initial twelve (12) month period in which Seller's ability to Deliver has been prevented, in whole or in part, by such an event of Force Majeure and without duplication of any similar request to delay termination has been or will be made under the TSA with respect to the same event or condition, and (B) Seller pays the full Transmission Service Payment (as defined in the TSA) that Buyer otherwise would be required to pay U.S. Transmission Provider in such instance, then Buyer shall forbear terminating this Agreement under this Section 10.1(c) for the duration of such period; provided, further, however, that, in the event that Buyer, acting in a Commercially Reasonable Manner, determines that Seller is not fulfilling its minimum Delivery obligations hereunder at the end of such period, or if Seller fails to exercise diligent, Commercially Reasonable Efforts to timely effectuate such plan, Buyer may A termination of this Agreement terminate this Agreement. pursuant to this Section 10.1(c)(i) shall be without further recourse to either Party. In no event will any delay or failure of performance caused by any conditions or events of Force Majeure extend this Agreement beyond its stated Term;

in the case of a failure by Seller to satisfy any of its obligations to (ii) Deliver any of the Products or any portion of the Products that is caused by or arises from a Force Majeure under, and as defined in, the TSA that is not caused, in whole or in part, by any act or omission by Seller or any of its Affiliates and that directly prohibits or prevents Seller from performing its obligations under this Agreement, Buyer shall have the right, upon written notice to Seller, to terminate this Agreement on the later of (x) the date that is twelve (12) months after the commencement of such Force Majeure or (y) the termination of the period during which Buyer is forbearing from terminating the TSA to permit U.S. Transmission Provider to resolve such Force Majeure under Section 15.1(c) of the TSA. A termination of this Agreement pursuant to this Section 10.1(c)(ii) shall be without further recourse to either Party. In no event will any delay or failure of performance caused by any conditions or events of Force Majeure extend this Agreement beyond its stated Term.

# **11. DISPUTE RESOLUTION**

### 11.1 <u>Consultation</u>.

In the event of any dispute, controversy or claim between the Parties (a) arising out of or relating to this Agreement (collectively, a "Dispute"), in addition to any other remedies provided hereunder, the Parties shall initially attempt to resolve any Dispute through consultations between the Parties. Subject to Section 11.2 and except as expressly provided otherwise in this Agreement, if a Dispute has not been timely resolved pursuant to this clause (a) within fifteen (15) Business Days after written notice of such Dispute has been given, then either Party may seek to resolve such Dispute in the courts of the Commonwealth of Massachusetts or a U.S. District Court in the Commonwealth of Massachusetts and any appellate court from any thereof; provided, however, that, if the Dispute is subject to Section 11.2, then either Party may elect to proceed with the mediation through FERC's Dispute Resolution Service. If one Party fails to participate in the consultations provided for in this Section 11.1, the other Party can initiate mediation prior to the expiration of the fifteen (15) Business Days. Unless otherwise agreed, the Parties will select a mediator from the FERC panel. The Parties may, by written agreement signed by both Parties, alter any time deadline, location(s) for meeting(s), or procedure outlined herein or in FERC's rules for mediation. The procedure specified herein shall be the sole and exclusive procedure for the resolution of Disputes.

(b) All negotiations, consultations, and mediations pursuant to this Section 11.1 shall be deemed to be confidential and shall be treated as compromise and settlement negotiations, and no evidence with regard to any proposal made during such negotiations, consultations, or mediations shall be admissible in any FERC proceeding or filing under Section 11.2 or in any other judicial or other proceeding.

(c) The fees and expenses associated with mediation shall be divided equally between the Parties, and each Party shall be responsible for its own legal fees, including but not limited to attorney fees, associated with any Dispute.

### 11.2 Disputes to be Resolved by FERC.

(a) In the event that a Dispute over any matter under the primary or exclusive jurisdiction of FERC (a "FERC Jurisdictional Issue") is not resolved in accordance with Section 11.1, either Party shall have the right to file for relief with FERC; provided that, in the event either Party objects to the reference of any such matter to FERC on the ground that such matter is not subject to the primary or exclusive jurisdiction of FERC, the matter shall be referred to FERC for resolution of the Dispute as to whether or not such matter is subject to the primary or exclusive jurisdiction of FERC. Nothing contained in this Agreement shall be construed as precluding a Party from filing any answer, protest or other opposition to any FERC filing made by the other Party, unless expressly prohibited under the terms of this Agreement.

(b) In the event that any Dispute is submitted to FERC for resolution as provided in Section 11.2(a), the Party submitting the Dispute to FERC shall be responsible for providing written notice of such filing to the other Party and to U.S. Transmission Provider. The Parties agree to support the consolidation of Disputes involving FERC Jurisdictional Issues that arise under this Agreement and are common to disputes involving FERC Jurisdictional Issues

that arise under the TSA or any other Proposal Agreement (as defined in the TSA). Notwithstanding the foregoing, in the event that FERC determines that it does not have the jurisdiction to, or otherwise does not want to, hear or determine any portion of a Dispute or other matter so referred to FERC, either Party may seek to resolve such Dispute in the courts of the Commonwealth of Massachusetts or a U.S. District Court in the Commonwealth of Massachusetts and any appellate court from any thereof.

11.3 <u>Consent to Jurisdiction</u>. Subject to Section 11.2, each Party agrees that any legal action or proceeding with respect to or arising out of this Agreement or the TSA shall be brought in or removed to the courts of the Commonwealth of Massachusetts or a U.S. District Court in the Commonwealth of Massachusetts and any appellate court from any thereof. By execution and delivery of this Agreement, each Party hereby accepts for itself and in respect of its property, generally and unconditionally, the jurisdiction of the aforesaid courts. The Parties irrevocably consent to the service of process out of any of the aforementioned courts in any such action or proceeding by the mailing of copies thereof by registered or certified airmail, postage prepaid, to the applicable Party at its respective addresses for notices as specified in Article 17. Nothing herein shall affect the right to serve process in any other manner permitted by law. Each Party hereby waives any right to stay or dismiss any action or proceeding under or in connection with this Agreement or the TSA brought before the foregoing courts on the basis of forum non-conveniens.

11.4 <u>Waiver of Jury Trial</u>. EACH PARTY IRREVOCABLY AND UNCONDITIONALLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN ANY LEGAL ACTION, PROCEEDING, CAUSE OF ACTION, OR COUNTERCLAIM ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY.

### **12. CONFIDENTIALITY**

12.1 <u>Nondisclosure</u>. Buyer and Seller each agrees not to disclose to any Person and to keep confidential, and to cause and instruct its Affiliates, officers, directors, employees, partners and representatives not to disclose to any Person and to keep confidential, any non-public information relating to the terms and provisions of this Agreement, and any information relating to the Products to be supplied by Seller hereunder, and such other non-public information that is designated as "Confidential." Notwithstanding the foregoing, any such information may be disclosed:

(a) to the extent Buyer determines it is appropriate in connection with efforts to obtain or maintain the Regulatory Approval or to seek rate recovery for amounts expended by Buyer under this Agreement;

(b) as required by applicable laws, regulations, filing requirements, rules or orders or by any subpoena or similar legal process of any Governmental Entity so long as the disclosing Party gives the non-disclosing Party written notice at least two (2) Business Days prior to such disclosure, if practicable;

(c) to the Affiliates of either Party and to the consultants, attorneys, auditors, financial and other advisors, lenders or investors or potential lenders or investors and their advisors of either Party or their Affiliates, but solely to the extent they have a need to know that information;

(d) in order to comply with any rule or regulation of ISO-NE, other system operators, as applicable, or of any stock exchange or similar Person or for financial disclosure purposes;

(e) to the extent the non-disclosing Party shall have consented in writing prior to any such disclosure; and

(f) to the extent that the information was previously made publicly available other than as a result of a breach of this Section 12.1;

provided, however, that, in each case, that the Party seeking such disclosure shall, to the extent practicable, use Commercially Reasonable Efforts to prevent or limit the disclosure. The Parties shall be entitled to all remedies available at law or in equity to enforce or seek relief in connection with this Section 12.1.

12.2 <u>Public Statements</u>. No public statement, press release or other voluntary publication regarding this Agreement or the transactions to be made hereunder shall be made or issued without the prior consent of the other Party.

### **13. INDEMNIFICATION**

13.1 <u>Indemnification Obligations</u>. Seller shall indemnify, defend and hold Buyer and its partners, shareholders, directors, officers, employees and agents (including, but not limited to, Affiliates and contractors and their employees), harmless from and against all liabilities, damages, losses, penalties, claims, demands, suits and proceedings of any nature whatsoever arising from or related to Seller's execution, delivery or performance of this Agreement, or Seller's negligence, gross negligence, or willful misconduct, or Seller's failure to satisfy any obligation or liability under this Agreement or Seller's failure to satisfy any regulatory requirement or commitment associated with this Agreement.

13.2 <u>Failure to Defend</u>. If Seller fails to assume the defense of a claim meriting indemnification, Buyer may at the expense of Seller contest, settle or pay such claim, and shall promptly reimburse Buyer for all costs incurred by Buyer associated therewith.

# 14. ASSIGNMENT AND CHANGE OF CONTROL

14.1 <u>Prohibition on Assignments</u>. Except as permitted under this Article 14, this Agreement (and any portion thereof) may not be assigned by either Party without the prior written consent of the other Party, which consent may not be unreasonably withheld, conditioned or delayed. The Party requesting the other Party's consent to an assignment of this Agreement will reimburse such other Party for all "out of pocket" costs and expenses such other Party reasonably incurs in connection with that consent, without regard to whether such consent is provided. When assignable, this Agreement shall be binding upon, shall inure to the benefit of,

and may be performed by, the successors and assignees of the Parties, except that no assignment, pledge or other transfer of this Agreement by either Party shall operate to release the assignor, pledgor, or transferor from any of its obligations under this Agreement (and shall not impair any Credit Support given by Seller hereunder or the Seller Guaranty) unless the other Party (or its successors or assigns) consents in writing to the assignment, pledge or other transfer and expressly releases the assignor, pledgor, or transferor from its obligations thereunder.

14.2 <u>Permitted Assignment by Seller</u>. Buyer's consent shall not be required for Seller or its Affiliates to pledge or assign the Hydro-Québec Power Resources, this Agreement or the revenues under this Agreement, in each case, to any Affiliate of Seller; provided, however, that, if Seller requests Buyer's consent to an assignment, (a) Buyer shall provide that consent subject to Buyer's execution of a consent to assignment in a form acceptable to Buyer and Seller, and (b) Seller will reimburse Buyer for all "out of pocket" costs and expenses Buyer reasonably incurs in connection with that consent, without regard to whether such consent is provided. Provided that Seller maintains its ability to continue to satisfy Seller's obligation to perform under this Agreement, Buyer's consent shall not be required for Seller or its Affiliates to sell or transfer any interest in the Hydro-Québec Power Resources.

14.3 <u>Change in Control over Seller</u>. Buyer's consent shall be required for any change in Control over Seller, except to the extent such change in Control results from the direct or indirect transfer of interests in Hydro-Québec, which consent shall not be unreasonably withheld, conditioned or delayed and shall be provided if Buyer reasonably determines that such change in Control does not have a material adverse effect on Seller's creditworthiness or Seller's ability to perform its obligations under this Agreement.

14.4 <u>Permitted Assignment by Buyer</u>.

Seller:

- (a) Buyer shall have the right to assign this Agreement without consent of
  - (i) in connection with any merger or consolidation of Buyer with or into another Person or any exchange of all of the common stock or other equity interests of Buyer or Buyer's parent for cash, securities or other property or any acquisition, reorganization, or other similar corporate transaction involving all or substantially all of the common stock or other equity interests in, or assets of, Buyer; provided that (x) either (A) the proposed assignee's credit rating is at least either "BBB-" from S&P or "Baa3" from Moody's or (B) the proposed assignee's credit rating is equal to or better than that of Buyer at the time of the proposed assignment or (y) if such approval is required under applicable Law, the transaction associated with such assignment, has been approved by the MDPU or the appropriate Governmental Entity, in each case, with an express assumption of Buyer's obligations hereunder in writing, reasonably acceptable to Seller and Buyer; or
  - (ii) to any substitute purchaser of the Products; provided that (x) both(A) the proposed assignee's credit rating is at least either "BBB-"

from S&P or "Baa3" from Moody's and (B) the proposed assignee's credit rating is equal to or better than that of Buyer at the time of the proposed assignment, and (y) if such approval is required under applicable Law, such assignment has been approved by the MDPU or the appropriate Governmental Entity, in each case, with an express assumption of Buyer's obligations hereunder in writing, reasonably acceptable to Seller and Buyer. For purposes of clarification, an assignment of this Agreement pursuant to this clause (ii) includes an assignment to any third party other than the successor in interest to Buyer pursuant to a transaction to which clause (i) applies.

14.5 <u>Prohibited Assignments</u>. Any purported assignment of this Agreement not in compliance with the provisions of this Article 14 shall be null and void.

### 15. TITLE; RISK OF LOSS

Title to and risk of loss related to (a) Qualified Clean Energy shall transfer from Seller to Buyer at the Delivery Point and (b) Qualified Shortfall Energy shall transfer from Seller to Buyer at the Delivery Point or the NEMASSBOST Hub, as applicable. Title to and risk of loss related to the associated Environmental Attributes shall transfer to Buyer when the same are credited to Buyer's GIS account(s) or the GIS account(s) designated by Buyer to Seller in writing. Seller warrants that it shall deliver to Buyer the Products free and clear of all liens and claims therein or thereto by any Person.

### 16. AUDIT

16.1 <u>Audit</u>. Each Party shall have the right, upon reasonable advance notice, and at its sole expense (unless the other Party has defaulted under this Agreement, in which case the Defaulting Party shall bear the expense) and during normal working hours, to examine the records of the other Party to the extent reasonably necessary to verify the accuracy of any statement, charge or computation made pursuant to this Agreement. If requested, a Party shall provide to the other Party additional information documenting the quantities of Qualified Clean Energy or Qualified Shortfall Energy, as applicable, and associated Environmental Attributes Delivered or provide hereunder. If any such examination reveals any overcharge, the necessary adjustments in such statement and the payments thereof shall be made promptly and shall include interest at the Late Payment Rate from the date the overpayment was made until credited or paid.

16.2 <u>Access to Financial Information</u>. Seller shall provide to Buyer, within fifteen (15) days after receipt of Buyer's written request, applicable financial information and statements of Hydro-Québec, as reasonably necessary for Buyer to address any reasonable inquiries relating to Seller's financial resources.

# **17.** NOTICES

Any notice or communication given pursuant hereto shall be in writing and (a) delivered personally (personally delivered notices shall be deemed given upon written acknowledgment of receipt after delivery to the address specified or upon refusal of receipt); or (b) delivered by

electronic mail (notices sent by electronic mail shall be deemed given upon confirmation of delivery); in each case, addressed as follows or to such other addresses as may hereafter be designated by either Party to the other in writing:

If to Buyer:	Attention: Renewable Contract Manager, Environmental Transactions National Grid 100 East Old Country Road, Second Floor Hicksville, NY 11801-4218 Email: RenewableContracts@nationalgrid.com With a copy to: ElectricSupply@nationalgrid.com
With a copy to:	Legal Department Attn: Cynthia R. Clark, Esq. Assistant General Counsel National Grid 175 East Old Country Road Hicksville, NY 11801 Email: cynthia.clark@nationalgrid.com
If to Seller:	H.Q. Energy Services (U.S.) Inc. 75, René-Lévesque Boulevard West, 18 <sup>th</sup> Floor Montréal (Québec) Canada H2Z 1A4 Attention: General Manager

For billing purposes only:

H.Q. Energy Services (U.S.) Inc. 75, René-Lévesque Boulevard West, 18<sup>th</sup> Floor Montréal (Québec) Canada H2Z 1A4 Attention: Billing Manager

For scheduling purposes only:

H.Q. Energy Services (U.S.) Inc.
75, René-Lévesque Boulevard West, 18<sup>th</sup> Floor Montréal (Québec) Canada
H2Z 1A4
Attention: Manager Whole Sale Market – East

For credit purposes only:

H.Q. Energy Services (U.S.) Inc. 75, René-Lévesque Boulevard West, 18<sup>th</sup> Floor Montréal (Québec) Canada H2Z 1A4 Attention: Credit Risk Advisor With a copy to:

H.Q. Energy Services (U.S.) Inc. 75, René-Lévesque Boulevard West, 18<sup>th</sup> Floor Montréal (Québec) Canada H2Z 1A4 Attention: General Counsel and Vice President – Legal Affairs

# **18. WAIVER AND MODIFICATION**

This Agreement may be amended and its provisions and the effects thereof waived only by a writing executed by the Parties, and no subsequent conduct of any Party or course of dealings between the Parties shall effect or be deemed to effect any such amendment or waiver. No waiver of any of the provisions of this Agreement shall be deemed or shall constitute a waiver of any other provision hereof (whether or not similar), nor shall such waiver constitute a continuing waiver unless otherwise expressly provided. The failure of either Party to enforce any provision of this Agreement shall not be construed as a waiver of or acquiescence in or to such provision. Buyer shall determine in its sole discretion whether any amendment or waiver of the provisions of this Agreement shall require Regulatory Approval or MDPU filing and/or approval. If Buyer determines that any such approval or filing is required, then such amendment or waiver shall not become effective unless and until Regulatory Approval or such other approval is received, or such MDPU filing is made and any requested MDPU approval is received.

## **19. INTERPRETATION**

19.1 <u>Choice of Law</u>. Interpretation and performance of this Agreement shall be in accordance with, and shall be controlled by, the laws of the Commonwealth of Massachusetts (without regard to its principles of conflicts of law).

19.2 <u>Headings</u>. Article and Section headings are for convenience only and shall not affect the interpretation of this Agreement. References to articles, sections and exhibits are, unless the context otherwise requires, references to articles, sections and exhibits of this Agreement. The words "hereof" and "hereunder" shall refer to this Agreement as a whole and not to any particular provision of this Agreement.

19.3 <u>Forward Contract</u>. The Parties acknowledge and agree that this Agreement and the transactions contemplated hereunder are a "forward contract" within the meaning of the United States Bankruptcy Code.

19.4 <u>Standard of Review</u>. The Parties acknowledge and agree that the standard of review for any avoidance, breach, rejection, termination or other cessation of performance of or changes to any portion of this integrated, non-severable Agreement (as described in Article 24) over which FERC has jurisdiction, whether proposed by Seller, by Buyer, by a non-party of, by FERC acting *sua sponte* shall be the "public interest" standard of review set forth in *United Gas Pipe Line Co. v. Mobile Gas Serv. Co.*, 350 U.S. 332 (1956) and *Federal Power Comm'n v. Sierra Pac. Power Co.*, 350 U.S. 348 (1956). Each Party agrees that if it seeks to amend any applicable power sales tariff during the Term, such amendment shall not in any way materially

and adversely affect this Agreement without the prior written consent of the other Party. Each Party further agrees that it shall not assert, or defend itself, on the basis that any applicable tariff is inconsistent with this Agreement.

19.5 Change in ISO-NE Rules and Practices. This Agreement is subject to the ISO-NE Rules and ISO-NE Practices. If, during the Term, any ISO-NE Rule or ISO-NE Practice relating to the Delivery of Energy to the Delivery Point is terminated, modified or amended or is otherwise no longer applicable, resulting in a material alteration of a material right or obligation of a Party hereunder, the Parties agree to negotiate in good faith in an attempt to amend or clarify this Agreement to embody the Parties' original intent regarding their respective rights and obligations under this Agreement; provided that neither Party shall have any obligation to agree to any particular amendment or clarification of this Agreement. The intent of the Parties is that any such amendment or clarification reflect, as closely as possible, the intent, substance and effect of the ISO-NE Rule or ISO-NE Practice being replaced, modified, amended or made inapplicable as such ISO-NE Rule or ISO-NE Practice was in effect prior to such termination, modification, amendment, or inapplicability; provided that such amendment or clarification shall not in any event alter (a) the purchase and sale obligations of the Parties pursuant to this Agreement or (b) the Price. In the event that the Parties cannot agree upon such amendments within sixty (60) days after such ISO-NE Rule or ISO-NE Practice change described above, the Dispute shall be resolved in accordance with Article 11.

19.6 <u>Dodd Frank Act Representations</u>. The Parties agree that this Agreement (including all transactions reflected herein) is not a "swap" within the meaning of the Commodity Exchange Act and the rules, interpretations and other guidance of the Commodity Futures Trading Commission ("<u>CFTC rules</u>"), and that the primary intent of this Agreement is physical settlement (i.e., actual transfer of ownership) of the nonfinancial commodity and not solely to transfer price risk. In reliance upon such agreement, each Party represents to the other that:

(a) With respect to the commodity to be purchased and sold hereunder, it is a commercial market participant, a commercial entity and a commercial party, as such terms are used in the CFTC rules, and it is a producer, processor, or commercial user of, or a merchant handling, the commodity and it is entering into this Agreement for purposes related to its business as such;

(b) It is not registered or required to be registered under the CFTC rules as a swap dealer or a major swap participant;

(c) It has entered into this Agreement in connection with the conduct of its regular business and it has the capacity or ability to regularly make or take delivery of the commodity to be purchased and sold hereunder;

(d) With respect to the commodity to be purchased and sold hereunder, it intends to make or take physical delivery of the commodity;

(e) At the time that the Parties enter into this Agreement, any embedded volumetric optionality in this Agreement is primarily intended by the holder of such option or

optionality to address physical factors or regulatory requirements that reasonably influence demand for, or supply of, the commodity to be purchased and sold hereunder;

(f) With respect to any embedded commodity option in this Agreement, such option is intended to be physically settled so that, if exercised, the option would result in the sale of the commodity to be purchased or sold hereunder for immediate or deferred shipment or delivery;

(g) The commodity to be purchased and sold hereunder is a nonfinancial commodity, and is also an exempt commodity or an agricultural commodity, as such terms are defined and interpreted in the CFTC rules.

To the extent that reporting of any transactions related to this Agreement is required by the CFTC rules, the Parties agree that Seller shall be responsible for such reporting (the "**Reporting Party**"). The Reporting Party's reporting obligations shall continue until the reporting obligation has expired or has been terminated in accordance with CFTC rules. Buyer, as the Party that is not undertaking the reporting obligations shall timely provide the Reporting Party all necessary information requested by the Reporting Party for it to comply with CFTC rules.

### 19.7 <u>Change in Law or Buyer's Accounting Treatment, Subsequent Judicial or</u> <u>Regulatory Action</u>.

If, during the Term, there is a change in Law or accounting standards or (a) rules or a change in the interpretation or applicability thereof that, in each case, would result in adverse balance sheet or creditworthiness impacts on Buyer associated with this Agreement or the amounts paid for Products purchased hereunder, Buyer shall prepare an amendment to this Agreement to avoid or mitigate such impacts. Buyer shall use Commercially Reasonable Efforts to prepare such amendment in a manner that mitigates any material adverse effect(s) on Seller (as identified by Seller acting reasonably) that could reasonably be expected to result from such amendment, but only to the extent that such mitigation can be accomplished in a manner that is consistent with the purpose of such amendment. Seller agrees to execute such amendment provided that such amendment does not (unless Seller otherwise agrees) (i) alter the purchase and sale obligations of the Parties, including any changes to the Products, (ii) alter the Price or (iii) materially increase the cost to Seller of performance under this Agreement (except as contemplated by Sections 4.7(b) and 4.7(c)). In the event that the Parties cannot agree upon such amendment within sixty (60) days after the change described above necessitating such amendment, the Dispute shall be resolved in accordance with Article 11.

(b) Upon a determination by a court or regulatory body having jurisdiction over this Agreement or any of the Parties hereto, or over the establishment and enforcement of any of the statutes or regulations or orders or actions of regulatory agencies (including the MDPU) supporting this Agreement or the rights or obligations of the Parties hereunder that any of the statutes or regulations supporting this Agreement or the rights or obligations of the Parties hereunder that any of the statutes or regulations of regulatory agencies (including the Parties hereunder, or orders of or actions of regulatory agencies (including the MDPU) implementing such statutes or regulations, or this Agreement on its face or as applied, violates any applicable Law (including the State or Federal Constitution) (an "Adverse Determination"), each Party shall have the right to suspend performance under this Agreement without liability. Seller may deliver and sell Products to a third party during any period of time for which Buyer suspends

payments or purchases under this Section 19.7(b). Upon an Adverse Determination becoming final and non-appealable, this Agreement shall be rendered null and void, and no Termination Payment shall be due to either Party.

#### 20. ASSIGNMENT OF TRANSMISSION RIGHTS

As of the Effective Date, Buyer has entered into the TSA, pursuant to (a) which Buyer has agreed to pay for all costs of transmission related to Delivery by Seller over the U.S. Transmission Line hereunder. Notwithstanding anything to the contrary herein, including but not limited to the allocation of costs and responsibilities described in Section 3.6, Section 4.2(c) and Section 4.5(b), the Parties acknowledge and agree that (i) Buyer has the obligation to pay its Proportionate Share (as defined in the TSA) of the Transmission Service Payments (as defined in the TSA) pursuant to the TSA, (ii) such Transmission Service Payments are calculated under Section 8.1 of the TSA based on the cost of transmission service over the U.S. Transmission Line. In respect of Seller's obligations to deliver, Buyer hereby transfers, assigns and conveys to Seller all of its rights, title and interest in and to the Firm Transmission Service, Other Transmission Rights and Market Products (each as defined in the TSA) and Seller hereby accepts the assignment of such rights, title and interest in and to such Firm Transmission Service, Other Transmission Rights and Market Products (each as defined in the TSA) and collectively referred to as "TSA Transmission Rights"). If and to the extent Seller determines from time to time, and in its sole discretion, that all or any portion of the TSA Transmission Rights are not required to accomplish Delivery, Seller shall then offer to resell such unused TSA Transmission Rights to third parties in accordance with applicable Law as may then be in effect (including the terms and conditions of FERC Order No. 890 et seq., if applicable). Seller may post such offers on the Open Access Same-Time Information System (OASIS) site established by U.S. Transmission Provider pursuant to Section 10.4 of the TSA. Seller shall use Commercially Reasonable Efforts to maximize the revenues obtained through resales of any TSA Transmission Rights that Seller determines to be excess to its needs. Seller shall credit the Proportionate Share (as defined in the TSA) of ninety percent (90%) of the proceeds Seller receives through any resale of TSA Transmission Rights, net of reasonable fees (including attorneys' fees) and other expenses incurred in connection with such resale, to Buyer pursuant to Section 5.2(d). If Buyer receives from U.S. Transmission Provider any proceeds of a resale of TSA Transmission Rights assigned to Seller pursuant to this Article 20, Buyer shall then credit ten percent (10%) of such proceeds, net of reasonable fees (including attorneys' fees) and other expenses incurred in connection with such resale, to Seller pursuant to Section 5.2(d).

(b) Notwithstanding the assignment of the TSA Transmission Rights from Buyer to Seller under this Article 20, nothing set forth in this Agreement shall be construed to require Seller to pay, reimburse or indemnify Buyer for (i) any amounts that are the responsibility of U.S. Transmission Provider under the terms of the TSA, including, but not limited to, costs for Network Upgrades or (ii) any amounts paid by Buyer under the TSA, except in the case of clause (ii), (x) as provided in Sections 3.5(j) and 10.1(c)(i) and Exhibit H, (y) in connection with any Cover Damages due from Seller or (z) in connection with any Termination Payment due from Seller. Nothing in the preceding sentence shall be construed to limit any amounts that Seller would otherwise owe Buyer in accordance with Section 13.1.

#### 21. WAIVER OF IMMUNITIES

The Parties acknowledge and agree that this Agreement and the transactions contemplated hereby constitute a commercial transaction. To the extent a Party (including any assignees of a Party's rights or obligations under this Agreement) may be entitled, in any jurisdiction, to claim for itself, or any of its assets, revenues or properties, sovereign or other immunity, as the case may be, from service of process, suit, the jurisdiction of any court or arbitral tribunal, attachment (whether in aid of execution or otherwise) or enforcement of a judgment (interlocutory or final) or award or any other legal process in a matter arising out of or relating to this Agreement, each Party agrees not to claim or assert, and hereby waives, such immunity. Without limiting the generality of the foregoing, each Party agrees that the waivers set forth in this Article 21 shall have the fullest scope permitted under the Immunities Act and under any other applicable Law related to sovereign immunity.

## 22. COUNTERPARTS; FACSIMILE SIGNATURES

Any number of counterparts of this Agreement may be executed, and each shall have the same force and effect as an original. Facsimile signatures hereon or on any notice or other instrument delivered under this Agreement shall have the same force and effect as original signatures.

#### 23. NO DUTY TO THIRD PARTIES

Except as provided in any consent to assignment of this Agreement, nothing in this Agreement nor any action taken hereunder shall be construed to create any duty, liability or standard of care to any Person not a Party to this Agreement.

#### 24. SEVERABILITY

Except as otherwise provided in Article 8 or Section 19.5 or 19.7, if any term or provision of this Agreement or the interpretation or application of any term or provision to any prior circumstance is held to be unenforceable, illegal or invalid by a court or agency of competent jurisdiction, the remainder of this Agreement and the interpretation or application of all other terms or provisions to Persons or circumstances other than those which are unenforceable, illegal or invalid shall not be affected thereby, and each term and provision shall be valid and be enforced to the fullest extent permitted by law so long as all essential terms and conditions of this Agreement for both Parties remain valid, binding and enforceable and have not been declared to be unenforceable, illegal or invalid by a Governmental Entity of competent jurisdiction. The Parties acknowledge and agree that essential terms and conditions of this Agreement, and that the essential terms and conditions of this Agreement for Buyer also include, without limitation, the terms and conditions of Section 19.6.

#### **25. INDEPENDENT CONTRACTOR**

Nothing in this Agreement shall be construed as creating any relationship between Buyer and Seller other than that of Seller as independent contractor for the sale of Products, and Buyer as principal and purchaser of the same. Neither Party shall be deemed to be the agent of the

other Party for any purpose by reason of this Agreement, and no partnership or joint venture or fiduciary relationship between the Parties is intended to be created hereby. Nothing in this Agreement shall be construed as creating any relationship between Buyer and the Interconnecting Utility.

### 26. ENTIRE AGREEMENT

This Agreement shall constitute the entire agreement and understanding between the Parties hereto and shall supersede all prior agreements and communications.

[Signature page follows]

IN WITNESS WHEREOF, each of Buyer and Seller has caused this Agreement to be duly executed on its behalf as of the date first above written.

# MASSACHUSETTS ELECTRIC COMPANY AND NANTUCKET ELECTRIC COMPANY, D/B/A NATIONAL GRID, as Buyer

By:	CRE
Name: John V. Vaugba	
Title: Authorized Signatory	

H.Q. ENERGY SERVICES (U.S.) INC., as Seller

By: \_\_\_\_\_

Name: David Murray

Title: Chairman of the Board and President

[Signature Page to the Power Purchase Agreement]

IN WITNESS WHEREOF, each of Buyer and Seller has caused this Agreement to be duly executed on its behalf as of the date first above written.

## MASSACHUSETTS ELECTRIC COMPANY AND NANTUCKET ELECTRIC COMPANY, D/B/A NATIONAL GRID, as Buyer

By: \_\_\_\_\_

Name: John V. Vaughn

Title: Authorized Signatory

H.Q. ENERGY SERVICES (U.S.) INC., as Seller

By: Name: David Murray

Title: Chairman of the Board and President

[Signature Page to the Power Purchase Agreement]

## EXHIBIT A

## DESCRIPTION OF HYDRO-QUÉBEC POWER RESOURCES

The Hydro-Québec Power Resources shall include the following facilities as of the Effective Date:

	Name of Facility	Location
Facility #1	Robert-Bourassa	Québec
Facility #2	La Grande-4	Québec
Facility #3	La Grande-3	Québec
Facility #4	La Grande 2-A	Québec
Facility #5	Beauharnois	Québec
Facility #6	Manic-5	Québec
Facility #7	La Grande-1	Québec
Facility #8	René-Lévesque (Manic-3)	Québec
Facility #9	Jean-Lesage (Manic-2)	Québec
Facility #10	Bersimis-1	Québec
Facility #11	Manic-5-PA	Québec
Facility #12	Outardes-3	Québec
Facility #13	Sainte-Marguerite-3	Québec
Facility #14	Laforge-1	Québec
Facility #15	Bersimis-2	Québec
Facility #16	Outardes-4	Québec
Facility #17	Eastmain-1-A	Québec
Facility #18	Carillon	Québec
Facility #19	Romaine-2	Québec
Facility #20	Toulnustouc	Québec
Facility #21	Outardes-2	Québec
Facility #22	Eastmain-1	Québec
Facility #23	Brisay	Québec
Facility #24	Péribonka	Québec
Facility #25	Laforge-2	Québec
Facility #26	Trenche	Québec
Facility #27	La Tuque	Québec
Facility #28	Beaumont	Québec
Facility #29	Romaine-1	Québec
Facility #30	McCormick	Québec
Facility #31	Rocher-de-Grand-Mère	Québec
Facility #32	Paugan	Québec
Facility #33	Rapide-Blanc	Québec

	Name of Facility	Location
Facility #34	Shawinigan-2	Québec
Facility #35	Shawinigan-3	Québec
Facility #36	Manic-1	Québec
Facility #37	Rapides-des-Îles	Québec
Facility #38	Chelsea	Québec
Facility #39	Sarcelle	Québec
Facility #40	La Gabelle	Québec
Facility #41	Première-Chute	Québec
Facility #42	Les Cèdres	Québec
Facility #43	Rapides-Farmer	Québec
Facility #44	Rapides-des-Quinze	Québec
Facility #45	Chute-des-Chats	Québec
Facility #46	Rapide-des-Cœurs	Québec
Facility #47	Grand-Mère	Québec
Facility #48	Rapide-2	Québec
Facility #49	Rapide-7	Québec
Facility #50	Chute-Allard	Québec
Facility #51	Bryson	Québec
Facility #52	Mercier	Québec
Facility #53	Rivière-des-Prairies	Québec
Facility #54	Hart-Jaune	Québec
Facility #55	Chute-Hemmings	Québec
Facility #56	Sept-Chutes	Québec
Facility #57	Drummondville	Québec
Facility #58	Saint-Narcisse	Québec
Facility #59	Chute-Bell	Québec
Facility #60	Mitis-1	Québec
Facility #61	Mitis-2	Québec
Facility #62	Romaine-3	Québec

## EXHIBIT B

## SCHEDULE OF GUARANTEED QUALIFIED CLEAN ENERGY TO BE DELIVERED FROM THE HYDRO-QUÉBEC POWER RESOURCES

	Jan	Feb	Mar	Apr	May	Jun	Jul	Aug	Sep	Oct	Nov	Dec
	(MWh											
	per											
Year	hour)											
1	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348
2	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348
3	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348
4	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348
5	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348
6	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348
7	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348
8	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348
9	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348
10	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348
11	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348
12	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348
13	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348
14	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348
15	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348
16	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348
17	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348
18	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348
19	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348
20	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348	498.348

#### EXHIBIT C

## SAMPLE MARKET EXPOSURE CALCULATION

		On-Peak	Off-Peak	On-Peak	Off-Peak	On-Peak	Off-Peak	
		Energy	Energy	Contract	Contract	Forward	Forward	
Future		Quantity	Quantity	Price	Price	Price <sup>1</sup>	Price <sup>2</sup>	Monthly Market
Year	Month	(MWh)	(MWh)	(\$/MWh)	(\$/MWh)	(\$/MWh)	(\$/MWh)	Exposure (\$)
								(I) = (G-E) * C +
(A)	(B)	(C)	(D)	(E)	(F)	(G)	(H)	(H-F) * D
1	Jan	168,561	202,210	\$51.51	\$51.51	\$89.10	\$69.10	\$9,893,088.55
	Feb	160,907	176,973	\$51.51	\$51.51	\$87.50	\$66.70	\$8,479,251.77
	Mar	176,535	194,236	\$51.51	\$51.51	\$53.20	\$44.50	-\$1,063,249.78
	Apr	170,873	187,937	\$51.51	\$51.51	\$34.30	\$26.10	-\$7,716,213.87
	May	168,561	202,210	\$51.51	\$51.51	\$31.60	\$23.60	-\$8,999,724.87
	June	170,873	187,937	\$51.51	\$51.51	\$32.70	\$23.80	-\$8,421,866.73
	July	168,561	202,210	\$51.51	\$51.51	\$40.30	\$27.40	-\$6,764,844.49
	Aug	176,535	194,236	\$51.51	\$51.51	\$39.50	\$26.70	-\$6,939,178.60
	Sept	162,900	195,911	\$51.51	\$51.51	\$33.60	\$23.30	-\$8,444,176.90
	Oct	176,535	194,236	\$51.51	\$51.51	\$33.20	\$23.70	-\$8,634,056.77
	Nov	162,900	195,911	\$51.51	\$51.51	\$39.40	\$30.40	-\$6,108,391.80
	Dec	168,561	202,210	\$51.51	\$51.51	\$60.00	\$51.30	\$1,388,622.38
2	Jan	168,561	202,210	\$52.80	\$52.80	\$90.90	\$70.50	\$10,001,297.91
	Feb	160,907	176,973	\$52.80	\$52.80	\$89.30	\$68.00	\$8,563,083.83
	Mar	176,535	194,236	\$52.80	\$52.80	\$54.30	\$45.40	-\$1,172,543.44
	Apr	170,873	187,937	\$52.80	\$52.80	\$35.00	\$26.60	-\$7,965,499.52
	May	168,561	202,210	\$52.80	\$52.80	\$32.20	\$24.10	-\$9,275,777.75
	June	170,873	187,937	\$52.80	\$52.80	\$33.40	\$24.30	-\$8,671,152.38
	July	168,561	202,210	\$52.80	\$52.80	\$41.10	\$27.90	-\$7,007,185.09

<sup>1</sup> CME Group, NYMEX ISO-NE Mass Hub On-Peak data,

http://www.cmegroup.com/trading/energy/electricity/nepool-internal-hub-5-mw-peak-calendarmonth-day-ahead-swap-futures\_quotes\_settlements\_futures.html#tradeDate=05%2F24%2F2018, accessed May 24, 2018.

<sup>2</sup> CME Group, NYMEX ISO-NE Mass Hub Off-Peak data,

http://www.cmegroup.com/trading/energy/electricity/nepool-internal-hub-5-mw-off-peakcalendar-month-day-ahead-swap-

<u>futures\_quotes\_settlements\_futures.html#tradeDate=05%2F24%2F2018</u>, accessed May 24, 2018.

		On-Peak	Off-Peak	On-Peak	Off-Peak	On-Peak	Off-Peak	
		Energy	Energy	Contract	Contract	Forward	Forward	
Future		Quantity	Quantity	Price	Price	Price <sup>1</sup>	Price <sup>2</sup>	Monthly Market
Year	Month	(MWh)	(MWh)	(\$/MWh)	(\$/MWh)	(\$/MWh)	(\$/MWh)	Exposure (\$)
	Aug	176,535	194,236	\$52.80	\$52.80	\$40.30	\$27.20	-\$7,179,127.13
	Sept	162,900	195,911	\$52.80	\$52.80	\$34.30	\$23.80	-\$8,695,057.26
	Oct	176,535	194,236	\$52.80	\$52.80	\$33.90	\$24.20	-\$8,891,658.79
	Nov	162,900	195,911	\$52.80	\$52.80	\$40.20	\$31.00	-\$6,323,391.10
	Dec	168,561	202,210	\$52.80	\$52.80	\$61.20	\$52.30	\$1,314,811.10
3	Jan	168,561	202,210	\$54.12	\$54.12	\$92.70	\$71.90	\$10,098,384.15
	Feb	160,907	176,973	\$54.12	\$54.12	\$91.10	\$69.40	\$8,654,476.83
	Mar	176,535	194,236	\$54.12	\$54.12	\$55.40	\$46.30	-\$1,292,960.23
	Apr	170,873	187,937	\$54.12	\$54.12	\$35.70	\$27.10	-\$8,225,549.48
	May	168,561	202,210	\$54.12	\$54.12	\$32.80	\$24.60	-\$9,562,953.76
	June	170,873	187,937	\$54.12	\$54.12	\$34.10	\$24.80	-\$8,931,202.34
	July	168,561	202,210	\$54.12	\$54.12	\$41.90	\$28.50	-\$7,240,427.87
	Aug	176,535	194,236	\$54.12	\$54.12	\$41.10	\$27.70	-\$7,430,198.78
	Sept	162,900	195,911	\$54.12	\$54.12	\$35.00	\$24.30	-\$8,956,701.94
	Oct	176,535	194,236	\$54.12	\$54.12	\$34.60	\$24.70	-\$9,160,383.94
	Nov	162,900	195,911	\$54.12	\$54.12	\$41.00	\$31.60	-\$6,549,154.72
	Dec	168,561	202,210	\$54.12	\$54.12	\$62.40	\$53.30	\$1,229,876.69
4	Jan	168,561	202,210	\$55.47	\$55.47	\$94.60	\$73.30	\$10,201,203.40
	Feb	160,907	176,973	\$55.47	\$55.47	\$92.90	\$70.80	\$8,735,733.43
	Mar	176,535	194,236	\$55.47	\$55.47	\$56.50	\$47.20	-\$1,424,500.14
	Apr	170,873	187,937	\$55.47	\$55.47	\$36.40	\$27.60	-\$8,496,363.76
	May	168,561	202,210	\$55.47	\$55.47	\$33.50	\$25.10	-\$9,844,396.75
	June	170,873	187,937	\$55.47	\$55.47	\$34.80	\$25.30	-\$9,202,016.62
	July	168,561	202,210	\$55.47	\$55.47	\$42.70	\$29.10	-\$7,484,793.77
	Aug	176,535	194,236	\$55.47	\$55.47	\$41.90	\$28.30	-\$7,672,969.97
	Sept	162,900	195,911	\$55.47	\$55.47	\$35.70	\$24.80	-\$9,229,110.93
	Oct	176,535	194,236	\$55.47	\$55.47	\$35.30	\$25.20	-\$9,440,232.22
	Nov	162,900	195,911	\$55.47	\$55.47	\$41.80	\$32.20	-\$6,785,682.66
	Dec	168,561	202,210	\$55.47	\$55.47	\$63.60	\$54.40	\$1,154,040.10
5	Jan	168,561	202,210	\$56.86	\$56.86	\$96.50	\$74.80	\$10,309,412.77
	Feb	160,907	176,973	\$56.86	\$56.86	\$94.80	\$72.20	\$8,819,565.49
	Mar	176,535	194,236	\$56.86	\$56.86	\$57.60	\$48.10	-\$1,570,870.89
	Apr	170,873	187,937	\$56.86	\$56.86	\$37.10	\$28.20	-\$8,762,736.76
	May	168,561	202,210	\$56.86	\$56.86	\$34.20	\$25.60	-\$10,140,670.58
	June	170,873	187,937	\$56.86	\$56.86	\$35.50	\$25.80	-\$9,487,183.33
	July	168,561	202,210	\$56.86	\$56.86	\$43.60	\$29.70	-\$7,727,134.37
	Aug	176,535	194,236	\$56.86	\$56.86	\$42.70	\$28.90	-\$7,930,572.00
	Sept	162,900	195,911	\$56.86	\$56.86	\$36.40	\$25.30	-\$9,515,872.35

		On-Peak	Off-Peak	On-Peak	Off-Peak	On-Peak	Off-Peak	
		Energy	Energy	Contract	Contract	Forward	Forward	
Future		Quantity	Quantity	Price	Price	Price <sup>1</sup>	Price <sup>2</sup>	Monthly Market
Year	Month	(MWh)	(MWh)	(\$/MWh)	(\$/MWh)	(\$/MWh)	(\$/MWh)	Exposure (\$)
	Oct	176,535	194,236	\$56.86	\$56.86	\$36.00	\$25.70	-\$9,734,911.34
	Nov	162,900	195,911	\$56.86	\$56.86	\$42.60	\$32.80	-\$7,036,563.03
	Dec	168,561	202,210	\$56.86	\$56.86	\$64.90	\$55.50	\$1,080,228.82

#### EXHIBIT D

#### QUALIFIED CLEAN ENERGY AND ENVIRONMENTAL ATTRIBUTE PRICING

1. <u>Price for Products</u>. Commencing on the Commercial Operation Date, the Price for the Delivered Products in nominal dollars shall be as follows:

	On-Peak	Off-Peak
Contract	Price	Price
Year	(\$/MWh)	(\$/MWh)
1	\$51.51	\$51.51
2	\$52.80	\$52.80
3	\$54.12	\$54.12
4	\$55.47	\$55.47
5	\$56.86	\$56.86
6	\$58.28	\$58.28
7	\$59.74	\$59.74
8	\$61.23	\$61.23
9	\$62.76	\$62.76
10	\$64.33	\$64.33
11	\$65.94	\$65.94
12	\$67.59	\$67.59
13	\$69.28	\$69.28
14	\$71.01	\$71.01
15	\$72.78	\$72.78
16	\$74.60	\$74.60
17	\$76.47	\$76.47
18	\$78.38	\$78.38
19	\$80.34	\$80.34
20	\$82.35	\$82.35

To the extent that the Delivered Products for any month include Qualified Shortfall Energy that cures a TSA Delivery Shortfall, as provided in Section 4.3(c)(ii), the Price to be paid for each megawatt-hour of such Qualified Shortfall Energy shall be increased by the amount that would have been due from Buyer to the U.S. Transmission Provider under the TSA with respect to such TSA Delivery Shortfall had the Non-Excused Outage giving rise to that TSA Delivery Shortfall not occurred, converted to a \$/MWh basis assuming 100% Hourly Availability (as defined in the TSA) at the full amount of the Contract Capacity (as defined in the TSA) for that month.

Example: If the Price equals \$50.00/MWh, the TSA rate equals \$10,800.00 per MW of Contract Capacity (as defined in the TSA) per month, and the number of hours in such month equal 720, then the Price for each megawatt-hour of the Qualified Shortfall Energy that cures such TSA Delivery Shortfall for such month equals \$65.00/MWh.

Solely for purposes of Buyer's accounting for the Products purchased under the Agreement and Price to be paid for such Products, the Price per MWh for the Products shall be allocated between Energy and Environmental Attributes as follows:

(a) Energy = The \$/MWh price of Energy for applicable month shall be equal to the weighted average Locational Marginal Price in that month (also on a \$/MWh basis) for the Node on the Pool Transmission Facilities that is the Delivery Point.

(b) Environmental Attributes = the Price less the Energy allocation determined above for the applicable billing period, expressed in \$/MWh.

If the market price at the Delivery Point in the Real-Time Energy Market or the Day-Ahead Energy Market, as applicable based on the market in which the applicable Internal Bilateral transaction was entered pursuant to Section 4.2(a), for Energy Delivered by Seller is negative in any hour, the payment to Seller for deliveries of Qualified Clean Energy or Qualified Shortfall Energy, as applicable, shall be reduced by the difference between the absolute value of the hourly LMP at the Delivery Point and \$0.00 per MWh for that Qualified Clean Energy or Qualified Shortfall Energy, as applicable, for each such hour. Each monthly invoice shall reflect a reduction for all hours in the applicable month in which the LMP for the Energy at the Delivery Point is less than \$0.00 per MWh.

Examples: If delivered Qualified Clean Energy equals 1 MWh and Price equals \$50.00/MWh:

LMP at the Delivery Point equals (or is greater than) Buyer payment of Price to Seller Seller credit/reimbursement for negative LMP to Buyer Net Result: Buyer pays Seller \$50 for that hour	\$0.00/MWh \$50.00 \$0.00
LMP at the Delivery Point equals Buyer payment of Price to Seller Seller credit/reimbursement for negative LMP to Buyer Net Result: Seller credits or reimburses Buyer \$100 for that hour	-\$150.00/MWh \$50.00 \$150.00

## EXHIBIT E

## DESCRIPTION OF THE NEW TRANSMISSION FACILITIES

[See Attachment A to the TSA]

## EXHIBIT F

## FORM OF PROGRESS REPORT

For the Quarter Ending: [\_\_\_\_\_]

Status of progress toward permitting for the Québec Line:

Status of progress toward acquisition of real property rights for the Québec Line:

Events expected to result in delays in achievement of permitting or acquisition of real property rights for the Québec Line:

Current projection for Commercial Operation Date: [\_\_\_\_\_]

## EXHIBIT G

## FORM OF SELLER GUARANTY



75, boulevard René-Lévesque oust 5<sup>ième</sup> étage Montréal, Québec, Canada H2Z 1A4

## **GUARANTY AGREEMENT**

This Guaranty Agreement ("Guaranty"), dated as of \_\_\_\_\_\_, 201\_, is made and entered into between Hydro-Québec, a body politic and corporate, duly incorporated and regulated by Hydro-Québec Act (R.S.Q., chapter H-5) and having its head office and principal place of business at 75, René-Lévesque Boulevard West, Montréal, Québec, Canada, H2Z 1A4, hereinafter referred to as the "Guarantor" and Massachusetts Electric Company and Nantucket Electric Company, d/b/a National Grid, a corporation duly organized under the laws of the State of Massachusetts, having its principal place of business at 40 Sylvan Road, Waltham, MA 02451-1120, United States of America, hereinafter referred to as the "Beneficiary".

WHEREAS the Beneficiary and H.Q. ENERGY SERVICES (U.S.) INC., a corporation created under the laws of the State of Delaware, having its place of business at 225 Asylum Street, 27th Floor, Hartford, CT 06103, United States of America (hereinafter referred to as "HQUS"), an indirectly owned subsidiary of the Guarantor, have executed a Power Purchase Agreement dated as of June 13, 2018 (hereinafter referred to as the "Agreement");

WHEREAS the Guarantor will directly or indirectly benefit from the Agreement;

WHEREAS the Beneficiary has required that the Guarantor unconditionally guarantee to the Beneficiary all payment obligations of HQUS under the Agreement; subject to a maximum dollar limitation as provided in Section 1 of this Guaranty;

NOW THEREFORE, in consideration of the premises, the Guarantor hereby agrees with the Beneficiary as follows:

**Section 1.** <u>Payment Guaranty</u>. The Guarantor absolutely, irrevocably, and unconditionally guarantees to the Beneficiary, its successors and endorsees and assignees, the prompt payment when due of all present and future amounts owed by HQUS to the Beneficiary in accordance with the Agreement, upon a written demand by the Beneficiary stating that HQUS has failed to fulfill its obligations and liabilities resulting from the Agreement and that the amount claimed is due and payable to the Beneficiary, provided that the liability of the Guarantor under this Guaranty shall be limited in the aggregate to FIFTEEN MILLION U.S. Dollars (U.S. \$15,000,000) plus all reasonable expenses incurred by the Beneficiary to enforce its rights against the Guarantor under this Guaranty including, without limitation, attorneys' fees, court costs and similar costs.

At the Guarantor's request, the Beneficiary shall provide the Guarantor with information respecting the obligations and liabilities of HQUS with regard to the Agreement.

**Section 2.** <u>Nature of Guarantee</u>. The Guarantor's obligations hereunder shall be subject to all the contractual protections, limitations, waivers, exclusions and rights which HQUS has under the Agreement, and the Guarantor shall be entitled to the benefits of any modification of, amendment to, waiver of or consent to departure from the Agreement to the extent, if any, HQUS would have been entitled to such benefits. Nonetheless, this Guaranty shall not be deemed discharged, impaired or affected by the existence, validity, enforceability, perfection, or extent of any collateral for any obligations under the Agreement of HQUS.

**Section 3.** <u>Consents, Waivers and Renewals</u>. The Guarantor agrees that the Beneficiary may at any time and from time to time, either before or after maturity thereof, without notice to or further consent of the Guarantor, extend the time of payment of any payment obligations under the Agreement, exchange or surrender any collateral therefor, or renew the Agreement, and may also make any agreement with HQUS or with any other party to, or person liable for, or any of the obligations contemplated in the Agreement, or interested therein, for the extension, renewal, payment, compromise, discharge or release thereof, in whole or in part, or for any modification of the terms thereof or of any agreement between the Beneficiary and HQUS or any such other party or person, without in any way impairing or affecting this Guaranty. The Guarantor agrees that the Beneficiary may resort to the Guarantor for payment in virtue of the Agreement, whether or not the Beneficiary shall have resorted to any collateral security, or shall have proceeded against any other obligor principally or secondarily obligated with respect to any of the obligations in virtue of the Agreement.</u>

**Section 4.** <u>Subrogation</u>. In any case, including HQUS's insolvency, the Guarantor will not exercise any rights, which it may acquire by way of subrogation, before all amounts due to the Beneficiary under the Agreement shall have been paid in full. Subject to the foregoing, upon payment of all the obligations related to the Agreement, the Guarantor shall be subrogated to the rights of the Beneficiary against HQUS and the Beneficiary agrees to take, at the Guarantor's expense, such steps as the Guarantor may reasonably request to implement such subrogation.

**Section 5.** <u>Cumulative Rights</u>. No failure on the part of the Beneficiary to exercise, and no delay in exercising any right, remedy or power hereunder shall operate as a waiver thereof, nor shall any single or partial exercise by the Beneficiary of any right, remedy or power hereunder preclude any other future exercise of any right, remedy or power. Each and every right, remedy and power hereby granted to the Beneficiary or allowed to it by law or other agreement shall be cumulative and not exclusive of any other, and may be exercised by the Beneficiary from time to time.

Section 6. <u>Waiver of Notice</u>. The Guarantor waives notice of the acceptance of this Guaranty, notice of dishonor, presentment and demand, except as set forth in Section 1, notice of exercise of any right and all other notices whatsoever.

## Section 7. <u>Representations and Warranties</u>.

The Guarantor represents and warrants that:

a) It is a corporation duly organized, validly existing and in good standing under the laws of the jurisdiction of its incorporation and has full corporate power to execute, deliver and perform this Guaranty.

b) The execution, delivery and performance of the Guaranty have been and remain duly authorized by all necessary corporate action and do not contravene any provision of law or the Guarantor's constitutional documents or any contractual restriction binding on the Guarantor or its assets.

c) The Guaranty constitutes a legal, valid and binding obligation of the Guarantor enforceable against the Guarantor in accordance with its terms, subject, as to enforcement, to bankruptcy, insolvency, reorganization and other similar laws and to general principles of equity.

**Section 8.** <u>Setoff and Counterclaims</u>. The Guarantor shall be entitled to assert all rights and defenses that HQUS may be entitled to under the Agreement, including, but not limited to, any setoff or counterclaims that HQUS is or may be entitled to. Notwithstanding the preceding sentence, the liability of the Guarantor under the Guaranty shall not be affected because of the bankruptcy, insolvency, dissolution or liquidation of HQUS.

**Section 9.** <u>**Termination**</u>. This guarantee of payment is a continuing guarantee effective during the term of the Agreement and until complete performance by HQUS of its obligations under the Agreement and payment in full of all obligations.

Section 10. <u>Assignment</u>. Neither Party may assign its rights, interest or obligations hereunder to any other person without the prior written consent of the Guarantor or the Beneficiary, as the case may be; provided that the Beneficiary may assign its rights, interest and obligations under this Guaranty to the assignee of its rights in the Agreement in the event of a permitted assignment thereunder.

**Section 11.** <u>Notices</u>. All notices or other communications in respect of this Guaranty shall be in writing, and delivered by hand or by registered mail (return receipt requested), overnight courier service or given by facsimile (except for a demand of payment) and addressed or directed as follows:

If to the Guarantor:

**HYDRO-QUÉBEC** Attention : Vice-President Financing, Treasury and Pension Fund 75, René-Lévesque Boulevard West 5<sup>th</sup> floor Montréal (Québec) Canada H2Z 1A4 Facsimile: (514) 289-5409

If to the Beneficiary:

Attn: Renewable Contract Manager, Environmental Transactions National Grid 100 East Old Country Road, Second Floor Hicksville, NY 11801-4218 United States of America Email: RenewableContracts@nationalgrid.com With a copy to: ElectricSupply@nationalgrid.com

or such address as the Guarantor or the Beneficiary may give notice to the other party, from time to time.

Section 12. <u>Successors; Governing Law</u>. This Guarantee shall be binding upon the Guarantor, its successors and assignees, and shall be governed by and construed in accordance with the laws of the State of New York.

Section 13. <u>Entire Agreement</u>. This Guaranty constitutes the entire agreement of the Guarantor and the Beneficiary pertaining to the subject matter hereof and supersedes all prior written or oral agreements and understandings between the Guarantor and the Beneficiary with respect to the subject matter hereof.

**Section 14.** <u>Amendments</u>. No amendments or modifications of or to any provision of this Guaranty shall be binding until in writing and executed by the Guarantor and the Beneficiary.

IN WITNESS WHEREOF, the Guarantor hereto has executed this Guaranty, as of the date set forth above.

## HYDRO-QUÉBEC

By: \_\_\_\_\_

Name: Title:

# MASSACHUSETTS ELECTRIC COMPANY AND NANTUCKET ELECTRIC COMPANY, d/b/a NATIONAL GRID

By: \_\_\_\_\_

Name: Title:

#### EXHIBIT H

#### DESCRIPTION OF METHODOLOGY TO ENSURE INCREMENTAL HYDROELECTRIC GENERATION

- 1. Within forty-five (45) days after the end of each Contract Year during the Services Term, Seller will provide to Buyer the following information for such Contract Year:
  - a. the annual quantity of Energy produced by the Hydro-Québec Power Resources (excluding Qualified Clean Energy and Qualified Shortfall Energy Delivered under the Agreement) physically delivered to any node on the ISO-NE Pool Transmission Facilities, in accordance with the terms of this Exhibit H and the ISO-NE Rules and ISO-NE Practices ("**Baseline Hydroelectric Generation Imports**"). The Baseline Hydroelectric Generation Imports"). The Baseline Hydroelectric Generation Imports may be delivered across any of the transmission ties existing as of the Effective Date between the New England Control Area and adjacent control areas, as well as across the Additional HQUS TSA Capacity (as defined in the TSA).
  - b. the annual quantity for the "Minimum Required Baseline Hydroelectric Generation Imports" which shall be calculated as follows: 9.45 TWh, multiplied by "A", and then multiplied by "B", and then, for Contract Years eleven (11) through twenty (20) only, multiplied by "C", and then multiplied by "D", and then multiplied by "E", and then reduced, if applicable, by "F".

Where:

A = the ratio calculated as the number of On-Peak hours in the Contract Year for which the Day-Ahead Energy Market LMP at the ISO-NE Internal Hub was equal to or greater than the Floor Price (as defined below), divided by the total number of On-Peak hours in the Contract Year, where the "**Floor Price**" equals \$25 per MWh for the first Contract Year, and shall increase by the rate (if positive) at which the average of the monthly Quebec Consumer Price Index, as published by Statistics Canada, for the twelve (12) month period that ended on March 31 of the preceding Contract Year increased over the average of the monthly Quebec Consumer Price Index, as published by Statistics Canada, for the twelve (12) month period that ended on March 31 of the preceding Contract Year increased over the average of the twelve (12) month period that ended on March 31 of the Contract Year immediately prior to the preceding Contract Year. If such rate is not positive, the Floor Price shall remain the same as in the preceding Contract Year.

B = the ratio calculated as the aggregate amount (in MW) of the yearly Total Transfer Capabilities, for imports to New England, of the HQ Phase II, Highgate, New York AC, and Maritimes (New Brunswick) AC transmission ties for the Contract Year, divided by 3304 MW (i.e., the historical annual average of such aggregate amount for the years 2014 through 2016, as provided in Attachment 1). If this calculated ratio is greater than 1, then B shall be set to 1.

C = the ratio calculated as Hydro-Quebec's net electricity exports out of Québec for the Contract Year, divided by 29.1 TWh (i.e., the historical annual average of the corresponding exports outside of Québec for the years 2014 through 2016). If this calculated ratio is greater than 1, then C shall be set to 1.

D = the ratio calculated as the sum of , plus the product of multiplied by the ratio calculated as the MW of HQ Phase II firm transmission capacity rights which were acquired by the Seller and its Affiliates for the Contract Year divided by 781 MW (i.e., the historical average of the such rights held by the Seller and its Affiliates for the years 2014 through 2016, as provided in Attachment 1). If this calculated ratio is greater than 1, then D shall be set to 1. The Seller must actively and timely pursue the acquisition of these firm transmission rights for such Contract Year, using Commercially Reasonable Efforts, but no less than the standard practices it used to acquire such rights for the years 2014 through 2014 through 2016, or D shall be set to 1.

E = the ratio calculated as the sum of M, plus the product of M multiplied by the ratio calculated as the MW of Firm Point-to-Point Service export capacity rights through New Brunswick at the New Brunswick / ISO-NE interface which were acquired by the Seller and its Affiliates for the Contract Year divided by 300 MW (i.e., the historical average of the Firm Point-to-Point Service export capacity rights held by the Seller and its Affiliates for the years 2014 through 2016, as provided in Attachment 1). For the avoidance of doubt, as an example, if the Seller for some reason was only able to acquire 300 MW of Firm Point-to-Point Service export capacity rights through New Brunswick for half of the Contract Year, rather than as a yearly right for the entire year, then the calculated ratio for E would equal 0.90. If this calculated ratio is greater than 1, then D shall be set to 1. The Seller must actively and timely pursue the acquisition of these firm transmission rights for such Contract Year, using Commercially Reasonable Efforts, but no less than the standard practices it used to acquire such rights for the years 2014 through 2016, or E shall be set to 1.

F = for the Contract Year, only the amount, if any, of Deliveries of Qualified Shortfall Energy to cure TSA Delivery Shortfalls in excess of 5.32 TWh.

- c. the annual quantity for the "**Baseline Hydroelectric Generation Imports Shortfall**" which shall be calculated by subtracting the Baseline Hydroelectric Generation Imports from the Minimum Required Baseline Hydroelectric Generation Imports; for the avoidance of doubt, the Baseline Hydroelectric Generation Imports Shortfall can be a positive value, zero, or a negative value.
- d. for each Contract Year from the fifth through the twentieth Contract Year, the "**Five-Year Average Baseline Hydroelectric Generation Imports Shortfall**," which shall be calculated as the average of the Baseline Hydroelectric Generation Imports Shortfall for the Contract Year and the Baseline Hydroelectric Generation Imports Shortfall as previously determined for each of the immediately preceding four (4) Contract Years.

2. Within sixty (60) days after the end of the fifth Contract Year during the Services Term, if the Five-Year Average Baseline Hydroelectric Generation Imports Shortfall determined for the fifth Contract Year during the Services Term is a positive value, then Seller shall reimburse Buyer for an amount (the "Baseline Hydroelectric Generation Imports Shortfall Damages") determined by the following formula:

(Five-Year Average Baseline Hydroelectric Generation Imports Shortfall determined for the fifth Contract Year / average of the Minimum Required Baseline Hydroelectric Generation Imports determined for each of the first five (5) Contract Years) x (the total Transmission Service Payments (as defined in the TSA) incurred by Buyer under the TSA over the first five (5) Contract Years, to the extent such Transmission Service Payments are not reimbursed by Seller as Cover Damages, plus any amounts due from Buyer to Seller pursuant to Section 4.3(c)(ii) over the first five (5) Contract Years)

3. Within sixty (60) days after the end of each of Contract Years six (6) through twenty (20) during Services Term, if the Five-Year Average Baseline Hydroelectric Generation Imports Shortfall applicable to the Contract Year is a positive value, then Seller shall reimburse Buyer for the Baseline Hydroelectric Generation Imports Shortfall Damages determined by the following formula:

(Five-Year Average Baseline Hydroelectric Generation Imports Shortfall applicable to the Contract Year / average of the Minimum Required Baseline Hydroelectric Generation Imports applicable to the Contract Year and for each of the immediately preceding four (4) Contract Years) x (the total Transmission Service Payments (as defined in the TSA) incurred by Buyer under the TSA in the Contract Year, to the extent such Transmission Service Payments are not reimbursed by Seller as Cover Damages, plus any amounts due from Buyer to Seller pursuant to Section 4.3(c)(ii) in the Contract Year) x (the applicable Percentage as provided in the table below).

Contract Year for which calculation is being made	Percentage
Contract Years 6-10	80%
Contract Years 11-15	60%
Contract Years 16-20	40%

4. Each calculation of the Baseline Hydroelectric Generation Imports Shortfall Damages shall be reasonably detailed in order to permit Buyer to confirm the calculations provided by Seller. Buyer shall provide to Seller any documentation or other information reasonably required to permit Seller to confirm the amount of Transmission Service Payments (as defined in the TSA) paid by Buyer under the TSA. Seller shall provide to Buyer any documentation or other information reasonably requested by Buyer in order to permit Buyer to confirm the amounts used to determine the Baseline Hydroelectric Generation Imports, the Minimum Required Baseline Hydroelectric Generation Imports, the Baseline Hydroelectric Generation Imports Shortfall, and the Five-Year Average Baseline Hydroelectric Generation Imports Shortfall and the Baseline Hydroelectric Generation Imports Shortfall Damages provided by Seller.

- 5. In no event shall Buyer reimburse or otherwise make any payment to Seller under this Exhibit H.
- 6. The rights provided in this Exhibit H shall collectively be the sole and exclusive remedies of Buyer with respect to any failure by Seller to deliver the Minimum Required Baseline Hydroelectric Generation Imports.
- 7. If and when the TSA is terminated, the Parties shall no longer be bound by the terms and provisions of this Exhibit H, except to the extent necessary to enforce the rights and obligations of the Parties arising under this Exhibit H before such termination.

Attachment 1 – Supporting Information

## Historical Total Transfer Capabilities

HQ Phase II: 2014 1496 MW 2015 1419 MW 2016 1251 MW

For HQ Phase II, the value represents the yearly average of the hourly minimum of these 2 points: HQT-NE export TTCs as posted by TransEnergie, and HQ Phase II import TTCs as posted by ISO-NE.

Highgate: 2014 210 MW 2015 212 MW 2016 216 MW

For Highgate, the value represents the yearly average of the hourly minimum of these 2 points: HQT-HIGH export TTCs as posted by TransEnergie, and HQ Highgate import TTCs as posted by ISO-NE.

New York AC: 2014 1109 MW 2015 1104 MW 2016 1228 MW

For New York AC, the value represents the yearly average of the hourly minimum of these 4 points: HQT-MASS export TTCs as posted by TransEnergie, HQ\_GEN\_WHEEL TTCs as posted by NYISO, N.E.\_LOAD\_SANDY PD TTCs as posted by NYISO, and NY North import TTCs as posted by ISO-NE.

Maritimes (New Brunswick) AC: 2014 469 MW 2015 673 MW 2016 524 MW

For Maritimes (New Brunswick) AC, the value represents the yearly average of the hourly minimum of these 3 points: HQT-NB export TTCs as posted by TransEnergie, HQ to Mepco export TTCs as posted by NB Transmission and System Operator, and New Brunswick import TTCs as posted by ISO-NE.

Aggregate Annual TTC: 2014 3,284 MW 2015 3,408 MW 2016 3,219 MW

## Historical Annual Average TTC for 2014-2016: 3,304 MW

## HQ Phase II Firm Transmission Capacity Rights

	January	February	March	April	May	June	July	August	September	October	November	December	Avg
2 014	779	779	779	779	779	780	780	780	780	780	780	780	780
2 015	780	780	780	780	780	780	790	790	790	790	850	850	795
2 016	768	768	768	768	768	768	768	768	768	768	768	768	768

## Yearly Firm Point-to-Point Service Export Capacity Rights through New Brunswick

Query Status of Requests			transstatus			
Back New Query Re-run Query						
	Custom	er Information		Sel	ler Information	
	Company (DUNS)		Company (DU			
	MEHQ (25683)	0548)	NBPC (205489156)			
	Company Con		Company Cor	ntact		
	Administrator f Phone: (506) 44	13-6520	Administrator Phone: (506) 4	43-652	0	
	Fax: (506) 458- Email: tsa@nbr	4642 ower.com	Fax: (506) 458 Email: tsa@nb		com	
	Affiliated with		Comments			
	NO Status Notifica				5-08-01 Tariff changes.	
	Status Notifica	non	Provider Com Timing Rules (		e:	
	Comments					
	Path and PO	R/POD Information		Transn	nission Information	
	Point Of	Point Of	Increment		Class	
	Receipt	Delivery	YEARLY		FIRM	
	HQ-HVDC	MEPCO	Туре		Period	
	Path Name			OINT	FULL_PERIOD	
		-HVDC to MEPCO//	Window		Sub Class	
	Source	Sink	SLIDING		(none)	
			Curtailment I	Priority	7	
			NERC: 7 Other: 5			
	Dates	and Times		Req	uest Information	
	Start Time		Туре	• 1	Competing Request Flag	Impacted
	Nov 30, 2007 2	3:00 ES	ORIGINAL		N	885
	Stop Time Nov 30, 2022 2	3-00 ES	Capacity		Capacity Granted	
	Queued Time		Requested 300		300	
	Nov 05, 2007 1	5:11 ES	Offer Price		rice Price Units	
	Time of Last U	Jpdate	25,069.44	25,06		
	May 06, 2016		Ceiling Price	1	Negotiated Price Flag	
	Response Tim		25,069.44			
	Nov 05, 2007 2	3:16 ES	Status		Preconfirmed	
			CONFIRMED		YES	
			Status Comm (none)	ents		
			Reference Numl			
		<b>Posting Deal Reques</b>	t Sale Kelated	i Selle (none		
			Ancillary Servi	ices		
	Ancillary Serv	ices Required SC:M;	RV:M			
	Ancillary Serv	ices Link				
		Rea	ssignment Infor	rmation	1	
	Reference # 0	Capacity Start Time	Stop Time			
a 11, 2018 10:54:25 Eastern Time (Server ID: B)						